# **Asset Management for Retaining Walls**

Final report

Submitted to

Michigan Department of Transportation ORBP Number OR15-114

by

Adda Athanasopoulos Zekkos, PhD (PI) Jerome Lynch, PhD (coPI) Dimitrios Zekkos, PhD, PE (coPI) Athena Grizi, PhD (PostDoc) Kidus Admassu (Graduate Student Researcher) Brahim Benhamida, PE (MSG) Rich J. Spino, PE (MSG) Matthew Mikolajczyk, PE (formerly MSG)

Department of Civil & Environmental Engineering University of Michigan 2350 Hayward St Ann Arbor, MI 48109-2125

March 2020

# **TECHNICAL REPORT DOCUMENTATION PAGE**

1. Report No.	2. Government Accession N	<b>3</b> .	. MDOT Project Manager			
SPR-1676		Eric Burns				
21111070						
4. Title and Subtitle		5	. Report Date			
Asset Management for Retaining	ng Walls	Ň	March 23, 2020			
l los es l'inningenient i et i te minin	8	-				
		6	. Performing Organization	1 Code		
7. Author(s)		8	. Performing Organization	Report No.		
Adda Athanasopoulos-Zekkos,	Jerome Lynch, Dimi	trios Zekkos, N	None			
Athina Grizi, Kidus Admassu, I	Brahim Benhamida,	Rich J. Spino,				
Matthew Mikolajczyk						
9. Performing Organization Name and Addr	ess	1	0. Work Unit No. (TRAIS)			
The Regents of the University of	of Michigan					
3003 South State St.						
Ann Arbor MI 48109		1	1. Contract or Grant No.			
		2	2016-0068 Z2			
12. Sponsoring Agency Name and Address		1	3. Type of Report and Peri	od Covered		
Michigan Department of Transp	portation	F	Final Report			
Office of Research and Best Pra	actices	1	0/01/2016 to 12/31	/2019		
8885 Ricks Road						
Lansing ML 18917		14	4. Sponsoring Agency Code	e		
Lansing, wit 40917	Lansing, Mil 40917					
15. Supprementary Notes						
16. Abstract						
The work described here repre-	esents an attempt to	develop a com	prehensive risk ma	anagement		
framework for the asset manage	ement of retaining w	all structures. The	e work presented ir	cludes the		
development of a consing start	any that any hayand	have attractional in a	maatama ta aggagg ti			
development of a sensing strategy that can be used by structural inspectors to assess the coupled						
performance of the wall structu	re and the geotechni	cal system it supp	ports. A reliability f	framework		
was developed using first-order	r reliability methods	(FORM) to asses	ss the reliability fac	$(\beta)$ for		
wall components and incorpo	rates the consequer	ices of failure v	vith the estimated	structural		
reliability factors to provide a	basis for risk asse	ssment of retaining	ing walls. A new	inspection		
manual was developed to reflec	t the instrumentation	strategies and ris	sk analyses develor	ped		
inandar was developed to reflect the instrumentation strategies and fisk analyses developed.						
17. Key Word		18. Distribution Staten	nent			
Retaining Walls, Asset Management, Field No restrictions. This document is available						
Monitoring, Ground Settlement	Monitoring, Ground Settlement Thresholds. Ito the public through the Michigan					
		Deneutro public		witchigall		
		Department of I	ransportation.			
19. Security Classify. (of this report)       20. Security Classify. (of this page)			21. No. of Pages	22. Price		
Unclassified	Unclassified		334			

# TABLE OF CONTENTS

Technical Rep	ort Documentation Page	i
Table of Conte	ents	ii
List of Figures		V
List of Tables		viii
List of Symbol	ls	ix
List of Abbrev	iations	xi
Acknowledger	nents	xii
Executive Sum	1mary	xiii
1. Introdu	ction	1
1.1. O	rganization of report	1
1.2. O	bjectives	1
1.3. R	eview of Existing Guidelines	3
1.	3.1. Brutus and Tauber (2009) Synthesis Report and recommendations	16
1.	3.2. Minnesota DOT Synthesis Report (2013)	18
1.	3.3. National Park Service (NPS) Wall Inventory and Condition Assessment	
	Program (WIP) (2010)	19
1.	3.4. North Carolina DOT (2015)	25
1.	3.5. Colorado DOT (2003)	33
1.	3.6. Colorado DOT (2016)	37
1.	3.7. Wisconsin DOT (2017)	47
1.	3.8. Alaska DOT (2013)	52
1.	3.9. Pennsylvania DOT (2010)	54
1.	3.10. Nebraska DOR (2009)	58
1.	3.11. British Columbia Ministry of Transportation - Canada (2013)	60
1.	3.12. Victoria Highway Department - Australia (2014)	61
1.	3.13. Kentucky Transportation Cabinet (1998)	66
1.	3.14. City of Seattle (2014)	70
1.	3.15. The City of Cincinnati (2015)	75
1.4. M	ISE Wall Issues and Guidelines	78
1.5. R	etaining Wall Monitoring and Health Assessment Technologies	91
1.	5.1. LiDAR Mapping and Assessment	94
1.	5.2. Photogrammetry for condition assessment of ERS	96
1.	5.3. Inspection of retaining walls using Ground Penetrating Radar (GPR)	98
1.	5.4. Condition Assessment of Mechanically Stabilized Earth (MSE) Walls	100
2. Design	and Deployment of Wireless Monitoring Systems for the I-696 and M-10	
Retaini	ng Walls	103
2.1. Ir	ntroduction	103
2.2. R	etaining Wall Systems	105
2.	2.1. Overview of the I-696 and M-10 Retaining Wall Systems	105

		2.2.2. On-Site Visual Assessment	105
	2.3.	Long-Term Monitoring System Design and Installation	106
		2.3.1. Sensor Placement on the Wall Panels	107
		2.3.2. Sensing Transducers	111
		2.3.3. Wireless Sensing Node	113
		2.3.4. Dual-Axis Tilt Calculation	115
		2.3.5. Packaging Wireless Sensor Nodes	116
		2.3.6. Field Installation on the Wall Systems	116
	2.4.	Data Acquisition Process	117
3.	Resp	oonse Data Analysis of the I-696 and M-10 Retaining Wall Systems	122
	3.1.	Introduction	122
	3.2.	Performance of the Wireless Sensor Units	123
	3.3.	Tall I-696 Retaining Wall Panel	125
		3.3.1. Description of the Wall System	125
		3.3.2. Daily Mean Responses	128
		3.3.3. Response Scatter Plots	132
		3.3.4. Wall Deflection Curves	134
	3.4.	Short I-696 Retaining Wall System	146
	3.5.	M-10 Retaining Wall System	151
		3.5.1. Description of the Wall System	151
		3.5.2. Preliminary Observations from 2012 and 2013	153
		3.5.3. Daily Mean Responses and Scatter Plots	155
4.	Deve	eloping the Michigan Earth Retaining Structure Element Inspection Manual	159
	4.1.	Introduction	159
	4.2.	Reviewing Existing Wall Management Methods	159
	4.3.	Conducting Field Review of 10 MDOT Retaining Wall Systems	160
	4.4.	Development of Earth Retaining Structure Criteria	161
	4.5.	Develop Structure and Appraisal (SI&A) Items for Retaining Walls	161
	4.6.	Develop Inspection Procedure for Retaining Walls (MiERSEIM)	162
5.	Risk	Assessment of the Tall I-696 Retaining Wall System	164
	5.1.	Introduction	164
	5.2.	Risk Assessment Framework	166
	5.3.	Reliability of the I-696 Tall Wall	170
		5.3.1. Steel Reinforcement Capacity	171
		5.3.2. Effect of Corrosion	172
		5.3.3. Strain Response Analysis	174
		5.3.4. Reliability Analysis	180
	5.4.	Risk Assessment	182
6.	Cone	clusions – Recommendations	187
	6.1.	Conclusions	187

	6.2.	Recon	nmendations for Future Work	191
7.	Reco	ommenc	ded Implementation Plan	192
	7.1. List of products expected from research			
	7.2.	nce for research results	192	
	7.3.	Activi	ties for successful implementation	192
	7.4.	Criteri	a for judging the progress and consequences of implementation	193
	7.5.	Costs	of implementation	193
8.	Refe	rences.		194
Ap	pendi	ces		202
	A. E	Existing	Tools Developed by Agencies	203
		A.1.	Materials from FHWA-NPS WIP (2010)	204
		A.2.	Materials from North Carolina DOT (2015)	210
		A.3.	Materials from Colorado DOT (2016)	214
		A.4.	Materials from Wisconsin DOT (2016)	224
		A.5.	Materials from British Columbia Ministry of Transportation (2013)	227
		A.6.	Materials from Victoria Roads (2014)	232
		A.7.	Materials from the Kentucky Transportation Cabinet (1998)	242
		A.8.	Materials from the City of Seattle (2014)	246
		A.9.	Materials from the Ohio DOT (2007)	251
		A.10.	Materials from the Utah DOT (2009)	254
	B. N	Aoveme	ent of the M-10 Wall Panel During January 2020	261
		<b>B</b> .1.	Introduction	261
		B.2.	Visual Inspection of Wall	261
		B.3.	Measured Wall Movement	
		B.4.	Conclusions	
	C.M	ichigan	Earth Retaining Structure Element Inspection Manual	268

# LIST OF FIGURES

Figure 1.1. Proposed processes for a geotechnical asset management plan	15
Figure 1.2. WICAS home screen	27
Figure 1.3. Wall Search menu	28
Figure 1.4. Bridge zone	37
Figure 1.5. Effective Bridge Zone	
Figure 1.6. Separation of wall at bridge zone	
Figure 1.7. Structure Map in SAMI application	40
Figure 1.8. Bridge zone limits	53
Figure 1.9. Observed problems in MSE walls	84
Figure 1.10. LiDAR mapping of MSE wall	95
Figure 1.11. Retaining wall on M-10	98
Figure 1.12. Semi-automated survey apparatus	99
Figure 2.1. M-10 and I-696 retaining wall site locations	104
Figure 2.2. M-10 retaining wall system distress (June 2017)	106
Figure 2.3. Issues observed on I-696 wall panels (June 2017)	107
Figure 2.4. Possible vertical wall deformations of the M-10 wall	108
Figure 2.5. M-10 instrumentation: two tilt-meters, two strain gages & two thermistors	109
Figure 2.6. I-696 instrumentation on two separate panels 40 feet apart	110
Figure 2.7. Strain gage set-up	112
Figure 2.8. Urbano wireless sensor node for data collection	114
Figure 2.9. Two axes used for dual-axis tilt sensing	115
Figure 2.10. Inside the wireless sensor enclosure	117
Figure 2.11. Installation of instrumentation on the I-696 wall	118
Figure 2.13. Screenshot of the online data management portal (Exosite)	119
Figure 2.12. Data acquisition architecture for the I-696 and M-10 retaining wall moni	toring
systems	119
Figure 3.1. Performance of the wireless sensor nodes on the M-10 and I-696 panels	121
Figure 3.2. Dimensions and structural details of the 'tall' wall panel	125
Figure 3.3. The daily mean response of the I-696 tall wall panel (tilt)	127
Figure 3.4. The daily mean response of the I-696 tall wall panel (strain)	128
Figure 3.5. Night and day daily mean response of the I-696 tall wall panel (tilt)	129
Figure 3.6. Night and day daily mean response of the I-696 tall wall panel (strain)	129
Figure 3.7. Relationships between I-696 tall wall mid-height tilt, mid-height strain, and b	ottom
strain as a function of wall temperature	131
Figure 3.8. Relationships between I-696 tall wall mid-height tilt and bottom strain as a fun	nction
of cumulative precipitation	131

Figure 3.9. Relationships of the I-696 tall wall top tilt to cumulative precipitation, temperature	e
and cumulative precipitation	132
Figure 3.10. Cross-sectional view of the I-696 tall wall and its backfill lateral loads	133
Figure 3.11. Static backfill earth pressures, surcharge and hydrostatic pressure	136
Figure 3.12. Cracked section assumption and cracked section transformation	137
Figure 3.13. Tapering of the I-696 cantilever RC retaining wall and the parapet wall	138
Figure 3.14. Lateral earth pressures on tall I-696 wall: backfill, surcharge, hydrostatic	139
Figure 3.15. Discretization of the I-696 cantilever wall	140
Figure 3.16. Actual and theoretical deflected shapes of the tall I-696 RC cantilever wall	141
Figure 3.17. The daily mean response of the I-696 short wall panel (tilt)	143
Figure 3.18. The daily mean response of the I-696 short wall panel (strain)	144
Figure 3.19. Night and day daily mean response of the I-696 short wall panel (tilt)	145
Figure 3.20. Night and day daily mean response of the I-696 short wall panel (strain)	145
Figure 3.21. Relationships between I-696 short wall bottom strain as a function of temperature	re
and cumulative precipitation	146
Figure 3.22. Relationships between I-696 short and tall wall panels	146
Figure 3.23. Actual and theoretical deflected shapes of the short I-696 RC cantilever wall	147
Figure 3.24. Structural details of the M-10 wall	148
Figure 3.25. Preliminary investigation of the M-10 wall in 2012-2013	149
Figure 3.26. Wall response time histories of the M-10 wall (tilt)	150
Figure 3.27. Wall response time histories of the M-10 wall (strain)	152
Figure 3.28. Scatter plots of M-10 wall panel response to temperature	152
Figure 5.1. Illustrations of how to go from visual inspection to risk-based asset management	159
Figure 5.2. Proposed risk assessment framework for retaining wall structures	161
Figure 5.3. Illustration of demand and capacity distributions in relationship to the reliabilit	y
index, $\beta$	162
Figure 5.4. PDF of Yield Strength of ASTM A615 Grade 60 Steel Reinforcement Bars	166
Figure 5.5. Percent section/diameter reduction for different corrosion activity ratings	167
Figure 5.6. Overview of the reliability and risk assessment framework for the I-696 retaining	g
wall system using tilt and strain measurements from wall front side	168
Figure 5.7. Load effect estimation from tilt responses	169
Figure 5.8. Top and mid-height tilt for tall I-696 wall panel assuming no corrosion for variation	IS
in surcharge load, q, and water saturation level, h <sub>sat</sub>	170
Figure 5.9. Daily mean response of tilt and estimation of surcharge load, q, and water saturatio	n
level, h <sub>sat</sub>	171
Figure 5.10. Daily maximum bending moment variation at the base of the wall for three states of	of
assumed corrosion in the tensile steel reinforcement	172
Figure 5.11. Vertical steel reinforcement strain at the base of the wall backside due to flexura	al
moment	172
Figure 5.12. Strain profiles on the I-696 tall and short walls at different seasons of the year	174

Figure 5.13. Scatter plot showing wall strain response to thermal load versus change in	
temperature is linear17	'4
Figure 5.14. Histograms of estimated vertical reinforcement steel tensile stress compared to	
probabilistic model of steel reinforcement yield stress17	'5
Figure 5.15. Reliability index values for varying levels of section loss in the tall I-696 retaining	
wall panel backside vertical steel reinforcement17	7
Figure B.1. Manually measured movement of the M-10 wall panel at the top wall parapet in	
January 202025	56
Figure B.2. Top road surface at the top of the suspected failed M-10 wall panel (January 2020);	
major tensile crack in asphalt surface evident parallel to wall face	6
Figure B.3. Measurement of the M-10 wall panel including early January 2020 with a high	
precipitation event on January 12, 2020 evident25	57
Figure B.4. Top tilt of instrumented M-10 retaining wall panel histogram25	58
Figure B.5. 3D point cloud generated by Pix4D from aerial UAV imagery collected on January	
22, 2020 and cross section of Panel 425	;9
Figure B.6. Plan view of 3D point cloud model with nine panels of the M-10 retaining wall	
system identified and measured angle of tilt of the wall at each panel	50
Figure B.7. Deformation curves of the drifted M-10 retaining wall system	51

## LIST OF TABLES

Table 1.1. Summary of agencies with Inventory or Inspection Program	5
Table 1.2. Classification of Wall Function and Type according to the FHWA-NPS a	and
Wisconsin DOT programs	7
Table 1.3. ERS Data Attributes	9
Table 1.4. Proposed condition rating system	30
Table 1.5. Risk Assessment Matrix	31
Table 1.6. Consequence of Failure Criteria Definitions	31
Table 1.7. Failure Likelihood Criteria Definitions	32
Table 1.8. Action Assessment Table	32
Table 1.9. Inspection Assessment Table	32
Table 1.10. Properties of Elements and Components	34
Table 1.11. Maintenance Actions	34
Table 1.12. Routine Inspection Tasks	35
Table 1.13. Condition States for Elements	36
Table 1.14. Common material defects	47
Table 1.15. Repair timeline depending on priority level	51
Table 1.16. Performance ratings assigned to wall elements in Pennsylvania DOT inspect	ion
process	57
Table 1.17. Critical components for retaining walls	62
Table 1.18. Routine maintenance activities on bridges and other structures	63
Table 1.19. Kentucky wall inspection study	67
Table 1.20. AASHTO metal loss model and backfill specifications	79
Table 1.21. Summary of US State MSE wall corrosion assessment programs	80
Table 1.22. Summary of backfill requirements for some State DOTs	81
Table 1.23. Monitoring techniques and measuring parameters for Earth Retaining Structures	92
Table 1.24. Monitoring technologies and references	93
Table 2.1. Urbano performance specifications	114
Table 3.1. Reinforcement bars of the tall panel of the I-696 wall system	124
Table 3.2. Typical bulk and submerged densities of selected soils	134
Table 3.3. Static backfill earth pressures (no surcharge or hydrostatic pressure)	135
Table 3.4. Formulae to calculate backfill, surcharge and hydrostatic pressures	136
Table 3.5. Reinforcement bars of the wall panel of the M-10 wall system	149
Table 5.1. Updated FHWA condition ratings to include the reliability index (Bi) thresholds t	hat
bound corresponding condition rating codes	176
Table 5.2. Red-yellow-green (R-Y-G) risk categories mapping reliability index ( $\beta$ ) values w	vith
consequence of failure event	178
Table 5.3. Consequence Categories for Highway Retaining Wall Structures	178

# LIST OF SYMBOLS

β	Reliability index
h <sub>sat</sub>	Level of soil saturation
q	Optimal surcharge load
ε	Strain
σ	Stress
Т	Temperature
Ec	Modulus of elasticity of concrete
Ĕ	Modulus of elasticity of steel
F	Fahrenheit
и	Microns
ρ'	Submerged density of soil
$\rho_h$	Bulk density of soil
Posat	Saturated density of soil
M	Flexural moment in wall element
psi	Pounds per square inches
ksi	Kilo-pounds per square inches
θ	Tilt
E <sub>eff</sub>	Effective modulus of elasticity
I <sub>am</sub>	Cracked moment of inertia
v	Depth to Neutral Axis (NA) of section
f.	Yield strength of steel
f.	Compressive strength of concrete
R	Risk of failure
P <sub>c</sub>	Probability of failure
С.	Consequence of failure
$C_{\$}$	Canacity
מ	Demand
C C	Limit state function
4	Area of steel on tensile side
$A_S$	Area of steel on compressive side
$A_s$	Cumulative rain
C <sub>rain</sub>	Rankine's earth pressure coefficient
n <sub>a</sub>	Lateral pressure intensity
w c/c	Center to center
19	Poisson's ratio
GF	Gage factor
Vin	Input voltage
$V_{n}$	Output voltage
$^{\circ}C$	Degree Celsius
a	Gravitational acceleration
ft.or'	Linear feet
in or "	Linear inches
#	Bar size number

<u>Բ</u>	Center line
MHz	Mega hertz
kB	Kilo bytes
Ø	Internal angle of friction

## LIST OF ABBREVIATIONS

American Association of State Highway and Transportation Officials
Analog-to-Digital Converter
Digital Terrain Model
Far Side
Federal Highway Administration
First order reliability method
Geotechnical Asset Management
Ground Penetrating Radar
Ground Water Table
Light Detection and Ranging
Load Resistance Factor Design
Linear Variable Displacement Transformer
Moving Ahead for Progress in the 21st Century Act
Michigan
Michigan Bridge Element Inspection Manual
Michigan Earth Retaining Structure Element Inspection Manual
Michigan Structure Inspection Manual
Mechanically Stabilized Earth
Near Side
National Cooperative Highway Research Program
Reinforced Concrete
System for Asset Management and Inspection
South Bound
Structure from Motion
Structure Management Section
Serial Peripheral Interface
Universal Asynchronous Receiver Transmitter
Unmanned Aerial Vehicle
Visual Inspection Information

## ACKNOWLEDGEMENTS

The authors would like to thank the Michigan Department of Transportation (MDOT) for f unding this project. They wish to thank especially Eric Burns, Project Manager, for the co ordination on lane closures and access to the instrumented sites, and his immediate response to the research team's inquiries. They also wish to thank the entire Research Panel for their thoug htful comments and guidance throughout the project. The authors would finally like to thank Ms. Cassandra Champagne, PhD student at the Univ. of Michigan, for her help with the 3D point cloud model of the I-696 site after movement was detected in January 2020.

## **EXECUTIVE SUMMARY**

The purpose of transportation asset management is to meet life-cycle performance goals through the management of physical assets in the most cost-effective manner (FHWA, 2013). Currently, many agencies manage geotechnical features on the basis of "worst-first" conditions, reacting to failures and incurring significant safety, mobility, environmental, and other intangible costs. Whereas, this may be an appropriate response for failures following natural hazards, the goal of geotechnical asset management is to implement project planning and selection on the basis of "most-at-risk" for the asset class with consideration of collective and site specific risks throughout the system life cycle. This report presents results of a research project focused on developing a comprehensive risk management framework for the asset management of retaining wall structures, as well as a new inspection manual for retaining walls.

An instrumentation strategy was proposed for monitoring highway retaining wall systems. Emphasis was placed on a strategy suitable for rapid installation after wall construction to provide data on wall behavior. The project considered instrumentation that can be installed on exposed wall surfaces and that is easy to deploy. The instrumentation strategy adopted tiltmeters to measure wall tilt, long-gage strain gages to measure thermal and flexural strains, and thermistors to measure wall temperatures. 11 wireless sensor units were installed on the retaining wall systems along the I-696 (two wall panels) and M-10 (one wall panel) freeway corridors that collected a total of 16 measurements from the three wall panels and the performance of the instrumented retaining wall systems was assessed using the data collected. An aim of this project was the development of a risk assessment framework that utilized structural monitoring data to define the reliability of the retaining wall system. Combined with the consequence of exceed the limit state obtained from the reliability analysis, the risk of the asset can be determined. Quantifying risk in this way established for the first time, a clear pathway of using monitoring data within a broader GAM strategy. An especially valuable feature of the proposed risk assessment procedures was the explicit inclusion of visual inspection information to define structural conditions critical to the evaluation of structural capacity.

Finally, the report includes the newly developed Michigan Earth Retaining Structure Element Inspection Manual (MiERSEIM). As part of its development, ten sites from a list of 74 provided by the Michigan Department of Transportation (MDOT) were chosen based on factors including type, size, condition, risk/consequence view, past monitoring information, and ease of access. A t eam of personnel from the MDOT, University of Michigan, and the Mannik & Smith Group (MSG) visited each site to determine inspection criteria and methods through hands-on inspection. After all the investigation and data layout was finalized, the MiERSEIM was developed using a format that mirrors the Michigan Bridge Element Inspection Manual (MiBEIM ) for its familiarity.

## **1. INTRODUCTION**

### 1.1 Organization of Report

Chapter 1 presents the objectives of this research report and an extensive literature review. Chapter 2 provides details on the instrumentation deployed at the two MDOT sites, and the collected data. Chapter 3 provides a detailed overview of how the wireless monitoring systems performed over the one-year monitoring period. Chapter 4 presents the process of developing the new inspection manual (MiERSEIM). Chapter 5 discusses a data-driven risk assessment methodology based on long-term monitoring data and visual inspection information. Chapter 6 lists research conclusions and recommendations for future research. Finally, Chapter 7 provides a recommended implementation plan as a product of this research and Chapter 8 includes all references for the report.

### 1.2 Objectives

The overarching goal of this study is to define an inspection process and develop a comprehensive risk assessment framework for the asset management of MDOT's vast inventory of retaining wall structures. To accomplish this goal, the project has 7 research and implementation objectives:

<u>Objective #1:</u> Conduct Detailed Review of Reported Retaining Wall Management Methods and Failure Case Studies: A detailed review of identified retaining wall failures in Michigan (Jansson, 2007; Jansson, 2013) in addition to other failure case studies in other states will illuminate retaining wall system parameters that should be monitored for health assessment.

<u>Objective #2:</u> Conduct Field Review of MDOT Retaining Wall Systems: A detailed field review to be performed of 10 retaining wall systems in Michigan.

<u>Objective #3:</u> Assess the State-of-Practice in Retaining Wall Monitoring and Repair: Multiple monitoring methods have been proposed for monitoring retaining walls ranging from stationary wired, to stationary wireless, all the way to mobile monitoring platforms (such as those mounted on UAVs and cars).

<u>Objective #4:</u> Propose an Effective Monitoring Strategy: A monitoring strategy is developed for MDOT that feeds quantitative data directly to the proposed risk management framework. A permanent monitoring and inspector-operated non-destructive evaluation (NDE) methodologies is proposed as part of the larger risk management framework.

<u>Objective #5:</u> Development of Data-Driven Risk Management Framework: The team will develop a reliability assessment methodology to assess the factor of safety and probability of failure of retaining wall systems. Coupled with a quantification of failure consequences, a risk assessment methodology is developed to aid MDOT decision making.

<u>Objective #6:</u> Develop Structure and Appraisal (SI&A) Items for Retaining Walls: To acquire detailed records of the MDOT inventory of retaining walls, items required to index retaining wall assets to be added to MDOT's SI&A Codes. Existing codes for bridges and other highway assets to be used as much as possible. The code set proposed for retaining walls will ensure data is available for the risk management procedures developed.

<u>Objective #7:</u> Develop Inspection Procedure for Retaining Walls: A detailed and in-depth inspection procedure will be developed and incorporated in the Michigan Structure Inspection Manual (MiSIM). A stand-alone inspection manual (similar to MiBEIM) to also be written. Procedures will include recommended instrumentation, inspection frequencies according to the type of wall, condition, design, functionality, consequences and other factors relevant to the risk assessment.

Retaining walls are important infrastructure assets which are generally overlooked compared to bridges and pavements in terms of asset management practices. To date, there are several highway agencies that have established retaining wall inventory and inspection programs and very few that have retaining wall asset management programs. Most of the agencies are currently trying to develop their own asset management programs. With the Moving Ahead for Progress in the 21<sup>st</sup> Century Act (MAP-21 2012), state highway agencies are required to "develop a risk-based asset management plan for the National Highway System to improve or preserve the condition of the assets and the performance of the system, including signs and sign structures, earth retaining walls and drainage structures". An earth retaining structure and a retaining wall are defined as:

**Earth Retaining Structure (ERS):** Any structure intended to stabilize an otherwise unstable soil mass by means of lateral support or reinforcement (Sabatini et al. 1997).

**Retaining wall:** A wall which face makes an angle of 70 degrees or more with the horizontal and retains earth (National Highway Institute). Recognizing that earth retention structures other than cut and fill walls may need to be captured in the Inspection and Inventory (I&I) program, some groups have changed this criterion to a 1:1 face angle to also include earth retention structures such as rock buttresses, gabion walls, rockeries, etc. that don't directly meet the NHI design definition but are nonetheless critical assets.

As indicated by the Federal Highway Administration (FHWA), ERSs are "constructed in challenging site conditions, including mountainous terrain, soft ground, and sites that are below water". Newer ERS systems "require that engineered materials such as plastics, concrete, and steel be buried in harsh underground environments...that may adversely influence the long-term engineering properties of the materials. ERSs often have assumed design lives of 100 years, but knowledge of actual design life for these structures is minimal and failures that have occurred to date have happened without warning. Repairing these failed structures is very expensive, complex, and difficult".

In addition, a large number of the ERSs in the United States date from the major Interstate Highway construction in the late 1950s through the 1970s. The earliest of these Interstate-era walls are approaching the end of their anticipated service lives. In the case of the U.S. National Park Service (NPS), most of the walls date to the 1930s through 1940s, when most of the major parks were developed (Brutus and Tauber 2009).

Inspections of ERS should be based, to the extent possible, on the relevant techniques and procedures used in bridge inspection. These are described in detail in the FHWA's "Bridge Inspector's Reference Manual 2006" (Ryan et al. 2012). These techniques reflect decades of experience and there is no need to reinvent them (Brutus and Tauber 2009).

### **1.3 Review of Existing Guidelines**

An extensive literature review was conducted to identify guidelines followed at the State and National level for inspection and assessment of the performance of retaining wall systems. Currently, 23 transportation agencies have implemented inventory and/or inspection programs for their earth retaining systems in the United States and abroad:

- 1. Federal Highway Administration (FHWA)/US National Park Service (NPS)
- 2. California Department of Transportation
- 3. Colorado Department of Transportation
- 4. Kansas Department of Transportation
- 5. Maryland Department of Transportation
- 6. Minnesota Department of Transportation
- 7. Missouri Department of Transportation
- 8. New York State Department of Transportation
- 9. Oregon Department of Transportation
- 10. Pennsylvania Department of Transportation
- 11. The City of Cincinnati
- 12. New York City Department of Transportation
- 13. British Columbia Ministry of Transportation (Canada)
- 14. Alaska Department of Transportation
- 15. Kentucky Transportation Cabinet
- 16. Nebraska Department of Roads
- 17. North Carolina Department of Transportation
- 18. Ohio Department of Transportation
- 19. Utah Department of Transportation
- 20. Vermont Agency of Transportation
- 21. Wisconsin Department of Transportation
- 22. The City of Seattle
- 23. VicRoads, Victoria State Department (Australia)

The data is based on the highway agency programs for which full access was granted and a summary is presented in Table 1.1 and the following sections. Details of individual highway wall inventory and inspection programs will be discussed in subsequent sections.

AGENCY	YEAR	ROUTINE INSPECTION FREQUENCY (years)	HEIGHT (ft)	WALL FACE ANGLE (degrees)	DISTANCE FROM ABUTMENT (ft)	DATABASE	RATING SYSTEM
National Parks Service	2010	10	4	45	40	Visidata, Access	5-100
Alaska DOT	2013	5	4	45	100	Access	Good, Fair, Poor
Colorado DOT	2016	6	4	45	40	SAMI	0-9
North Carolina DOT	2015					Access, Oracle	1-4
Pennsylvania DOT	2010	5			100	iForms	2-8
Nebraska DOR	2009						0-9
New York City DOT	1998	5	6				Safe, Safe w/minor repair, Safe with repair and monitoring, Unsafe
New York State DOT	2015		6		33	ArcMap	1-7
Wisconsin DOT	2011	6	5				Good, Fair, Poor, Severe
Oregon DOT	2007	5	4			Access	Good, Fair, Poor
Utah DOT	2009					MAP Window GIS, Access	Yes or No
Ohio DOT	2007						Yes or No
City of Cincinnati	1990	6	2			FoxPro, Oracle, ArcGIS	0-4
City of Seattle	2009		4			Access, Hanson	0-100
British Columbia MOT	2013	TBD	6.5	45		DataBC, ArcGIS	Excellent, Good, Fair, Poor, Very Poor
VicRoads-Australia	2014	2-5	4.9	45	16.4		1-4

 Table 1.1. Summary of agencies with Inventory or Inspection Program

### WALL CLASSIFICATION

Retaining walls can be divided in different categories based on their function and type. Walls that are constructed from the bottom up are fill walls, while walls built from the top to the bottom are cut walls. Most inventories include all wall types. Some agencies have inventory programs only for Mechanically Stabilized Earth (MSE) walls because of resource constraints or just because MSE walls represent the majority of the wall types in the State. Agencies that fall into the latter category are: Nebraska DOR, Ohio DOT, Pennsylvania DOT and Utah DOT. Table 1.2 depicts different wall functions and types used by the FHWA-NPS (DeMarco et al. 2010) and the Wisconsin DOT (2017). Some agencies include walls associated with bridges and culverts.

### WALL ACCEPTANCE CRITERIA

Agencies use different criteria when determining which retaining walls to include in their inventory program. A summary of different guidelines that are used is presented in Table 1.1. The main criteria are:

- Minimum exposed height of wall (visible or total): 2-6 ft
- Minimum height of retained earth
- Minimum length of wall (PennDOT)
- Minimum face slope greater than 45 degrees
- Wall batter, location relative to roadway, association with bridge/culvert
- Walls with distance from roadway bridge abutment: 16.4 ft to 100 ft
- Wall ownership

Noise walls do not retain earth, thus are usually under another inventory program. Bridge and culvert walls are usually part of the Bridge and Culvert Inspection Program, respectively. Agencies use a specific distance from the bridge abutment to decide whether to include a wall to the Earth Retaining Wall Program. For example, 40 ft is used by the FHWA-NPS and Colorado DOT, 100 ft is used by Alaska DOT and Pennsylvania DOT, 10 m is used by New York State DOT and 5 m is used by Victoria State Department in Australia. New York State and New York City DOTs do not include railroad-owned walls in their programs. It is suggested that an ERS Inventory include all walls regardless of ownership, since a potential failure (e.g. New York City

Department of Buildings (2007) and Tennessee Department of Transportation (2003)) might impact the highway facility (Brutus and Tauber 2009).

### **RANKING METHODS OF ERSs**

The following methods are used by different agencies to identify and locate retaining walls:

- Physically locating the wall in the field by walking or driving to the wall
- Using aerial surveys
- Using as-built drawings or records
- Using roadway video surveys
- Using Google Maps, Bing Maps
- Using Asset Identification
- Through LiDAR and ARAN surveys
- Flood insurance, drainage, public utility maps
- Using a Geographical Information System
- Using staff knowledge
- Adding new walls as constructed

### WALL ATTRIBUTES

Brutus and Tauber (2009) compiled a list of 96 possible attributes from inventory forms provided by several agencies that currently maintain inventory or inspection programs (Table 1.3). Some of the data is to be collected in the field and some can be completed back in the office. The level of detail in the database is to be decided by the highway agency according to its needs, budget and other factors.

Wall Function		Wall Type		
FHWA-NPS	Wisconsin DOT	FHWA-NPS	Wisconsin DOT	
Fill Wall	Fill Wall	Anchor, Tieback H-Pile	Anchor, Sheet Pile	
Cut Wall	Cut Wall	Anchor, Micropile	Anchor, Soldier Pile	
Bridge Wall	Dockwall	Anchor, Tieback Sheet Pile	Anchor, Secant/Tangent	
Head Wall		Bin, Concrete	Bin, Concrete Cantilever, Sheet Pile	
Switchback Wall		Bin, Metal Cantilever, Soldier Pile		
Flood Wall		Cantilever, Concrete	Cantilever, Secant/Tangent	
Slope Protection		Cantilever, Soldier Pile	Cast-in-place, Gravity	
		Cantilever, Sheet Pile	Cast-in-place, Cantilever	
		Crib, Concrete	Gravity, Gabion	
		Crib, Metal	Gravity, Modular Block	
		Crib, Timber MSE, Modular Blocks		
		Gravity, Concrete Block/Brick MSE, Precast Panel		
		Gravity, Mass Concrete MSE, Wire Face		
		Gravity, Dry Stone Soil Nail		
		Gravity, Gabion		
		Gravity, Mortared Stone		
		MSE, Geosynthetic Wrapped Face		
		MSE, Precast Panel		
		MSE, Segmental Block		
		MSE, Welded Wire Face		
		Soil Nail		
		Tangent/Secant Pile		

# Table 1.2. Classification of Wall Function and Type according to the FHWA-NPS and Wisconsin DOT programs

SURVEY LOG DATA	FUNCTION DATA	STRUCTURAL DATA, PRELIMINARY	CONDITION DATA FROM INSPECTION	CONSEQUENCES OF FAILURE FACTORS
ID number	Functional type	Wall face material	Inspection report	Critical wall height
Date of Survey	Supported feature	Apparent wall type	Inspection date	Critical distance
Times of arrival and departure	Protected feature	Wall surface treatment	Name of inspector	Roadway type and lanes
Surveyed by	Photo(s) of supported and/or protected features	Wall top feature	Prior documentation reviewed	Sensitive facility supported
Weather	DIMENSION DATA, GENERAL	Top of wall attachments	Potential failure type	Sensitive facility protected
Soil Moisture	Exposed height	Wall face attachments	Condition rating	COF rating
Work-zone safety devices or measures	Total length	STRUCTURAL DATA, VERIFIED	Performance rating	Traffic volumes
Special access equipment	Wall face slope	Structural type	Projected replacement date	Interchange distances
LOCATION DATA	Total height	Total wall face	Recommended action type	Utilities near top of wall
GPS location coordinates	Estimated area of exposed face	Estimated replacement cost per square foot	Recommended action summary	Utilities near base of wall
Location	Exposed height at beginning point	Cost estimate reference	HISTORY AND OWNERSHIP	Utilities on wall face
Offset	Exposed height at end point	Estimated total replacement cost	Year built	Detour length
Location photos	Height above retained soil	Wall face angle as built	New or retrofit	Affected locations
District/political subdivision	Upslope angle	Foundation type	Design service life	ACTION PRIORITY
End coordinates	Downslope angle	Proprietary type	Current owner	Action approved
Bridge/culvert association	Criterion length	Fill material	Owner contact information	Action priority
Other related feature	Offset criterion portion	CONDITION DATA, PRELIMINARY	Original owner	Action date scheduled
Access constraints	Photo(s) of top profile	Checklist conditions	Original contract number	Action completed
Did constraints affect accuracy?	Roadside features above	Inspection priority	Original cost	
Block and lot number	Roadside features below	Condition photos and sketches	Maintenance/repair /modification record	
Photo(s) of access constraints	Photos of roadside features		Original designer	
			Original contractor	

## Table 1.3. ERS Data Attributes (after Brutus and Tauber 2009)

### **INSPECTION FREQUENCY**

The frequency of inspection varies on the conditions found. Routine wall inspections range from two to ten years (Table 1.1), with the most common being a 5-year interval. More frequent inspections may be triggered by (Brutus and Tauber 2009):

- Walls exhibiting poor performance
- The environmental setting (regional climate, geology, etc.). In cold climates, for instance, walls susceptible to freeze-thaw cycles may require more frequent inspections
- The age of the wall. Older walls may require more frequent inspections
- Certain recent wall types (e.g., steel reinforced earth retaining structures or MSEs) where long-term performance records are not available
- The consequence of failure
- Occurrence of an event, such as flood or weather-related damage, or a vehicle impact, or an earthquake, etc.

### WALL ELEMENTS ASSESSMENT

The FHWA-NPS WIP divides the wall elements that need to be evaluated into primary and secondary. Other agencies use a simple check list to assess the overall wall condition. The following wall elements are assessed by highway agencies:

- Wall type
- Foundation
- Wall alignment
- Facing structure/treatments
- Surface coatings
- Attachments
- Guardrails/parapets
- Backfill material
- Backfill slope
- Drainage
- Erosion

- Vegetation
- Roadway
- Curb/Berm/SW/shoulder
- Adjacent features

### CONDITION OBSERVATIONS

After the wall elements are identified, their condition is assessed in terms of observed element distress or deterioration. The FHWA-NPS WIP evaluates the elements in terms of type, severity, extent and urgency. Brutus and Tauber (2009) recommend checking the following conditions when inspecting a retaining wall:

- 1. Wall or parts of it, out of plumb, tilting or deflected
- 2. Bulges or distortion in wall facing
- 3. Some elements not fully bearing against load
- 4. Joints between facing units (panels, bricks, etc.) are misaligned
- 5. Joints between panels are too wide or too narrow
- 6. Cracks or spalls in concrete, brick, or stone masonry
- 7. Missing blocks, bricks, or other facing units
- 8. Settlement of wall or visible wall elements
- 9. Settlement behind wall
- 10. Settlement or heaving in front of wall
- 11. Displacement of coping or parapet
- 12. Rust stains or other evidence of corrosion of rebars
- 13. Damage from vehicle impact
- 14. Material from upslope rockfall or landslide adding to load on wall
- 15. Presence of graffiti (slight, moderate, heavy)
- 16. Drainage channels along top of wall not operating properly
- 17. Drainage outlets (pipes/weep holes) not operating properly
- 18. Any excessive ponding of water over backfill
- 19. Any irrigation or watering of landscape plantings above wall
- 20. Root penetration of wall facing
- 21. Trees growing near top of wall

22. Any other observations not listed above

Since it is easier to identify drainage problems by inspections during or after heavy rains, the inspection engineer should arrange for such inspections, if the adequacy of the drainage system is in doubt.

### **RATING SYSTEM**

The types of rating systems used to evaluate ERSs can vary between qualitative assessments and quantitative assessments. The FHWA-NPS program uses a "Condition narrative" which is a descriptive narrative of element condition. These narratives are then converted to a numerical "Condition rating" ranging from 1 to 10. A wall performance rating is also determined along with the element condition ratings, using again a scale from 1 to 10, and the combination of these two create an overall wall performance rating ranging from 5 to 100. Conversion of this numeric rating to a qualitative description can be approximately achieved by dividing the rating by 10 and comparing it to the element and wall performance definitions.

Some agencies, e.g. North Carolina DOT and VicRoads simply follow the four-level rating scale that is used in AASHTO's "Manual for Bridge Evaluation" (AASHTO 2010a). The City of Cincinnati uses a scale from 0 to 4. Nebraska DOR and Colorado DOT follow the 0 to 9 scale found in the "Recording and Coding Guide for the Structure Inventory and Appraisal of the Nation's Bridges" published by FHWA (US DOT 1995). Oregon DOT has a three-level rating system based on Good, Fair and Poor condition ratings. Ohio DOT and Utah DOT have simple "yes" or "no" condition rating responses. The New York City DOT and the New York State DOT use a scale from 1 to 7. A summary of the different rating systems utilized is shown in Table 1.1.

### **RISKS ASSOCIATED WITH ERS FAILURE**

After the wall condition is evaluated, risk of the likelihood and consequences of an adverse event is assessed. Most of the agencies do not have a risk-based asset management plan for retaining walls with the exception of FHWA-NPS, Colorado DOT and North Carolina DOT. Some of the consequences of failure to be considered are:

• Failure potential

- Extent of failure
- Threats to life safety
- Budget impacts for wall repair
- Link criticality/redundancy
- Average Daily Traffic (ADF) impacts
- Disruption of adjacent utility lines (water mains, electrical conduits)
- Impact to environmental resources

Recommendations for further actions are then provided. Depending on the level of risk that is determined the inspector can recommend:

- No action is needed
- Monitor of the wall is required to determine what action should be followed
- Maintenance: routine maintenance activities are required to delay wall deterioration
- Repair: non-routine restoration of wall elements is required
- Rehabilitation: replacement of wall elements or the entire structure is required

### DATABASE/MAPPING SOFTWARE

Agencies maintain their retaining wall databases through different platforms. A single interface that would provide access to inventory, inspection and geospatial data, as well as photographs, drawings and documents would be the best approach. Some agencies use the PONTIS bridge management system to inventory their walls. Other popular databases that are used are Microsoft Access, Oracle, ESRI GIS and other GIS software (Table 1.1).

Some agencies link their Wall Management System to other management systems. For example, Minnesota's DOT system is linked to the permitting department, Pennsylvania's DOT system is linked to Roadway Management, Planning & Programming System, and Maintenance Management System and British Columbia's system is linked to its road inventory management system.

### PERSONNEL REQUIREMENTS

Data collection and condition assessment can be a one-step or two-step process. Data collection consists of recording wall attributes and condition assessment is the evaluation of the wall condition. Most agencies use a two-step process, inventory and evaluation of the walls are done in two separate steps. Once the data collection is performed, the wall is scheduled for inspection if it meets the inventory criteria. Data collection can be performed by surveyors or technicians who will check and complete the field survey forms. Condition assessment must be done by a civil, structural or geotechnical engineer or certified bridge inspector. The offices that are usually responsible for conducting and managing retaining wall inventories and condition assessment are the bridge or structures unit, the geotechnical unit, the maintenance unit and the district offices (Brutus and Tauber 2009).

Findings that are followed by different agencies in terms of retaining wall inventory, inspection and asset management programs are provided in detail in the following sections. Some parts are taken directly from the agency guidelines.

An Asset Management Plan should include the following components according to AASHTO (2011):

- Data management
- Inventory and condition surveys
- Levels of Service
- Service Life
- Performance measures and condition indices
- Risk management
- Life cycle and benefit and costs analyses
- Decision support

An example of processes for a proposed geotechnical Asset Management Plan is presented in Figure 1.1.



Figure 1.1. Proposed processes for a geotechnical asset management plan (from Vessely 2013)

### 1.3.1. Brutus and Tauber (2009) Synthesis Report and recommendations

This publication presents methodologies aimed to help transportation agencies establish asset management programs for earth retaining structures, and specifically Inventory and Inspection (I&I) programs. The programs would then provide the essential information for a broader Asset Management program. A five-year interval for routine inspections is recommended.

A five-point rating scale is suggested as a sample rating system, as follows:

- 1. Excellent: No significant indication of distress or deterioration
- 2. Good: Some indications of distress/deterioration, but wall is performing as designed
- 3. Fair: Moderate or multiple indications of distress/deterioration affecting wall performance.
- 4. **Poor:** Significant distress/deterioration with potential for wall failure.
- 5. Critical: Severe distress/deterioration. Indications of imminent wall failure.

The consequences of failure that are considered in the performance of risk assessment include:

- Death or injury to persons, including facility users and those on adjacent properties or facilities
- Damage to property including vehicles, highway property or facilities, and adjacent property or facilities
- Disruption of highway operations, including full or partial closure of the roadway, or appurtenant facilities
- Disruption of adjacent utility lines, such as water mains or electrical conduits
- Environmental consequences, such as damage to a significant wildlife habitat or blockage of a watercourse
- Damage to cultural assets or sensitive land uses

A three-level consequence of failure rating system is suggested:

 Severe: High likelihood of injuries or death from debris falling on a heavily traveled roadway, on other heavily used adjacent areas, or from collapse of structures near top of wall. High likelihood of extensive or total-loss damage to vehicles or structures. Complete closure of a heavily traveled roadway requiring lengthy detours.

- 2. Significant: Low probability of injury to persons but likelihood of any of the following: (a) substantial property damage, (b) interruption of water or other utility service to a large area, (c) lengthy blockage of access to business properties or public facilities, (d) long-term damage to environmental or cultural resources, (e) closure of two or more lanes of a heavily traveled roadway, (f) full closure of any roadway with no alternative access or requiring lengthy detours.
- **3. Minor:** Low probability of either injury to persons or of damage to vehicles or non-highway property or facilities. Full roadway closures where alternative access is available. Closure of a single lane on a heavily traveled roadway.

It is recommended to collect information from ERSs that are being demolished because of a highway widening or alignment. A careful investigation can yield useful information for the design of future structures.

### **1.3.2.** Minnesota DOT Synthesis Report (2013)

The Minnesota DOT prepared a synthesis report (CTC and Associates 2013) of asset management programs followed by other transportation agencies. Three agencies provided guidance to most of the MnDOT's questions: FHWA for the National Park Service (NPS), Alaska Department of Transportation & Public Facilities (DOT&PF) and Oregon DOT. The main findings can be characterized as:

- Inventory Methods: FHWA relied on maintenance staff guidance to conduct its inventory of retaining walls. Alaska DOT&PF thus far has used only internal records, but in its next phase will recruit technicians to collect data in the field. These technicians will systematically target critical routes and interview district maintenance personnel to find concealed walls. Alaska DOT&PF hopes eventually to use such technologies as Light Detection and Ranging (LiDAR). Oregon DOT uses Google Maps and Bing Maps for visible walls and field visits for others.
- Attributes: Height, length, location, condition and wall type are typical attributes. FHWA suggests keeping data collection simple initially, although it collected data for an extensive number of attributes for its program.
- **Inspection:** Interviewees agreed on five years as an appropriate interval for routine inspection.
- Useful Life: This is a difficult topic that even FHWA is unsure how to manage in its database. Alaska DOT&PF and Oregon DOT have not yet addressed this topic.
- **Performance Measures:** FHWA has data collection forms and libraries in their Wall Inventory Program (WIP). Oregon DOT and Alaska DOT&PF have not yet developed performance measures.
- **Risk Management:** None of the interviewees had conducted an extensive risk analysis.

## **1.3.3.** National Park Service (NPS) Wall Inventory and Condition Assessment Program (WIP) (2010)

The FHWA Wall Inventory Program (WIP) is the best documented wall inventory program in the United States to date. Assessment of 3,500 retaining walls in 32 NPS properties across the US was conducted. The overall performance was very good despite the 60+ year age of the majority of the walls.

Concrete walls (1940s and 1950s era) are showing signs of deterioration, and many needed repair. Corrugated steel bin walls (1960s era), surprisingly used in near-coast applications, were rotting and failing from corrosion. MSE walls were still in good shape for the most part. FHWA has only conducted one inventory and condition assessment, so it doesn't have data on life-cycling. Stone walls were working the best, with more modern wall designs (steel elements) degrading the most and heading toward replacement quickest.

In the WIP database, less than 1% of the walls required replacement or substantial repair. About 3% required replacement of some elements. The bulk of the rest of the maintenance recommendations primarily involved drainage cleanouts, stone resetting/repointing and vegetation removal. No risk analysis has been completed. FHWA examined rate failure consequence, but did not roll up in the wall condition rating.

The best approach to creating a program is to develop a simplified inventory and condition screening method to locate and describe walls (type, size and location) on any given route using the cheapest labor available. The VisiData software was used, which is a program to view data collected along roadways. The VisiData Wall Location Form has information about:

- Road Inventory Program (RIP) route name and/or number
- Side of wall in which the wall is located when travelling in the direction of increasing RIP milepoints
- Approximate VisiData wall start and end milepoints
- Apparent wall function
- Apparent wall type
- Comments regarding wall accessibility, general wall condition, etc.

A team of two members, led by a Geotechnical, Structural or Geological Engineer is responsible for the following during the wall assessment:

- Accurately locate the wall (park, route number/name, milepoint, etc.)
- Describe wall dimensions and features
- Acquire descriptive photos
- Rate the condition of the wall and its key elements, as well as the reliability of the data supporting the wall rating
- Assess if further investigations are required
- Determine the design criteria used to construct the wall (if any)
- Determine the consequences of wall failure
- Determine whether the wall is a cultural resource or not
- Determine the appropriate repair/replace actions (no action/monitor, maintenance, repair element, replace element, replace wall, and/or investigate)
- Develop an appropriate work order, as needed, estimating investigation, maintenance, repair, replacement costs
- Conduct all aspects of the inspection in a manner promoting safety amongst the team and traveling public

The current WIP database was developed as a Microsoft Access application allowing migration to an Oracle platform for database management, rapid queries and future developments.

DeMarco et al. (2010) also recommended a robust location method. FHWA did not use GPS for the NPS program because it is unreliable in many park settings. Instead it used milepoints from Automatic Road Analyzer (ARAN) pavement surveys (which also had roadway GPS). This system worked well, but not without some issues while new ARAN cycles come online.

For the NPS WIP, 65 different attributes were collected to define, quantify, and assess the different variety of ERSs included in its database. As a result, their database application uses three forms for entering data collected during field inspections. The first form contains <u>general</u> <u>descriptions</u> including the ERS's location, function, type, age, facings, and surface treatments. The second form is used to enter <u>condition assessment data</u> for each individual wall element. The third form is used to enter <u>action assessment data</u> such as an overall wall condition rating, a wall

descriptions, and repair recommendations. The forms can be found in Appendix A, while some key parameters can be found below.

### **Re-inspection is based on:**

- > Total asset performance
- Wall type (metal and wire-faced walls need shorter inspection cycles due to deteriorating metal face elements)
- Wall location (walls subject to coastal marine environments, high annual precipitation, extreme freeze-thaw cycles, rapid vegetation growth)
- External event/park request (emergency relief events, landslides, rapidly developing wall failures, recent wall construction in the park)

Maximum of <u>10-year inspection cycle</u> and re-inspection of the total asset if condition rating is less than 70.

### WALL ACCEPTANCE CRITERIA

- Qualifying roads; walls located on paved park roadways and parking areas
- Relation to roadway asset
- Wall height  $\geq$  4 ft; culvert headwalls/wingwalls  $\geq$  6 ft
- Wall embedment; fully or partially buried retaining walls are included in the inventory (e.g. patterned ground anchor walls and buried portions of tieback soldier pile walls)
- Wall face angle  $\geq$  45 degrees (applies to tiered walls considered as single wall system)
- General acceptance; wall protects roadway or parking area and where failure would significantly impact the roadway

### WALL FUNCTION

Refers to the purpose of the retaining structure:

- Fill wall; supports specified soil or aggregate backfill
- Cut wall; supports natural ground
- Bridge wall; wingwalls that continue more than 40 ft beyond the abutment
- Culvert/Head wall;  $\leq 20$  ft total span
- Bridge wall; wingwalls that continue more than 40 ft beyond the abutment
- Culvert/Head wall;  $\leq 20$  ft total span
- Switchback wall; between upper and lower roadway on the inside of a switchback curve
- Flood wall; constructed along flood channels, inland surge walls and seawalls
- Slope protection (e.g. rock buttresses, riprap, stacked rock inlays)

# WALL TYPES

- Anchor (Tieback H-Pile/Soldier pile tieback, Micropile, Tieback Sheet Pile)
- Bin (Concrete, Metal)
- Cantilever (Concrete, Soldier Pile, Sheet Pile)
- Crib (Concrete, Metal, Timber)
- Gravity (Concrete Block/Brick, Mass concrete, Dry stone, Gabion, Mortared Stone)
- MSE (Geosynthetic wrapped face, Precast panel, Segmental block, Welded wire face)
- Soil nail
- Tangent/Secant Pile

# **ARCHITECTURAL FACINGS**

Elements that do not contribute to the support capacity of the structure include brick veneer, cementitious overlay, fractured fin concrete, formlined concrete, plain concrete, planted face, sculpted concrete, shotcrete, steel, stone, simulated stone, stone veneer and timber.

# SURFACE TREATMENTS

Coatings or treatments used to color, preserve or protect wall elements include bush hammer, color additive, galvanization, paint, preservative, silane sealer, stain, tar coatings and weathering steel.

# PHOTOS TO CAPTURE KEY WALL ELEMENTS

- Wall approach
- Wall frontal elevation
- Top of wall/roadway

- Wall face alignment
- Wall face detail
- Wall failure/deficiency detail

# WALL CONDITION ASSESSMENT

Primary and secondary elements are evaluated, first by descriptive condition narratives and then a 1-10 numerical "Element Condition Rating". The overall performance of the wall is evaluated and rated. Wall performance includes global wall distresses and evidence of prior repairs that may indicate component problems. Weighting Factors are applied to the Primary Elements and Wall Performance condition ratings. Once the overall wall performance and pertinent primary/secondary wall elements have been assessed and rated, the inspecting engineer rolls up the weighted element ratings into a "Final Wall Rating". This value ranges from 5-100 and is representative of the overall wall condition.

# **Consequence of Failure**

- Low: No loss of roadway, no-to-low public risk, no impact to traffic during wall repair/replacement;
- **Moderate**: Hourly to short-term closure of roadway, low-to-moderate public risk, multiple alternate routes available; or
- **High**: Seasonal to long-term loss of roadway, substantial loss-of-life risk, no alternate routes available.

# **Recommended Action**

Consideration is given to the Final Wall Rating, any identified requirements for further site investigations, the apparent design criteria employed at the time of the construction, any cultural concerns and the consequences of failure to determine a recommended action:

- No Action: The wall is fully functioning, with no action required at the time of the inspection.
- Monitor: The wall requires regular monitoring and/or investigation to determine the nature of observed distresses and what action may be required.

- **Maintenance**: Routine or cyclic maintenance is required to correct minor or low severity recurring deficiencies spanning a single wall element or the entire structure in order to minimize or delay further wall deterioration.
- **Repair Elements**: Minor to extensive repair of wall element(s) is required in the near-term to prevent rapid element deterioration, loss of performance or failure.
- **Replace Elements**: Replacement of specific wall element(s) or an entire section of wall is required in the near-term to preserve wall stability.
- **Replace Wall**: Replacement of the entire wall structure is required to reestablish the intended function of the wall.

#### **1.3.4.** North Carolina DOT (2015)

This research study (Rasdorf et al. 2015) included a literature review, an identification of ERS data attributes and critical elements of data collection, the development of data collection forms for inventory and condition assessment, the identification of five predominant retaining wall types of greatest interest to the North Carolina Department of Transportation (NCDOT), a study of existing rating systems, a pilot study of 15 geographically distributed ERS locations, the development of a condition assessment system for various retaining wall types, a field application study, and the development of a prototype database. 32 ERSs were inventoried and field surveyed with the database philosophy emphasizing simplicity.

The NCDOT has implemented an integrated asset management system (AMS), which is comprised of Pavement, Maintenance, and Bridge Management Systems and includes Asset Trade-Off Analyst in a single Oracle database utilizing an interactive user interface (Microsoft Access). The NCDOT's Bridge Management System (BMS) has functions that can be applied to ERSs.

Some of the data, such as the retaining wall type, location, and configuration details are static in nature while others, such as geometry are dynamic as the ERS is subjected to tilt, lateral deformation, and differential movement.

A total of 15 ERS sites containing a variation in the distribution of ERSs, with respect to location, retaining wall type, and condition were investigated (MSE, Soil nail, Anchored, Gravity, Cantilever). The data collected was utilized to develop a rating system for a quantitative condition assessment of various retaining wall types.

Two data collection forms were developed: the Wall Identification and Data Attributes Form and the ERS Field Condition Inspection Data Collection Form (Appendix A.2). For the first form, data fields were programmed with drop-down menus in order to minimize errors made by the inspectors. The condition assessment criteria of the second form included four categories: facing, movement, drainage and exterior. There are a total of 17 condition evaluation criteria among the four category observations:

1. **Facial Deterioration:** Missing facing units, spalling, delamination, weathering (splitting or rotting), other deterioration of the wall facing, or graffiti.

2. **Staining:** Discoloration of the facing of the wall from water, efflorescence, rust, or other evidence of corrosion.

3. Damage: Damage to the wall from vehicle impact or root penetration.

4. Cracking: Structural cracking that penetrates the facing of the wall.

5. **Joint Alignment:** Joints between facing units (panels, bricks, etc.) and/or adjacent wall sections that are inconsistent, misaligned, or uneven across the facing of the wall.

6. **Joint Spacing:** Joints between facing units (panels, bricks, etc.) that are too wide (exposing organic material) or too narrow (removing proper spacing).

7. Material Loss: The loss of backfill material through the facing of the wall.

8. **Defection/Rotation:** Wall or parts are visually out of plumb, tilting, or deflecting resulting in a negative or positive inclination beyond the wall's original batter.

9. Bulges/Distortion: Local bulges (outward bend or curve) or distortion in the wall facing.

10. Settlement: Settlement of wall, visible wall elements, or tension cracks behind wall.

11. Heaving: Upward movement or swelling of soil in front of wall.

12. **Erosion**: Disruption or loss of soil or backfill material over a wide area within the sphere of influence of the wall.

13. Scour: Evidence of localized material loss specifically at the wall or around the foundation.

14. **Internal/External Drains:** Evidence of improper passage of water through or over the facing of the wall (i.e., clogged drainage outlets (pipes or weepholes) or drainage channels along top of wall that are not operating properly).

15. **Wall Top Attachment:** Displacement, misalignment, or deterioration (staining, cracking, damage, etc.) of the wall top attachment (Fence or Handrail, Coping, Concrete Barrier Rail, Guardrail, etc.).

16. **Road/Sidewalk/Shoulder:** Cracks, depressions, heaves, and any other evidence of active earth movement within the sphere of influence of the wall.

# 17. Vegetation: Evidence of excessive vegetation on or around the wall.

If the findings from the field condition assessments indicate that an ERS is showing signs of failure, then the additional field evaluations should be undertaken immediately. For critical safety problems, a more detailed means of measurement may include LiDAR measurements taken at some point in time or at other regular intervals as determined by the NCDOT. For noncritical ERSs, LiDAR measurements could be taken only if and when an ERS element is distorted, deflecting, or settling.

To design and develop the WICAS, the North Caroline State University (NCSU) research team used the Microsoft Access database management software tool. A platform such as Sharepoint (web application) can be used in conjunction with MS Access services to create a database that is accessible via the internet. Figure 1.2 shows the main menu of the WICAS, which has six menu options.



Figure 1.2. WICAS home screen (from Rasdorf et al. 2015)

Wall search provides basic information of a wall; location, wall type and condition (Figure 1.3). Every wall is assigned a unique six-digit identification number. The first two digits indicate the county where the ERS is located and the last four digits identify the individual records created and stored in the database for that ERS.

File	Home	Create Externa	al Data 🛛 Databa	se Tools					
Views	Paste VI Clipbe	Cut Copy Format Painter bard G	Al Ascending Al Descending Remove So Sort & F	Selection ~ g	Refresh All~	New Σ Totals Save ♥ Spellir Collecte - ■ More Records	ng <b>(</b>	A Casherice → Golov Find ky Select - Find H G T 型   A - 型 - A - 型 - A - 目 田 - 日 田 - 日 - 日 Find Ky Select - Find T T T - Find Find   田 - 日 - 日 - 日 - 日 - 日 - 日 - 日 - 日 - 日 -	
*	Search For Walls With Inventory And Condition Assessment Data Search by Wall ID, Inspection Date, T.I.P. Number, County, Division, or Route Number: Reset								
	Wall ID:	Inspection Date:	T.I.P. Number:	County:	Division:	Route Number:	*	Wall ID:         910001         Wall Summary         Record ID:         0001	
	910001	5/21/2014	B-3254	Wake	5	US-70			
	910002	5/21/2014	B-3254	Wake	5	US-70		Division: 5 County: Wake Road System Primary Tier: Statewide	
	910003	5/21/2014	B-3254	Wake	5	US-70			
	910004	5/21/2014	B-3254	Wake	5	US-70		Route Number ITS-70 Route Name: Glanumood Auguna Wall Function: Bridge Abutment	
	910005	6/5/2014	R-2113	Wake	5	US-70		Roue Hunter. Cis-10 Roue Hunter. Chenwood Avenue. Wen Function Dirage Administra	
	910006	6/5/2014	R-2113	Wake	5	US-70			
	910007	5/28/2014		Wake	5	I-40		Location Description: South Wall Abutment - US 70 (Glenwood Avenue) over SR 1728 (Wade Avenue)	
	910008	6/5/2014		Wake	5	I-40			
	310009	6/23/2014		Durham	5	I-85			
	310010	6/23/2014		Durham	5	I-85		Wall Type: Anchored [AW] Wall Face: CIP Concrete Veneer: Stone	
	310011	6/23/2014		Durham	5	I-85			
	310012	6/23/2014		Durham	5	I-85			
an	310013	10/10/2014	U-4763B	Durham	5	NC-147		Last inspection Date: 5/21/2014	
- -	310014	10/10/2014	U-4763B	Durham	5	NC-147			
8	780015	10/10/2014	B-3231	Rockingham	7	SR-1378		Cata Wall Investor: Data Cata Condition Assessment Data Resista Homo Seron	
ig.	250016	9/19/2014	U-4756	Cumberland	6	SR-1404		Colo wan inventory Data Colo Condition Assessment Data Back to nome screen	
Se l	250017	9/19/2014	U-4756	Cumberland	6	SR-1404			
	250018	9/19/2014	U-4756	Cumberland	6	SR-1404			
	480019	8/12/2014	R-3833B	Iredell	12	1-77			
	480020	8/12/2014	R-3833B	Iredell	12	I-77			
	480021	8/12/2014	R-3833R	Tredell	12	T_77	-		

Figure 1.3. Wall Search menu (from Rasdorf et al. 2015)

The NCSU research team recommends the same rating approach outlined in AASHTO's "Manual for Bridge Element Inspection". The rating method is very similar to the one for VicRoads Technical Consulting in Victoria Australia (1 to 4 rating). Two inspectors surveyed the 32 ERSs to determine whether or not they could obtain similar average criteria ratings for the same ERSs.

The condition of an ERS is determined by performing field inspections and recording quantities for criteria with defects that correlate to a prescribed condition state (GOOD = 1, FAIR = 2, POOR = 3, and SEVERE = 4). The condition assessment is complete when the appropriate portion of the total quantity is stratified over the defined condition states (e.g., with respect to cracks in the wall facing 25% of the wall may be in FAIR condition and the remaining 75% in POOR condition). As with VicRoads and AASHTO, the sum of the individual condition percentages assigned to each criterion has to equal 100%. Once the appropriated percentages are

assigned to the 17 criteria listed on the "Field Condition Assessment Data Collection Form", they are then used in a weighted averaging process to determine a single value rating for each criterion. Table 1.4 shows the rating definitions and examples of the four condition states.

For example, for the "staining" criteria, the inspector determines that roughly 10% of the entire ERS showed signs of staining corresponding to a "Fair" condition state, 40% in a "Poor" condition state, and 50% in a "Severe" condition state. In this example, the majority of staining (roughly 90%) was deemed to be in a "Poor" or "Severe" condition because the steel sheet piles were severely rusted allowing groundwater to seep through the wall facing. As a result, when all the percentages (by rating) were aggregated together (using a weighted average), the overall criteria rating (average rating) was determined to be 3. The calculation for the "Average Rating" was determined in the following manner:

CATEGORY OBSERVATIONS		PERCENT BY RATING			TING	AVERAGE	COMMENTS
		1	2	3	4	RATING	COMMENTS
	Facial Deterioration	0	0	50	50	4	Severely weathered sheet piles.
	Staining	0	10	40	50	3	The steel sheet piles are severely rusted allowing groundwater to seep through the wall.

**Example Average Rating Calculation**:  $(1 \times 0\%) + (2 \times 10\%) + (3 \times 40\%) + (4 \times 50\%) = 3.4$ 

In accordance with the element condition rating definitions outlined in Table 1.4, this means the distressed "staining" criterion may result in a wall failure without near-term repair or replacement.

Condition State	Description	Example
l "GOOD"	Low Severity Distress The distress does not significantly compromise the wall's function, nor is there significant or severe distress to major structural elements.	A soldier pile wall may have moderately extensive minor surface corrosion on piles where protective paint has weathered and peeled, and may have wood lagging beginning to split. Distresses are very low overall, present over a modest amount of the wall, and do not require immediate or near-term attention.
	An average criteria rating of 1 indicates a criterion that is showing no distress whatsoever or is only beginning to show the first signs of distress or weathering.	
2 "FAIR"	Low-to-Medium Extent of Medium Severity Distress The distress does not compromise wall function, but lack of treatment may lead to impaired function and/or elevated risk of wall failure in the long term. An average criteria rating of 2 indicates a criterion with specific distresses that need to be mitigated in the near-term to avoid significant repairs or replacement in the longer term.	Numerous anchor struts holding MSE wire facing elements in place are beginning to break due to corrosion and suspected over-stressing of the connections at the time of construction. Although the overall function of the reinforced earth wall is not in jeopardy, failing wall facing baskets are allowing facing fill to spill out. If several overlying baskets experience this isolated element failure, significant wall face sag, and deformation may result at the top of wall, eventually impacting the overlying guardrail installation. The element should be inspected carefully along the entire wall and repaired as needed to forestall further facing basket deterioration.
3 "POOR"	Medium-to-High Extent of Medium Severity Distress The distress threatens wall function, and strength is obviously compromised and/or structural analysis is warranted. The criteria condition does not pose an immediate threat to wall stability and roadway closure is not immediately necessary. An average criteria rating of 3 indicates a distressed criterion that may result in a wall failure without near-term repair or	Mortar throughout a stone masonry wall is cracked, spalling, highly weathered, and often missing. Individual stone blocks are missing from the wall face, and adjacent blocks show signs of outward displacement. Although not an immediate threat to overall wall stability, stone block replacement and repointing throughout the wall in the near term are necessary to forestall rapid wall deterioration.
4 "SEVERE "	High Severity Distress The criteria condition is compromising the wall's performance and is threatening the overall stability of the wall at the time of inspection. The wall is in danger of failing, requiring the roadway be closed to all traffic until the wall can be replaced or stabilized. An average criteria rating of 4 indicates a severely distressed criterion that may result in a wall failure.	A 15-ft-tall cast-in-place concrete cantilever wall has a large open horizontal crack running the full length of the wall at the base of the stem. Vertical cracks are also beginning to open up in the wall face. Water is seeping from most wall cracks, and is running from the basal horizontal crack at several locations. The wall face has rotated outward, resulting in a negative batter of several degrees. The overlying guardrail is highly distorted above the wall and the adjacent roadway is showing significant settlement above the retained fill.

# Table 1.4. Proposed condition rating system (from Rasdorf et al. 2015)

Note: This table was adopted and modified from FHWA-CFLHD's, "Retaining Wall Inventory and Condition Assessment Program (WIP): National Parks Service Procedure Manual."

#### **Risk Assessment**

To assist the NCDOT with defining the relationship between qualitative ratings and time sensitive actions, the NCSU research team has developed a risk assessment matrix. In Table 1.5, risk is evaluated qualitatively as a function of both criticality ratings (i.e., whether the consequence of failure (COF) is "High" or "Low") and condition ratings (i.e., whether the likelihood of failure (LOF) is "Very High," "High," "Moderate," or "Low"). Definitions for the COF criteria and the LOF criteria are presented in Table 1.6 and Table 1.7, respectively. Table 1.8 and Table 1.9 present the recommended actions and inspection frequencies as a function of risk.

 Table 1.5. Risk Assessment Matrix (from Rasdorf et al. 2015)

Condition Rating (Likelihood) Criticality Rating (Consequence)	Low	Moderate	High	Very Likely
High	Moderate	Moderate	High	High
Low	Low	Low	Moderate	Moderate

 Table 1.6. Consequence of Failure Criteria Definitions (from Rasdorf et al. 2015)

Criticality Criteria Rating	Definition of Failure Consequence
Low	No threat to people or property. No loss of roadway or impact to traffic during wall repair or replacement.
High	Severe injuries to people or fatalities. Total-loss damage to structures or long-term damage to the environment, cultural resources, or other property. Complete closure (long-term) of heavily traveled roadways.

Condition Criteria Rating	Failure Likelihood Criteria Definitions [FHWA, 2013]
Low	A failure could occur but would require a remote circumstance to trigger failure.
Moderate	A failure could occur but evidence suggests the event could be either unlikely than likely.
High	There is evidence a failure will occur with only a minor triggering event.
Very High	There is significant evidence that failure has occurred or will occur without any further triggering events.

 Table 1.7. Failure Likelihood Criteria Definitions (from Rasdorf et al. 2015)

# Table 1.8. Action Assessment Table (from Rasdorf et al. 2015)

Risk	Actions
High	Remedial action is required in the short term.
Moderate	Remedial action is required in the long term.
Low	No action is required.

# Table 1.9. Inspection Assessment Table (from Rasdorf et al. 2015)

Risk	Inspection Frequency
High	Frequent Inspections.
Moderate	The inspection frequency should be based on the likelihood of failure (LOF). If the LOF is "Low-Moderate," the inspection frequency can be infrequent. Conversely, if the LOF is "High-Very High," inspections should occur more frequently.
Low	Infrequent Inspections.

#### **1.3.5.** Colorado DOT (2003)

Wall management includes the functions of inventory, inspection, condition assessment, maintenance, performance evaluation, and asset valuation. Wall management can be similar to bridge management, and indeed existing data organization and software tools for bridges can be adapted to use for walls.

The CDOT inventory contained 640 retaining walls (110 are MSE walls) and 110 sound barriers at the time; there were around 1250 walls in Colorado. MSE walls are the most common in Colorado.

CDOT defines walls as structures that retain fill and have width of at least 100 ft and height of at least 5 ft. Eight wall types are defined in the CDOT structure number coding guide: Cast-in-Place concrete, MSE, Masonry, Pre-Cast Elements, Pre-Cast Elements prestressed, Tie-back and Others. Wall elements were not established.

CDOT had no program for periodic inspection of walls, no standards and no rating system. The Inventory was for new walls only (built after 1998). The most frequent maintenance activity is removal of graffiti. Failures in walls are rare, and most walls serve with no maintenance beyond cleaning. The Guidance in Governmental Accounting Standards Board Primer (GASB) is used for asset management. A management system such as Pontis was suggested to be adapted to management of walls.

Table 1.10 presents elements and components of walls. Table 1.11 shows some of the wall actions. Table 1.12 presents some observations that should be made in the course of a routine inspection of retaining walls and Table 1.13 illustrates conditions states for elements.

		Quantities	Conditions	Actions
Elements	Feature at Top	Yes	Yes	No (1)
	Top attachment	Yes	Yes	Yes
	Facing	Yes	Yes	Yes
	Bottom attachment	Yes	Yes	Yes
	Feature at front	Yes	Yes	No (1)
Components	Membrane	Yes	No	Yes (2)
	Backfill	Yes	No	Yes (2)
	Fill Reinforcement	Yes	No	Yes (2)
	Anchors/other tensile inclusion	Yes	No	Yes (2)
	Drainage blanket	Yes	No	Yes (2)
	Internal drains	Yes	No	Yes (2)
	Foundation	Yes	No	Yes (2)

 Table 1.10. Properties of Elements and Components (from Hearn 2003)

 A wall management system is not directly concerned with maintenance of adjacent features, but the system will indicate that actions at walls can affect adjacent features.

(2) - Invasive actions are needed to replace or repair internal components.

Action		Walls & Barriers
Maintenance 217.01		Graffiti removal
	217.02	Removal of vegetation
	217.03	Clearing of drains
	217.04	Replacement of riprap or other random slope or channel protections
Repairs 217.11 Repair		Repairs to railings and barriers damaged by collision
	217.12	Sealing cracks in facing elements
	217.13	Patching concrete elements
Rehabilitation 217.21 Replacement of f		Replacement of facing panels
	217.22	Replacement of drains or membranes
	217.23	Replacement of anchors for facing
	217.24	Replacement of fill reinforcements
	217.25	Shoring

# Table 1.11. Maintenance Actions (from Hearn 2003)

Area	Element	Observations
Feature at top	Slope	Note any evidence of movement. May be evident as disruption of vegetation, as depressions or heaves, as scars as slumps or settlements.
		otherwise disturbing the slope.
	Pavement	Note cracks, depressions, heaves, and any other evidence of
		movement, especially note movement along the edge of
		pavement nearest to the retaining wall.
Wall Top	Railing,	Note deterioration wall attachments such as surface rust, minor
Attachment	Sidewalk	cracking, rot in timber, etc. that may require maintenance.
		Note rooting of vegetation that may disrupt wall attachments.
		Note cracking, misalignment, tilting or other evidence of
		movement.
		Note loss of fill or exposure of foundations for wall
		attachments
Wall Facing		Note graffiti on facing.
Barrier Facing		Note deterioration in wall/barrier facing such as surface rust,
		cracking, rot in timber that may require repair of facing
		elements.
		Note evidence of water drainage over the wall, or emerging from the facing in joints or cracks.
		Note vegetation rooted in joints or cracks.
		Note settlement, tilt, or cracking related to movement of wall.
		Observations of water infiltration or movement. Loss of fines
		Other settlement, tilt movement of the walls or barrier.
Wall Bottom	Railing,	Note evidence of movement at cover for toe of wall. May be
Attachment	Sidewalk,	evident as depressions, heaves, scars, damage to pavements,
	Splash	railings or barriers at toe.
<u> </u>	Block	
Sound Barrier	Posts	Note deterioration in posts such as rust and rot. Note any
		collision damage to posts
		Note evidence of movement of posts such as settlement or tilt.
		Note any exposure of foundations for posts, or other distress in
		toundations.

 Table 1.12. Routine Inspection Tasks (from Hearn 2003)

	State	Description
1	Good	No deterioration of elements. No movement, settlement, or misalignment.
2	Fair	Minor, repairable deterioration of elements. Minor movement settlement or misalignment
3	Poor	Deterioration that may require replacement of elements. Significant movement, settlement or misalignment that is not (vet) a threat to safety or stability of the wall.
4	Serious	Deterioration that impairs function of elements. Movement, settlement or misalignment that may threaten to safety or stability of the wall.
5	Critical	Deterioration or movement, etc., that requires emergency shoring, anchoring or closure of lanes adjacent to the wall

 Table 1.13. Condition States for Elements (from Hearn 2003)

# • MSE wall case

Bulging of ramp connecting WB I-70 to NB I-25 was observed. Distress in MSE wall resulted in cracking on the ramp's concrete roadway pavement that was several hundred feet long with sections of differential settlement up to 0.75 in. A bulge was also noticed at the facing of MSE wall. Investigation was done with survey targets and video system to inspect storm drainpipes. This problem, if inspected properly, could have been detected early (from information of movement with time) and the drainage system could have been fixed a long time ago to avoid the significant bulging of the wall that occurred over the last few years.

#### **1.3.6.** Colorado DOT (2016)

This report (Walters et al. 2016) compiles inventory data and provides consistent inspection condition rating and coding guidelines to facilitate management of transportation needs. Inspection data are captured using a tablet (quick and easy) and works together with the online application System for Asset Management and Inspection (SAMI). The Bridge Branch of CDOT manages retaining walls, noise walls and bridge retaining walls.

Criteria for wall definition:

- Noise wall:  $\geq 8$  ft in height
- **Retaining wall:**  $\geq$ 4 ft in height
- **Bridge wall:** The bridge retaining wall is located entirely within the bridge zone as shown in Figure 1.4. The <u>bridge zone</u> is a rectangular boundary created by measuring **40** ft perpendicular from the edge of the bridge on either side and 40 feet perpendicular from the face of the abutment (or abutment wall) along the approaches on either end of the bridge.



Figure 1.4. Bridge zone (from Walters et al. 2016)

One end, or corner, of the wall starts at the face of the abutment of the bridge and the entire length of the wall face is no greater than **200 ft**. Walls that fit this criterion should be inspected

as a single bridge wall to prevent the creation of additional smaller walls. Figure 1.5 shows the Effective bridge zone.



PLAN VIEW

#### Figure 1.5. Effective Bridge Zone (from Walters et al. 2016)

For a wall that begins within the bridge zone, but extends beyond the effective bridge zone, the wall should be separated into two walls at the 40-foot bridge zone mark. Figure 1.6 shows the <u>Separation of wall at bridge zone.</u>



PLAN VIEW

Figure 1.6. Separation of wall at bridge zone (from Walters et al. 2016)

A wall that is associated with two different bridges whose abutments are less than or equal to 200 ft apart should be inventoried as a single bridge wall.

Wingwalls or headwalls for bridges or culverts are not inspected. These structural elements are considered part of a bridge or culvert and would be addressed in CDOT's Routine Bridge or Minor Structure Inspections Programs.

#### Routine Inspection

- Retaining and Noise walls: <u>6 years</u>
- Bridge walls: <u>4 years</u>

The **System for Asset Management and Inspection (SAMI)** is composed of two interdependent pieces; mobile and in-office. Field data is collected using a mobile tablet device, and then uploaded to a web-based database at the end of each inspection day. The mobile unit collects photographs as well as location, condition, and appraisal data in accordance with the guidelines described in this manual. Once this data is uploaded to the web-based database, SAMI can be used to generate and submit reports, analyze data, budget, and schedule inspections inoffice. Figure 1.7 shows the "Structures Map" in the mobile SAMI application including line geometry, associated photos and defect locations. The "Elements" tab provides a list of appropriate defects based on the element type, which reduces any errors when an inspector notes defects associated with an element.

The inventory categories for wall structures are: (1) Structure identification, (2) Location and (3) Structure data. The condition ratings for the structure level inspection follow closely the language found in the "Recording and Coding Guide for the Structure Inventory and Appraisal of the Nation's Bridges", published by the US DOT (1995). A similar 0-9 scale, as seen in the Condition Ratings section should be used to rate the condition of the Main Wall Facing, Foundation, and Channel and Channel Protection. This evaluation data assists in the calculation of Condition Risk Rating for the wall, and should be recorded in the "Overall Structure Rating" field in the Inspection Report. Materials from the Colorado DOT can be found in Appendix A.3.



# Figure 1.7. Structure Map in SAMI application (from Walters et al. 2016)

Condition ratings are used to describe the existing, in-place wall as compared to the as-built condition. A scale of 0-9 is used to rate the condition of the Main Structure, Foundation, and Channel and Channel Protection.

Wall element inspection: Primary elements, Secondary elements, Incident elements

Maintenance categories: Do nothing, protect, repair, rehabilitate, replace

The defects are classified into two categories, National Bridge Element (NBE) Defects or Agency Defined Element (ADE) Defects, denoting the origin of the condition state language.

Defect condition rating: Good, Fair, Poor, Severe

Environmental Factors: Benign, Low, Moderate, Severe

## Wall Elements

- **Primary Elements:** Primary Elements are the main structural features of a wall, including the <u>wall face</u>, vertical supports, foundations, and anchors. They are subject to distress and deterioration and are the most important features rated during the wall condition assessment.
- Secondary Elements: Secondary Elements include the attachments, appurtenances, and surrounding features that can impact the performance of the wall. They exhibit a lower degree of distress and deterioration from the Primary Element and are also rated during the wall condition assessment. Secondary Elements include <u>coping, drainage elements</u>, <u>architectural facings, protective coatings, slopes and backfill, railings, and joints</u>.
- **Incidental Elements**: Some elements are considered incidental to primary or secondary wall elements and the location of these incidental elements should be noted under the general description of the parent element. If an incidental element is damaged or deteriorated, an appropriate work candidate should be created to address the issue.

## **Common Wall Structures**

- MSE segmental block retaining wall
- Precast panel MSE retaining wall
- Cast-in-place cantilever retaining wall
- Cantilever, Soldier Pile
- Post-and-Panel noise wall
- Free-standing noise wall

#### **Material Defects**

- Corrosion of material elements
- Cracking in material elements
- Connection distress in material elements
- Delamination/Spall/Patched area in material elements
- Exposed rebar/Welded wire/Fabric/Strands in concrete elements
- Exposed prestressing steel in concrete elements
- Cracking in prestressed concrete elements

- Efflorescence/Rust Staining in concrete and masonry elements
- Decay/Section loss in timber elements
- Check/Shake in timber elements
- Split/Delamination in timber elements
- Abrasion/Wear in material elements
- Deterioration in elements such as fiber reinforced plastics
- Mortar breakdown of masonry mortar
- Split/Spall in stone
- Masonry displacement
- Distortion from original line or grade of the element
- Bulging of wall facing elements
- Vertical rotation of elements
- Horizontal rotation of elements
- Separation of wall facing elements
- Graffiti on wall element
- Vegetation growth
- Blockage of drainage elements
- Effectiveness-Anchors
- Freeze-Thaw damage
- Leakage through/around sealed joints
- Loss of seal adhesion
- Seal damage
- Seal cracking
- Debris impaction
- Metal deterioration/Damage
- Joint material; deterioration, missing, loose or other defect
- Chalking in metal/concrete/masonry protective coating
- Peeling/Bubbling/Cracking in metal/concrete/masonry protective coatings
- Oxide film degradation color/Texture adherence in metal protective coatings
- Loss of effectiveness of metal/concrete protective coatings

- Wearing of concrete/timber protective coatings
- Effectiveness of internal concrete/timber protective system
- Backfill loss
- Water retention
- Erosion of any material adjacent to the wall
- Settlement in foundation elements
- Scour in foundation/facing elements
- Impact damage
- Temporary support on wall facing/vertical support/foundation
- Alkali-Silica reactivity

# CDOT retaining wall asset management plan

Considers the seven National Performance Areas under MAP-21/FAST legislation:

- Safety
- Infrastructure condition
- Congestion reduction
- System reliability
- Freight movement and economic vitality
- Environmental sustainability
- Reduced project delivery delays

# Consequences initially considered for plan development:

- Condition loss and damage to the wall
- Safety hazards to travelling public
- Potential traveler delay, congestion and mobility impacts
- CDOT maintenance expenses for wall repair
- Impacts to environmental resources
- Economic loss to users
- Private property damage

The risk analysis in the plan can be performed at multiple plan levels ranging from **qualitative** (subjective) levels of accuracy for higher level plan decisions to more rigorous **quantitative** (numerical) evaluations for specific wall assets. In both instances, the risk process can assign values to various conditions, the extent of infrastructure vulnerability and the measures used to manage adverse consequences.

#### Goals of risk-based asset management plan

- Shift the process towards a more proactive approach with a long-term view of the overall health of the statewide wall system and develop a multi-year investment plan that will support wall assets to optimize life-cycle costs.
- Routine maintenance activities intended to preserve wall assets and slow deterioration rates to obtain the anticipated life-cycle
- Wall system as a statewide asset class with long-term financial plans to maintain the system and reduce risk statewide

The risk-based asset management plan identifies walls with a high risk to mobility and economic consequences to provide CDOT the opportunity to manage risks using a **lowest life-cycle cost approach**. CDOT uses wall condition data from wall facing repair plans to manage similar deterioration conditions with a lower life cycle cost goal. A failure example was considered in the development of the plan approach in order to give CDOT the ability to identify, prioritize, and invest in mitigation efforts than can prevent a larger adverse event with economic consequences that exceed the required investment for preventive rehabilitation.

#### Areas of interest for risk-based management plan

- detailed inventory and appraisal of each wall asset
- internal CDOT operating costs to maintain the wall structure
- user costs associated with wall maintenance and/or adverse events

#### Risk concept for wall assets

Multiple plan tiers are used to prioritize wall assets on the basis of higher risk:

- 1. **Tier 1 Level**: inventory of retaining walls and subjective determination of likelihood and consequence of potential failure
- 2. **Tier 2 Level**: based on data collected during field inspections, assessment of risk to mobility and maintenance
- 3. **Tier 3 Level**: higher risk walls and cost-benefit analysis to determine preferred investment strategy; continued monitoring, rehabilitation or replacement

**Tier 1 Risk Score** = [Wall Condition] x [Failure Consequence x AADT Factor x Height Factor]

Low: risk score of 8 or less Medium: risk score between 8 and 12 High: risk score of 12 or higher

Tier 2 Maintenance Risk Score = Weighted Maintenance Risk Costs/Raw Maintenance Costs

Primary elements have greater priority and are weighted more heavily.

<u>Roadway Impact (RI) score</u>: estimates the potential mobility consequence associated with the wall structure.

RI = (Avg. Wall Height – Distance from Roadway In Front) +

(2 \* Avg. Wall Height – Distance from Roadway Carried)

User Costs = (Delay Time/3600) x (AADT Actual – AADT During the Delay) / 2)/24 x User Value x Occupancy Rate

- Delay time is assumed to be 2 hours for all walls
- AADT during any delay is assumed to be 33 percent of actual AADT
- User cost value = \$30.50 per hour
- Occupancy rate = 1.67 per vehicle

The **user costs** represent the consequence estimate in the determination of mobility risk. The likelihood (or probability) of an event based on the condition score is based on input from CDOT and consultant staff and reflect past experience and professional judgment.

The final mobility risk calculation is found as:

# Tier 2 Mobility Risk Score = User Costs x Wall Condition

645 walls have been inspected to date and their data were used to complete Tier 2 analysis.

# **Tier 3 Plan Level**

Targets of the wall management program:

- 95% of walls with a Level of Risk grade of C or worse
- Less than 1% of walls with CS4 (Condition State 4) defects at or above the C level
- Less than 1% of walls with deterioration accelerator condition states

<u>Performance target:</u> reduce and maintain number of walls with a level of risk of D or F to 5% of the total wall inventory.

#### **1.3.7.** Wisconsin DOT (2017)

Wisconsin DOT started a wall inventory program in 1999. Retaining walls with height greater than 5 ft should be inspected at intervals not to exceed 4 years. Four main types of retaining walls include: Rigid or Gravity, Cantilever, Anchored and MSE.

#### Structural Material Failure

Primary causes of in-service retaining wall failures include poor drainage, corrosion, facing deterioration, inadequate connection details, and latent construction defects. Failure of the construction material is frequently observed in older earth retention structures due to deterioration. Newer walls may exhibit structural material failure due to structural overstresses or poor material properties. Inadequate drainage behind the wall or an unexpected surcharge load can often cause material overstress. Impact damage may also fail the material, and is typically a result of a collision between a moving object and the earth retention structure.

#### Geotechnical Failure

Vertical movement: soil bearing failure, soil consolidation, erosion and foundation material deterioration

Lateral movement: slope failures, seepage, changes in soil characteristics and consolidation.

Rotational movement: saturation of backfill due to clogged drains, embankment erosion along the front of the wall and improper design.

Table 1.14 shows common material flaws observed during retaining wall inspections.

Stone/Concrete Masonry Units	Concrete	Timber	Metal	Other Materials
Construction defects	Cracking	Decay	Corrosion	Ultraviolet deterioration
Corner breaks	Scaling	Insect Infestation	Cracking	Material incompatibility
Cracked block	Fraying/Spalling/Exposed Reinforced Steel	Vermin damage		Corrosion damage
Efflorescence	Abrasion	Fire damage		Overstress damage
Embedded				
vegatative growth				
Abrasion				
Fraying/Spalling				
(Block edges)				
Freeze-thaw				
damage				
Manufacturing				
flaws				
Popouts				
Positioning-Guide				
damage				
Scaling				
Staining				
Structural Distress				
Wash Through				

## Table 1.14. Common material defects (from Wisconsin DOT 2017)

Only the visible features of the wall including the front face (facade), top, and sides of a wall will typically be inspected during a normal routine inspection. It is the inspector's duty to discern from distress through the observable components if other unseen issues are at play.

# **Recommended Inspection Procedures**

- 1. Arrive at site and set-up traffic control (if required)
- 2. Identify Structure Number
- 3. Perform Inspection

- Check wall for signs of settlement, rotation or bulging
- Inspect the vertical alignment of the wall with a plumb-bob. (Note: Most walls are constructed with a battered or sloped face)
- Examine the opening of the construction joints between sections of the wall
- Inspect joints near ground line for any fill material washing out from between panels or joint
- Inspect for erosion of the embankment material in front of the wall
- Inspect for heaving of the embankment material in front of the wall
- Inspect for settlement of the fill material behind the wall
- Examine the wall for deterioration of the material, such as cracking, spalling, and/or corrosion, noting the width, length, depth, and/or orientation of the deterioration
- Some wall types (post and panel) may require the inspector to randomly select a few posts and dig down 3-6 in below ground line to see if piling is deteriorating at the soil level
- Lagging or cribbing should be checked for excessive deflections. Excessive deflections may allow the solid behind to spill or wash out, causing settlement in the retained material above
- Examine previous areas of repair for soundness
- Check wall façade for evidence of water seepage, efflorescence or rust staining
- Examine anchorage systems if present. Fasteners and connections to the wall components should be checked for tightness and distress.
- Examine and probe drains for signs of clogging. Examine drainage around ends of wall and note if embankments have been experiencing erosion.
- Examine site grading for any locations that may prohibit proper drainage from behind the wall. Look for evidence of ponding above the wall, such as debris accumulation in the lower spots. Attempt to ascertain why water is not draining properly, and note in the inspection.
- Inspect sidewalk or roadway components above wall for signs or joint separation, potholes and areas of settlement.
- Examine vegetation growth along and above the wall. Root infiltration may create undesirable stresses on the wall and may induce cracking, bulging or failure.
- Examine the wall system for vehicular damage. Document the location and degree of damage.
- Note previous inspection frequency and recommend inspection frequency

- 4. Determine and record the overall rating of structure based on inspection findings
- 5. Determine and record all applicable maintenance items and a level of priority
- 6. Determine if an underwater dive inspection or an in-depth inspection needs to be scheduled to supplement the routine inspection and provide more information on the condition and performance of the wall. If determined to be needed, schedule in the HSI System
- 7. Review of inspection notes to ensure completeness and correctness
- 8. Document all CS3 and CS4 defects with a photo and/or a sketch
- 9. Remove any traffic control

Components are divided into Elements (Primary) and Assessments (Secondary). Both require that the inspector quantify specific conditions states, but Elements take it a step further and also require that the inspector define Defects specific to each Element for asset management purposes. WisDOT simplifies wall inspection into the following Defects: Wall deterioration, Wall movement, Masonry or Panel displacement and Scour.

The rating system is based on a four level scale: Good, Fair, Poor and Severe:

- *Good*: No, or very low distress observed in the wall elements and assessments. Defects are minor, and within the normal range for newly constructed or fabricated elements. Highly functioning wall that is only beginning to show the first signs of distress or weathering.
- *Fair*: Overall, the condition is satisfactory. Distress is present in wall elements and/or assessments, but does not compromise the wall function. Localized drainage issues, settlement, staining, washing of fines from backfill material that are minor.
- *Poor*: Overall condition of the wall is poor. Distress is present, but does not pose an immediate threat to wall stability and closure of facilities adjacent to structure is not necessary. Repair and/or replacement is needed in the near future.
- Severe: Critical condition. Major structural defects, or components have rotation, sliding, settlement, and/or overturning that is close to possible collapse. Wall is no longer serving the intended function, or is unstable and needs repair/replacement as soon as possible. Facilities adjacent to wall may need to be closed.

Upon completion of a wall inspection, the inspector is tasked with assigning an overall condition rating to the structure. This is a global evaluation and is used to determine inspection frequency

and other asset management functions. Therefore the inspector must take into account all elements and assessments noted during the inspection and the functionality of the entire structure.

Determining the priority level for each item depends significantly on how the functionality of the wall is impacted. Table 1.15 describes the repair timeline associated with each priority level. The inventory data sheet that is used by the Wisconsin DOT is provided in Appendix A.4.

 Table 1.15. Repair timeline depending on priority level (from Wisconsin DOT 2017)

<b>Priority Level</b>	Timeline Expectations
Low	Repair prior to next inspection, as funding allows
Medium	Repair within one year as funding allows
High	Repair within 90 days
Critical	Repair within the timeline specified by the inspector in the notes, but not to
	exceed 30 calendar days

#### 1.3.8. Alaska DOT (2013)

At present, the Alaska Department of Transportation and Public Facilities (AKDOT & PF) is in the first phase of the process for developing an inventory and inspection system which involves a survey of internal records to catalog ERS locations and gather basic information (CTC and Associates 2013). In the second phase, AKDOT & PF plans to validate and augment their inhouse data with data collected in the field. As part of the second phase, the AKDOT & PF will also rely on the guidance of FHWA-CFLHD to develop an ERS condition assessment process and establish a rating system to measure ERS performance.

Criteria for determining if a wall should be inventoried:

- Serves as an earth retention structure
- Belongs to a roadway asset that is owned and/or maintained by AKDOT&PF
- Culvert headwalls/wing walls  $\geq 6$  ft (total height, exposed plus embedded)
- Face angle  $\geq 45^{\circ}$
- Identify tiered wall system as one wall

Wall information is to be entered into a GIS-based database via a web interface. This preliminary phase of the Wall Inventory relies on gathering information from internal AKDOT&PF records, including-but not limited to- the following:

- As-built
- Road viewers, e.g. Google
- AKDOT&PF Digital Roadway Viewer Alaska
- DOT Highway Data Port
- Bridge Inventory/PONTIS
- Statewide Culvert Inventory
- Compilation of Bids (COB) sheets

Figure 1.8 shows the bridge zone limits.



Figure 1.8. Bridge zone limits (from AK DOT & PF 2013)

#### 1.3.9. Pennsylvania DOT (2010)

The inspection typically consists of an examination and recording of signs of damage, deterioration, movement, and if in water, evidence of scour.

PennDOT takes a tiered approach, with a "routine" wall inspection every five years and an "indepth" inspection (which includes a three-dimensional survey for MSE walls more than 100 ft long and more than 20 ft high) at either 10 or 15-year intervals. Unscheduled "special" inspections are to be performed after a significant event, such as a vehicular collision, extreme weather, or indication of wall movement.

Neither reinforcement nor backfill can be inspected; therefore, a close visual inspection of the facing panels and drainage facilities is required to provide information on all three of the major components. This includes visual inspection of the roadway surface (i.e., pavement) above the MSE wall for tension cracking. Inspection of the leveling pads, if visible, can provide information on scour, erosion or settlement. Inspection of the barriers can also provide important information regarding movement of the MSE wall.

#### **Field Inspection Procedures**

Many of the techniques from the bridge inspection are also applicable to retaining wall inspections. Establishing a baseline condition for retaining walls is crucial for effective future inspections.

- Inspect exposed wall faces, barriers and moment slabs, footings and joints for: arching, spalling, movement of joints, corrosion of members, locations of entrapped water/improper drainage, evidence of impact, condition of riprap, and/or indications of scour.
- Inspect wall for movement, rotation or settlement.
- Inspect crest of sloping backfill for evidence of soil stress or failure as an indication of settlement or wall movement.
- Inspect drainage facilities in the wall and in proximity of the wall (above and below the wall) to ensure proper function of drainage.

## MECHANICALLY STABILIZED EARTH WALL FIELD INSPECTION PROCEDURES

The critical factors affecting the long term performance of MSE walls are: corrosion of the soil reinforcement, improper drainage, improper backfill material and compaction, freezing of entrapped water, and movement of the entire MSE Mass (global stability).

Mechanically Stabilized Earth (MSE) retaining walls should be inspected for evidence of wall movement.

- Examine barrier and moment slab for evidence of movement as well as the MSE wall for evidence of bulging, bowing or panel offset.
- Perform a survey if movement is suspected to compare to initial inspection data to gauge amount of movement.
- Examine the roadway above MSE walls for indications of failing pavement or tension cracking. These may indicate a loss of fill.
- For MSE walls in front of sloping backfill, the crest of the embankment should be investigated for soil stress or failure, both of which may indicate settlement or wall movement.

The joints between panels of MSE Walls are to be inspected and examined for loss of backfill, change in spacing, and indications of settlement. The specification requirement for joint spacing is a maximum <sup>3</sup>/<sub>4</sub> in.

- Inspect walls for evidence of backfill loss (piles of aggregate at the base of the wall).
- Indicate visibility of backfill or fabric behind the panel through joints.
- Examine for evidence of damage to the geotextile fabric, if visible.
- Look for variation in joint spacing. Note vegetation growing in joints.
- Vertical slip (expansion joints) used on long lengths of walls should be investigated similar to panel joints. The initial spacing at the slip joint should be determined from design, shop or as-built drawings.

**Wall panels** shall be checked for cracking, spalling, other forms of deterioration, and collision damage.

**Drainage systems** through or along MSE walls should be inspected to verify water is free flowing into and out of the appropriate facility.

- Ensure that weep holes are free draining.
- Inspect all inlets to verify water is draining into the inlet, and flowing freely to the inlet and out of the outlet. Examine inlets for cracks.
- Inspect visually or use down-hole cameras (as appropriate) for all culverts and pipes contained or having portions in, behind, or above the MSE wall mass and for pipes or culverts which run above, adjacent to, or outlet through the MSE walls to verify pipes are free draining and water is flowing through (and not under or around) the pipe. Examine drainage pipes for cracking or damage with emphasis on areas where water may flow, or is flowing, into the MSE wall soil mass. Inspect outlet ends to verify free drainage or for evidence of migration of fill or other material.
- Inspect swales above the MSE wall. Verify rock fall or other materials (trees, etc.) are not blocking, redirecting, or restricting the flow of water through the drainage ditch above the MSE wall to the appropriate receptacle.
- Inspect collection and outlet basins to verify water is draining freely. Look for any signs of infiltration or migration of material which may prevent water from draining from the wall.
- Identify inappropriate appearance of water along the base of the wall (i.e., if water is appearing when weather conditions have been particularly dry). Note areas where there is inappropriate collection and/or lack of drainage for water along the length of the MSE Wall.
- Note erosion of soil along the base of the wall exposing or undermining the leveling pad.

Observed conditions are translated into ratings, as shown in Table 16, that are assigned to MSE wall elements:

- Anchorage
- Backfill
- Wall conditions such as bulging, joints, deterioration of face panels, connection of the backs, etc.
- Panels

- Drainage
- Foundation
- Parapets

# Table 1.16. Performance ratings assigned to wall elements in Pennsylvania DOT inspectionprocess (from Pennsylvania DOT 2010)

Rating	Rating Definition
8	Good condition. No apparent problems.
6	Satisfactory condition. Structural elements sound. Localized drainage problems, settlement, staining, washing of fines from backfill material.
4	Poor condition. Localized buckling, deteriorated face panels, joint problems, major settlement, ice damage.
2	Critical. Major structural defects, components have moved to point of possible collapse.
### 1.3.10. Nebraska DOR (2009)

This project (Jensen 2009) developed a condition rating system for MSE walls, in which rating criteria are specific to each element or wall condition. The MSE wall features that are assessed are:

- Wall tilting
- Structural cracking
- Facial deterioration
- Bowing of the wall
- Panel staining
- Exposure of fabric
- Loss of backfill
- Erosion in front of wall
- Erosion in back of wall
- Joint spacing
- Condition of "v-ditch" (i.e., drainage way at top of wall)
- Coping deterioration
- Drainage runoff
- Drainage at the front of the wall

A **rating scale** ranging from 0 to 9 (consistent with most bridge assessment procedures) is provided to describe the extent or severity of each feature. Rating criteria are specific to each element or wall condition rather than being generic. The database will then use the numbers from each inspection to create a Wall Performance Index (WPI) that allows NDOR to rate how each retaining wall is performing.

#### MSE Wall ID Numbering Convention



1. Indicates the type of wall present. "M" indicates a MSE wall, while "R" indicates a retaining or gravity wall.

2 Indicates the highway number associated with a specific wall. The MSE wall(s) form(s) one lateral boundary of this highway. Four spaces are provided. If the highway number is a 2 digit number, add two "0" as placeholders before the highway number (as illustrated above for I-80).

3. These five digits indicate the Reference Post location closest to the center of the wall.

4. The last digit is used to designate which side of the highway the wall is located on when the inspector is facing in the direction of ascending reference posts.

## 1.3.11. British Columbia Ministry of Transportation - Canada (2013)

The British Columbia Ministry of Transportation has both expanded its Bridge Management and Information System (BMIS) and revised its maintenance specification to include ERSs.

Condition Rating: Excellent, Good, Fair, Poor, Very Poor

### Wall Components and Conditions

- Hydrotechnical
  - > Channel bed conditions, Channel bank conditions, Skew, Adequacy of waterway
  - Erosion Protection
  - Substructure Scour

### • Foundation

Wall Foundation

### • Structure

- ➢ Movement of Wall
- Retaining Wall
- Embankment
- Tiebacks/Connectors
- ➢ Wall drainage system
- ➢ Coating
- ➢ Railings
- Roadway Flares

Examples of **instrumentation** used to monitor walls include slope indicators (drill holes with special casing installed), tell-tales (simple movement monitors across cracks in walls), surface monitoring hubs, piezometers (drill holes for monitoring groundwater levels), tilt meters, etc. Retaining wall inventory and inspection forms used by the British Columbia Ministry of Transportation are provided in Appendix A.5.

### 1.3.12. Victoria Highway Department - Australia (2014)

VicRoads Technical Consulting of Victoria, Australia, records its inspection data in a road asset system (RAS), which is an information system for all structures' inventory and condition data managed by the Network and Asset Planning Division (VicRoads, 2014). They produced the "Road Structures Inspection Manual" which applies to retaining walls, visual walls, and noise walls, along with many other roadway structures including bridges and culverts.

**<u>Retaining Wall:</u>** A structure with the primary purpose of retaining soil that is 1.5 m or greater in height and steeper than 1 to 1.5 or a structure that would result in a traffic or pedestrian hazard or damage to neighboring property upon failure. A retaining wall within 5 m of a bridge abutment is to be considered part of the bridge structure. Beaching (rock, paved or other) on cuttings or embankments where the slope of the cutting or the embankment is less than 1 to 1.5 is not considered to be a retaining wall but the 'surfacing' of the roadside.

**Level 1 – Routine Maintenance Inspection:** it is used to check the general serviceability of a structure and to ensure the safety of road users. Level 1 inspections may be completed in conjunction with routine road maintenance. Structures are subjected to two inspections per year.

**Level 2 – Structure Condition Inspections:** they are used to assess the condition of structures and their components. Level 2 inspections are managed on a statewide basis. The frequency of inspections is 2 to 5 years depending on the condition.

**Level 3** – **Engineering Investigations:** they are detailed engineering investigations and assessments of individual structures which are conducted <u>as required</u>. Frequency of investigations shall be determined for each structure and shall require ongoing review depending upon the performance, intensity of loading, rate of deterioration, if any, maintenance, strengthening, component replacement or similar that potentially influence safety and whole of life costing. Level 3 investigations may include non-destructive testing and/or sampling of materials for laboratory testing.

Critical components for retaining walls are shown in Table 1.17.

## Table 1.17. Critical components for retaining walls (from VicRoads 2014)

Critical components for retaining walls	
Retaining wall facing components including masonry blocks or bricks, precast concrete panels, timber sleepers and similar	<ul> <li>Evidence of lateral tilting, bulging that might indicate excessive earth pressure from settlement of contained fill, failure of drainage system, if any, behind the wall or similar</li> <li>Extended cracks through mortar and masonry components or precast concrete panels</li> </ul>
Supporting components including vertical posts and columns, metal or geo-synthetic anchor strips for reinforced soil walls and similar	<ul> <li>Evidence of lateral tilting, bulging that might indicate failure or extension of retaining systems in reinforced soil walls, or failure of vertical retaining posts or their foundations</li> <li>Evidence of movement of retaining walls that has caused permanent closure of expansion joints, spalling of concrete superstructure or substructure components</li> </ul>
Strip footing and pile foundations	<ul> <li>Signs of substantial settlement or rotation</li> <li>Significant exposure by erosion, settlement or other means</li> </ul>
Crib wall	Disintegration of blocks

The wall chainage is the distance measured from the Road Start. The General Location is either on the freeway or an adjacent ramp. GPS readings are required at the start and end of the wall together with the chainage at the start of the wall.

The following photographs are required:

- At the start and end of the wall
- A view along the wall
- Any components in condition states 3 or 4

Monitor inspections consist of non-destructive inspections of specific components to detect structural distress that may indicate reduced strength and include:

- Visual observation at arms-length and assessment of the condition of critical components
- Photography in order to compare the condition of critical parts of the structure with previous records

Routine maintenance activities on bridges and other structures are presented in Table 1.18.

# Table 1.18. Routine maintenance activities on bridges and other structures (from VicRoads2014)

Routine maintenance activities on br	idges and other structures
General	<ul> <li>Graffiti and other damage caused by vandalism</li> <li>Accident, fire or water damage</li> <li>Accumulation of dirt, bird and animal droppings and other debris on components preventing drainage, ponding, rusting of steel, seizure of bearings and other moving parts</li> <li>Vegetation growth in structural joints, mortar joints, cracks and other locations on and around structures</li> </ul>
Drainage	<ul> <li>Blocked scuppers and side entry pits on bridges, culverts and approaches</li> <li>Scour or settlement of bridge abutment batters, road approaches, behind retaining walls, foundations of bridge piers, sign structures, mast arms, retaining walls and other structures</li> <li>Blocked weepholes and signs of water penetration and inadequate drainage behind retaining walls, bridge and culvert wing walls and similar</li> </ul>
Deck joints	<ul> <li>Debris blocking or jamming joints</li> <li>Damaged waterproofing seals</li> <li>Missing or damaged bolts</li> </ul>
Bearings	<ul><li>Debris and dirt build up around bearings</li><li>Rusted steel bearings</li></ul>
Barriers	<ul> <li>Damaged, corroded and missing posts, rails, spacer blocks, and connections</li> <li>Approach barrier not constructed or connected</li> </ul>
Bituminous surfacing on structure roadway, footpaths and approaches	<ul><li>Uneven surface</li><li>Settlement of approaches</li></ul>
Signs, lighting and roadmarking	<ul> <li>Missing, damaged or corroded components, supports, connections</li> <li>Signs or roadmarking not legible</li> <li>Lights not working</li> </ul>
Waterways	<ul> <li>Blocked with debris and vegetation</li> <li>Scour and subsidence requiring beaching or other maintenance</li> </ul>

Retaining walls are generally made from timber, concrete, masonry and steel materials. Problems can occur with the foundations of the wall due to:

- Rot and termites in timber particularly at or just below ground level
- Corrosion in steel and cracked welds
- Reinforcement corrosion

- Cracking and spalling concrete
- Cracking of mortar or stone degradation in masonry walls
- Settlement, sliding or overturning of the wall
- Insufficient or ineffective weep holes to relieve pore pressure behind the wall

The inspector should also observe and record problems associated with ground movement that may be exerting unusual pressure on the walls.

### **Condition rating of components**

ERS condition assessments are divided into four individual elements: the wall facings or panels (measured by area), the column and horizontal supports (measured by unit), the foundations or supports (measured by length), and the hold-down bolts, base plates, and fittings (measured by unit). The ERS rating is then established by evaluating each individual element and assigning a conditional percentage to the portion of the element that meets the criteria in one of the four conditional states listed below. For example, if the facing of a concrete retaining wall is 100 meters long and has a 10 meter crack, 90% of the facing would be considered condition 1 and 10% would be condition 3. The sum of the individual condition percentages assigned to each element has to equal 100%. The approach used in this rating system closely resembles the 1-4 rating system outlined in AASHTO's "Manual for Bridge Evaluation" for its bridge element ratings (AASHTO, 2010a).

The manual includes a retaining wall structure condition inspection sheet and a condition rating system for retaining wall elements, including facing panels (area), column supports (unit), foundation (length) and connections (unit). In general, the condition ratings have been developed to describe the following conditions:

Condition state 1 Component is in good condition with little or no deterioration.

**Condition state 2** Component shows minor deterioration with primary supporting material showing the first signs of being affected. Intervention points for maintenance are generally as follows: Minor spalls or cracking of no real concern. Paintwork on steel components with spot rusting up to 5%.

**Condition state 3** Component shows advancing deterioration and loss of protection to the supporting material which is showing deterioration and minor loss of section. Intervention points for maintenance are generally as follows: Large spalls, medium cracking and defects should be programmed for repair works. Paintwork has spot rusting of up to 10%, which is the approximate limit for overcoating.

**Condition state 4** Component shows advanced deterioration, loss of effective section to the primary supporting material, is not performing as designed or is showing signs of distress or overstress. Intervention points for maintenance are generally as follows: Very large spalls or heavy cracking and defects should be repaired within the next 12 months. Paintwork beyond repair requires blasting back to bright metal and recoating.

The extent of each condition state affecting a component shall be measured as a percentage of the whole component. The percentages in each condition state (1, 2, 3 and 4) must add up to 100% of the whole component.

Each element is quantified as follows:

- Number of units making up the component (each)
- Length of the component (lin m) or
- Area of the component  $(m^2)$

When assessing condition rating, the inspector should first determine the worst condition affecting the component (e.g. Condition 4) and its extent, and progress to the best condition (e.g. Condition 1). The condition rating and its extent, for each element shall be recorded as a percentage of each condition state in the appropriate column on the Condition rating sheet. The quantities of each element and their condition are not required unless specified elsewhere. The accuracy of the percentages determined for each condition state shall be within  $\pm 5$  %.

For example, if the facing of a concrete retaining wall is 100 m long and has a 10 m crack, 90% of the facing would be considered condition 1 and 10% would be condition 3.

Materials from the Victoria Highway Department can be found in Appendix A.6.

### **1.3.13.** Kentucky Transportation Cabinet (1998)

In this study (Fleckenstein et al. 1998), approximately 209 walls were visually evaluated for long-term performance. The inspection included concrete crib, single-barrel and double-barrel culvert wing, metal bin, gabion, rigid concrete, keystone block, tiedback, mechanically stabilized earth, TechWall, and sound walls. Significant structural distress was observed in several of the wall systems. Table 1.19 shows the walls that were inspected in this study. Each wall system was divided into nine sections for analysis purposes. In addition, each wall system was photographed and video taped.

Several walls that were evaluated had significant problems and should be repaired. It was recommended that retaining wall structures be inspected annually by maintenance crews. It is apparent from this study that drainage plays a major role in the long-term performance of these structures. Past edge drain research indicates that 20 to 50 percent of edge drain outlets were not fully functional. It is likely that these percentages also were applicable to bridge-end drains. It was recommended that bridge-end drains be inspected during construction. It was also recommended that a full-scale study be conducted on the performance of bridge-end drainage and to evaluate the lateral earth pressures on return walls.

Materials from the Kentucky Transportation Cabinet can be found in Appendix A.7.

### 1. Concrete Crib Walls

Several types of distress were observed in the crib wall systems including: 1) migration of backfill through the crib members, 2) displaced crib members due to erosion, 3) cracking of the crib members, 4) spalling of the face of crib members, 5) slight bulging or tilting, and 6) sluffing of unanchored ends. The significance of the observed distress was ranked from "A" being slight "B" being moderate and "C" being severe.

The migration of the backfill through the cribbing appears to be the most significant problem observed in the crib wall systems. A non-erodible granular backfill should be used to prevent the

migration of the backfill, and the surrounding soil, through the crib structure. If the backfill is classified as erodible or unstable it shall be protected by geotextile fabric.

### 2. Culvert Wing Walls

Each wall was divided into ten different sections to help quantify repeated distress patterns. Wall systems have performed well. The only distress that was recurring was slight-to-moderate cracking and staining.

Spalling occurring on the horizontal face of the top slab of the culvert was likely due to water saturating the top of the culvert. A possible solution to deter spalling on the horizontal face would be to cover or seal the top horizontal face with a waterproof geomembrane. Maintenance crews should periodically check the inlet and outlet ends of the culvert for buildup of debris or vegetation, and take any necessary actions to clean up obstructions blocking the culvert.

Туре	No.
Concrete crib walls	4
Wing walls - double barrel culverts	32
Wing walls - single barrel culverts	20
Metal bin walls	5
Gabion walls	5
Rigid concrete retaining walls	13
Rigid concrete abutment (breast)	8
Rigid concrete abutment (non vertical)	5
Rigid concrete wing walls	18
Rigid concrete approach retaining walls	7
CIP concrete wall tied back	6

 Table 1.19. Kentucky wall inspection study (from Fleckenstein et al. 2003)

Туре	No.
Timber lagging tied back	4
Keystone modular block retaining wall	2
Reinforced Earth Co. Open bridge abutments	9
Reinforced Earth Co. Wing walls	18
Reinforced Earth Co. Closed bridge abutment	13
Reinforced Earth Co. Return walls	23
Reinforced Earth Co. Retaining walls	3
VSL retaining walls	2
TechWall ramp embankment	2
Sound barriers - Brick	1
Sound barriers - Metal sheet	1

### 3. Metal Bin Walls

Each wall type was divided into nine sections for inspection and analysis purposes. Significant distress was observed in the center of a wall. Five of the vertical support members had failed at the base of the wall, and bending was occurring in some of the horizontal members.

Failures in the vertical supports of the metal bin retaining wall are likely due to post construction settlement of the backfill. Further inspection and monitoring for vertical settlement were recommended at the time.

### 4. Gabion Walls

Bulging and/or sagging was observed in most of the walls. Consideration shall be given to placing a geotextile between the Gabion wall and the fill material if the fill material cannot be retained on the 100 mm (4-inch) sieve-the smallest size stone used to fill the Gabion baskets.

## 5. Rigid Concrete Retaining Walls

The majority of the distress observed with each wall type consisted of slight to moderate cracking or staining. The shrinkage cracks may be controlled by using more expansion and contraction joints.

Concrete walls used for bridge approach fill retaining walls have suffered the most severe recurring distress among the five different wall categories. Consideration is to be given to reinforcing the backfill of approach fills, and wall drains are to be placed with care. To insure proper drainage at bridge ends, the wall drains are to be inspected with a pipeline camera during the construction phase.

### 6. Modular Block Retaining Walls

Distress included cracking, sliding, bulging, migration of backfill between blocks and settlement behind the structure.

## 7. Tied Back Walls

Of the ten tied back walls inspected in this study, the group that had a cast-in-place wall in front of the tied back wall has performed the best. The two tied back walls that had used clips to hold the timber lagging in place were in need of immediate maintenance. This method of construction is not to be used in future tied back wall construction. The method of placing timber lagging between the webs of the H-Piles appears as a good alternative to using the clips. However, the tied back walls that used this type of construction were relatively new and were to be monitored further.

### 8. MSE Walls

Several problems were noted with the seventy-seven MSE walls. Settlement of the leveling pads was noted in several of the inspected MSE wing walls, which may be attributed to poor soil compaction. Cracking in MSE wall panels on return walls and bridge abutments had developed due to precast panels or cast-in-place blocks located between the bridge deck and the MSE wall. In future MSE wall construction, these precast panels or cast-in-place blocks are not to be used to bridge the two structures. Drainage was an additional problem that was noted during inspection of MSE bridge abutment and return walls. Water is to be diverted away from the reinforced approach fill. In areas where wall drains were installed in the approach fill, a miniature pipeline camera should be used to inspect the integrity of drainage system during construction. Also, water is not to be discharged from headwalls directly behind any retaining wall structure. Lastly, geotextile fabrics should be used in MSE wall construction.

## 9. Techwalls

There were two TechWalls inspected for this study. The TechWall is one of the accepted alternative walls selected by the Kentucky Department of Highways. However, this particular wall type has not been used very frequently in Kentucky. The two TechWalls that were inspected had already shown signs of distress. Newly accepted walls such as the TechWalls, and other retaining walls are to be thoroughly monitored during construction.

### **1.3.14.** City of Seattle (2014)

The City of Seattle has more than 500 retaining walls in its inventory. Most of these retaining walls were built prior to 1970. Some were built as early as 1900. Early 1990s, Roadway Structures made a preliminary inspection of all retaining walls in its inventory. After this inspection, the retaining walls were grouped into **4 categories**:

- 1. Retaining walls that are in good condition
- 2. Retaining walls that need minor to major maintenance.
- 3. Retaining walls that need frequent monitoring.
- 4. Retaining walls that need replacement.

Later in 1999 and 2000, all retaining walls **4 feet high and above** were inspected by a consultant (Agra). A **digital protractor** was used to measure plumpness of retaining walls.

The major types of retaining walls, which are widespread throughout the City, are as follows:

- Rockfacing
- Soldier pile wall with timber or concrete lagging
- Cantilevered concrete walls
- Gravity type concrete walls
- Crib-lock wall (concrete, timber or steel)
- Gabion walls
- MSE Retaining Walls

Inspection of retaining walls may vary depending on the type of retaining wall, its condition and the area where it is located. If the retaining wall is in an area of **steep slope** or/and in an area with **slide history**, the scope of inspection may include the <u>visual reconnaissance</u> and <u>assessment</u> of <u>apparent slope stability</u> of the adjacent area. If the condition of the retaining wall is bad and needs frequent monitoring, it may require **establishing points of reference** (base-line) for checking rate of deterioration.

### **Procedure of inspection:**

- 1. Pick an area and select retaining walls for inspection
- 2. Check out files for the selected retaining walls and review previous inspection report

- 3. Review appropriate plans in the vault for details of retaining wall.
- 4. Take out all the tools you need for inspection:
- 25 feet Tape
- 100 feet tape
- Camera
- Digital protractor (to be provided by SEATRAN)
- Hammer
- Bondo (patching paste)
- Crack comparator
- Machete
- 5. Assess condition of retaining wall. The following points can be used as a minimum guide line in condition assessment
- Check alignment and geometry and note if bulges, differential settlement, or differential tilting are apparent.
- Check rotation and compare with its original batter angle. Some of the retaining walls have built-in tilt measurement reference points. Using a digital protractor, measure tilt on the existing points and compare results with the previous measurement. For previous tilt measurement and location of reference points, refer to retaining wall file. If the retaining wall does not have tilt measurement reference points, establish a new one. For concrete retaining walls, use **anchor bolts**. For steel posts, make direct reading on the face of the **soldier piles**; indicating location of measurement. For timber posts, use survey tacks. Generally, tilt measurement is not required for retaining walls less than 6 feet in height subject to the inspector's judgment. It is not easy to put tilt measurement reference points on some retaining walls such as rockery or gabion. Tilt will not be measured for such walls. All new installation of reference points should be at accessible locations. Most of our previous installation was 5 feet above the ground and sometimes measured from the top.
- Assess slope stability. Look at the slope areas adjacent to the wall; make note of any indications of slope movement such has cracking or settlement at the top or heaving at the bottom which may indicate movement of the retaining wall.

- Check for cracks. Measure sizes and map their location. Assess cracks to determine if they are caused by thermal movement, shrinkage or other structural problems. If crack monitoring is required, clean a small area on the crack and patch with Bondo. Other crack monitoring methods can also be used if approved by SEATRAN.
- Check for any concrete spalling or delamination. Spalls with exposed rebar should be mapped and their sizes (area and depth) noted. Areas of delamination should also be mapped with approximate areas.
- Note if retaining wall has weep holes or other means of relief of subsurface water pressure.
- Check other miscellaneous defects such as rust, rot, damage, erosion, paint, rock pockets, weathering, root pressure, etc.
- 6. Record all findings. Include **sketches** if necessary to describe deficiency and location of defects of each component of retaining wall.
- 7. Take **pictures** that indicate the vicinity and close detail of the defect.
- 8. Write condition report for each wall that documents all information and other collected data. Report should include condition <u>rating</u> that illustrates the relative condition of each retaining wall. For consistency with previously made inspection report, we recommend the following guidelines be used in condition rating of each retaining wall:
- 9. Discuss major structural problems that affect the integrity and functionality of the retaining wall with Roadway Structure's Engineer before request for repair is made.

Each retaining wall is given <u>% condition rating</u> that numerically portrays its relative condition. Each defect is rated from **0 to 100**. If the wall has distinct components such as rails, walers or tiebacks, each defective component is also rated. Each rating is multiplied by a <u>weight factor</u> to get factored condition rating. Weight factor prorates the rating of each defect based on its role to the structural integrity of the retaining wall. Weight factor varies from **0 to 1**.

The overall % condition rating of retaining wall is calculated using the following formula:

$$\frac{i=n}{\sum (WF) G}$$

$$I=1 \quad i \quad i$$

$$\frac{i=n}{\sum (WF)}$$

$$i=n$$

$$\sum (WF)$$

$$i=1 \quad i$$

Where  $\mathbf{G} = \text{rating} (0 \text{ to } 100)$ 

WF = weight factor (0 to 1)

 $\mathbf{N} =$  number of defects rated

### **Rating guide table**

### Tilt measurement

Tilt in °	0-1	3-5	6-10	11-15	16-20	> 20
Rating (G)	100	90-100	70-90	40-70	20-40	0-20

<b>Condition rating</b>	Description of rating	Rating
81-100	Good condition	Good
51-80	Fair condition	Fair
31 to 50	Poor to condition	Poor

## Weight Factors

Weight factors are subjective and it is up to the inspector to decide. The table below can be used as a guide to help the inspector pick a number within the described range.

Deficiency	Tilting of wall	Slope movement	Bulging of wall	Crack	Differential Settlement	Rot	Rust
WF	1	1	1	0 to 0.7	1	0.5 to 0.8	0.5 to 0.8

The Microsoft-ACCESS based application program is used to keep inventory and inspection data for the retaining walls.

## **Project Assumptions**

The scope of work for this wall inspection is based on the following assumptions:

- 1. Assumes most of the retaining walls require tilt measurement;
- 2. Assumes some walls, especially new ones need monitoring points installed;
- 3. Assumes steel surface is uniformly flat and tilt reading can be taken without making reference points, such as dent.

4. Assumes segmental walls (gabion, rockery, ecology block or concrete crib) do not require tilt measurement.

Wall attributes that are collected during inspection are shown in Appendix A.8.

### 1.3.15. The City of Cincinnati (2015)

The Wall Stabilization and Landslide Correction Program is the specific program within The Department of Transportation and Engineering (DOTE) charged with the responsibility of maintaining the retaining walls within this transportation system. Every retaining wall within or adjacent to the right-of-way has been inventoried and is included in the Wall Inventory Tracking System (WITS) database; around 7,000 walls were inventoried with 5,125 being privately owned walls.

The Wall Inventory Tracking System (WITS) was created in 1991 and used Foxpro as the database software. In 2015, WITS was converted to a server based Oracle system and is accessed through ArcGIS. A wall inspection application was created that allows the inspector to complete the inspection documentations using a tablet in the field, eliminating the need to complete hand written documents and then reentering the data into a computer file.

Walls are inspected once every six years. Walls that are rated 3 (Poor) or 4 (Critical) condition are inspected yearly. Other inspections are performed if a wall is damaged in an automobile accident or if a complaint is received. All newly constructed, replaced or repaired walls are inventoried and inspected.

The rating system follows a five-level scale:

**0 to 1 Excellent:** No-to-very-low extent of very low distress. Defects are minor, are within the normal range for *newly constructed or fabricated* elements, and may include those resulting from fabrication or construction. Ratings of 0-1 are only given to elements with very minor to no distress whatsoever –conditions typically seen only shortly after wall construction or substantial wall repairs.

**1 to 2 Good:** Low-to-moderate extent of low severity distress. Distress does not significantly compromise the element's function, nor is there significant severe distress to major structural components. Ratings of 1 to 2 indicate highly functioning wall elements that are only beginning to show the first signs of distress or weathering.

**2 to 3 Satisfactory:** High extent of low severity distress and/or low-to-medium extent of medium to high severity distress. Distress present does not compromise element function, but lack of treatment may lead to impaired function and/or elevated risk of element failure in the long term. Ratings of 2 to 3 indicate functioning wall elements with specific distresses that need to be mitigated to avoid significant repairs or element replacement in the longer term.

**3 to 4 Poor:** Medium-to-high extent of medium-to-high severity distress. Distress present threatens element function, and strength is obviously compromised and/or structural analysis is warranted. The element condition does not pose an immediate threat to wall stability. A rating of 3 to 4 indicates marginally functioning, severely distressed wall elements in jeopardy of failing without element repair or in need of repair to prevent further deterioration at an accelerated rate.

**4 Critical:** Medium-to-high extent of high severity distress. Element is no longer serving intended function. Element performance is threatening overall stability of the wall at the time of inspection. In practice, a rating of 4 indicates a wall that is no longer functioning as intended, and is in danger of failing.

### **Retaining Wall Inspection Criteria**

Each item is rated: 0 = No Problems 1 = Minor Problems 2 = Moderate Problems 3 = Severe Problems 4 = Critical Problems N/A = Not Applicable Each Division – Given an Average Rating (Sum of Individual Items in Division / # of Items rated)

### **OVERALL WALL RATING (General Condition)**

Sum of the average of the four Divisions (Structural, Drainage, Cosmetic, Misc.) Excludes any N/A Ratings

Example:

Structural Avg. - 2.0

Drainage Avg. – 3.0

Cosmetic Avg. – 2.0

Miscellaneous Avg. N/A

Overall Wall Rating = 7 (Sum of Ratings) / 3 (Number of Subjects) = 2.3 (Rating)

## 1.4 MSE Wall Issues and Guidelines

Gerber (2012) compiled a synthesis report with the objective to determine methods used at the time to monitor, assess and predict the long-term performance of MSE walls, where "long-term" denotes the period of time from approximately one year after the wall is in service until the end of its design life (75 to 100 years). Of the 52 US and 12 Canadian targeted survey recipients, 39 and 5, respectively, responded. Fewer than one-quarter of state-level transportation agencies in the United States have developed some type of MSE wall inventory beyond that which may be captured as part of their bridge inventories. The agencies reported the most significant lessons learned, with the more popular topics being drainage, construction, backfill and modular block issues. An important conclusion of this synthesis is that there exists a need for greater recognition of MSE walls (and retaining walls in general) as important infrastructure assets. The 14 respondents who have MSE wall inventories are:

- Alberta, Canada
- California
- Colorado
- Kansas
- Minnesota
- Missouri
- Nebraska
- New York
- North Carolina
- North Dakota
- Ontario, Canada
- Tennessee
- Utah
- Wisconsin

Ohio, Pennsylvania and Washington also have MSE wall inventories, however they were not survey respondents. Some wall inventories are also maintained by city-level agencies. The cities of Cincinnati, New York City and Seattle maintain retaining wall inventories, including MSE walls. Methods used to manage MSE wall inventories include simple spreadsheets, MS Access, Oracle and PONTIS databases.

Most agencies monitor their MSE walls in response to known incidents or adverse performance. It appears that once MSE wall inventories are initially developed, additional information relative to ongoing performance is generally either not collected or not assessed for most walls. The reinforcement of the retained soil mass, which can be geosynthetic material or metallic straps or meshes, is critical for the assessment of the performance of MSE walls. The AASHTO (2010b) metal loss model and backfill specifications are shown in Table 1.20. Several US State agencies have conducted reinforcement corrosion studies.

Table 1.21 presents a summary of various of these efforts. Corrosion monitoring of steel reinforcement is typically accomplished by either retrieval of buried coupons or non-destructive electrochemical methods. For geosynthetic reinforcement, the primary performance issue is polymer degradation. At present, the only effective means of assessment is retrieval of buried specimens. Table 1.22 depicts backfill material requirements for different DOTs.

Metal Loss Model		Backfill Specifications			
Component Type (age)	Loss (µm/yr)	рН	5 to 10		
Zinc (first 2 years)	15	Resistivity	$\geq$ 3000 ohm-cm		
Zinc (to depletion)	4	Chlorides	$\leq$ 100 ppm		
Carbon steel (after steel depletion)	12	Sulfates	$\leq$ 200 ppm		
depiction		Organic Content	$\leq 1 \%$		

## Table 1.20. AASHTO metal loss model and backfill specifications (after Fishman and Withiam 2011)

### Table 1.21. Summary of US State MSE wall corrosion assessment programs (from Gerber

2012)

# Table 1.21. Summary of US State MSE wall corrosion assessment programs (from Gerber2012)

State	Description
California	Has been installing inspection elements with new construction since 1987, and has been performing tensile strength tests on extracted elements. Some electrochemical testing of in-service reinforcements and coupons has also been performed. Linear polarization resistance (LPR) and EIS tests were performed on inspection elements at selected sites as part of NCHRP Project 24-28 and results compared with direct physical observations on extracted elements.
Florida	Program focused on evaluating the impact of saltwater intrusion, including laboratory testing and field studies. Coupons were installed and reinforcements were wired for electrochemical testing and corrosion monitoring at 10 MSE walls. Monitoring has continued since 1996.
Georgia	Began evaluating MSE walls in 1979 in response to observations of poor performance at one site located in a very aggressive marine environment incorporating an early application of MSE technology. Exhumed reinforcement samples for visual examination and laboratory testing. Some in situ corrosion monitoring of in-service reinforcements and coupons at 12 selected sites using electrochemical test techniques was also performed.
Kentucky	Developed an inventory and performance database for MSE walls. Performed corrosion monitoring including electrochemical testing of in-service reinforcements and coupons at five selected sites.
Nevada	Condition assessments and corrosion monitoring of three walls at a site with aggressive reinforced fill and site conditions. Exhumed reinforcements for visual examination and laboratory testing; performed electrochemical testing on in-service reinforcements and coupons. A total of 12 monitoring stations were dispersed throughout the site providing a very good sample distribution.
New York	Screened inventory and established priorities for condition assessment and corrosion monitoring based on suspect reinforced fills. Two walls with reinforced fill known to meet department specifications for MSE construction are also included in program as a basis for comparison. Corrosion monitoring uses electrochemical tests on coupons and in-service reinforcements.
North Carolina	Initiated a corrosion evaluation program for MSE structures in 1992. Screened inventory and six walls were selected for electrochemical testing including measurement of half-cell potential and LPR. This initial study included in-service reinforcements, but coupons were not installed. Subsequent to the initial study, NCDOT has installed coupons and wired in-service reinforcements for measurement of half-cell potential on MSE walls and embankments constructed since 1992. LPR testing was also performed at approximately 30 sites in cooperation with NCHRP Project 24-28.
Ohio	Concerned about the impact of their highway and bridge de-icing programs on the service life of metal reinforcements. Performed laboratory testing on samples of reinforced fill but did not sample reinforcements or make in situ corrosion rate measurements.
Oregon	Preliminary study including (1) a review of methods for estimating and measuring deterioration of structural reinforcing elements, (2) a selected history of design specifications and utilization of metallic reinforcements, and (3) listing of MSE walls that can be identified in the ODOT system.
Utah	Extracted 22 wire coupons from one- and two-stage MSE walls all approximately 11 to 12 years old. Galvanization thickness was found to still be greater than initial specified values. Data to provide baselines for future assessments.

After Fishman and Withiam (2011).

## Table 1.22. Summary of backfill requirements for some State DOTs (from Raeburn et al.2008)

State Name	PI or <b>Φ</b>	Resistivity, R	Chlorides	Sulfates	рН
		(ohm-cm)	(ppm)	(ppm)	-
Federal	-	≥3000	≤100	≤200	5 to 10
Highway					
(FHWA)					
California	$PI \le 10$	≥1500	<500	<2000	5.5 to 10
Florida	PI < 6	≥3000	$\leq 100$	≤200	5 to 10
Georgia	-	≥3000	≤100	≤200	6 to 9.5
New York	PI < 5	≥3000	≤100	≤200	-
Ohio	$\Phi > 34$	≥3000	$\leq 100$	≤200	5 to 10 for steel
					reinforcement 4.5 to 9 for
					geosyn. reinf.
Washington	-	≥5000	Waived	Waived	5 to 10 for steel
		3000≤R<	$\leq 100$	≤200	reinforcement 4.5 to 9 for
		5000			geosyn. reinf.
Idaho	PI < 6	≥3000	Waived	Waived	4.5 to 9.5
			≤100	≤200	
Nevada	PI < 6	≥3000	≤100	≤200	5 to 10
Utah	<b>PI</b> < 6		$\leq 100$	≤200	6 to 9
	$\Phi > 34$				
Colorado*	$\Phi > 34$	≥3000	≤100	≤200	5 to 10
Oregon**	PI < 6	≥5000	Waived	Waived	4.5 to 9.5
		3000≤R<	$\leq 100$	≤200	
		5000			
Related	ASTM	AASHTO	AASHTO	AASHTO	AASHTO
Standards	D4318	T-288-91	T-291-91	T-291-91	T-289-91
	for PI		or ASTM	or ASTM	
			D-4327-88	D-4327-88	

\* In their specs, CDOT requires backfill material to be non-aggressive, hence FHWA criteria is assumed.

\*\*ODOT also requires the uniformity coefficient (Cu=D60/D10) of the granular backfill material to be smaller than 1.5 (ODOT 2007).

The most important conditions in assessing the long-term performance of MSE walls are corrosion and degradation of internal reinforcement and drainage according to the survey participants. Wall height, was surprisingly found to be among the least important features. In addition, most agencies believe that global stability and reinforcement rupture are the most likely failure modes for MSE walls in the long term. Overturning and facing failure are considered the least likely failure modes.

Once wall conditions are assessed, the assigned rating can be related to a specified action or is used to make programming decisions. For example, in the FHWA-NPS (DeMarco et al. 2010) system four additional items are considered before taking action: (1) are additional investigations

required (how reliable is our assessment); (2) what design criteria may have been used in planning the structure (was the structure engineered); (3) what aspects of the wall structure are historic or contribute to the cultural context of the road asset; and (4) what are the consequences of wall failure.

Alzamora and Anderson (2009) provided a list of design and construction inspection issues and commented that poor performance of MSE walls is mainly attributed to construction, but observations related to design, wall and material selection and mitigation of weak foundations have been reported:

### Design

- Geometry/Wall layout
- Obstructions
- Wall embedment
- Surface drainage
- Contractor experience

## **Construction/Inspection**

- Claims
- Backfill placement and compaction
- Panel joints
- Leveling pad
- Durability of facing

Texas DOT has had few issues with MSE walls, which are the majority (85%) of the walls in the State of Texas (Delphia 2012). The main MSE wall can be categorized as:

## • DESIGN

- ➢ Global Stability
  - Strength Conditions (short term and long term)
  - Presence of Water

Placement of Walls on Slopes, may be false economy

### • CONSTRUCTION/INSPECTION

- Embankment/Backfill Placement
- Foundation Soil preparation
- Obstructions Drilled shafts, drainage features.
- Damaged Reinforcements
- Connections
- Backfill Properties
- Panel alignment and panel joint spacing
- Loss of MSE backfill
- ➢ Corner details
- ▶ Hard point under MSE retaining wall leveling pad

It is recommended that the construction of the wall should be treated as a structure and not as an embankment. Texas DOT developed a shallow inlet standard 1'-10'' in depth. When inlets have to be placed behind the wall, Texas DOT uses the vertical stand pipe option and the shallow inlet standard.

Ohio DOT (Narsavage 2006) inspected 339 MSE walls. Figure 1.9 shows some of the observed problems that were identified. The Office of Structural Engineering would use the information fr om the inspections to develop an inspection program, similar to the Bridge Inspection Program. Condition ratings were developed consisting of simple "yes" or "no" responses. Inspection forms can be found in Appendix A.9.



Figure 1.9. Observed problems in MSE walls (Narsavage 2006)

## **Observed Problems**

- Sand leaking from joints
- Settlement of panels
- Uncontrolled drainage
- Deteriorating panels
- Erosion along MSE wall
- Vegetation in joints
- Bulging
- Drainage system

## **Design Changes for MSE Walls**

- Preference for certain MSE wall types. Acute angles should be avoided at bridge abutments and obstructions should be minimized.
- Abutments supported on spread footings only under certain conditions. All obstructions within reinforcing zone should be shown in a plan view.
- Drainage around MSE walls should be considered. The barrier should be extended past the MSE wall and catch basins should be placed beyond the MSE wall.
- Avoid utilities through or underneath MSE walls.

Utah DOT tried to locate all the MSE walls in the State and develop an inspection protocol based on the Ohio DOT inspection program, as well as an asset management database (Bay et al. 2010). Data from 104 MSE walls are contained in the database.

A team from Utah State University and another one from Brigham Young University inspected MSE walls in the State of Utah. The inspection form that was developed is separated into: 1) inspector and general information, 2) summary of key observations, 3) plan/drainage view, 4) cross-sections and 5) specific wall characteristics. The specific wall characteristics include eight categories: water and drainage, wall joints, wall facing, conditions at top of wall, foundation conditions and external stability, corrosion, impact and collision, and miscellaneous. Questions are answered as yes, no or unknown, while there is a choice of percentage of the wall exhibiting a characteristic. Inspection forms are provided in Appendix A.10.

### Observations

- Drainage at top/bottom of the wall
- Blocked drains
- Permanent water flows and runoff
- Internal drainage (leakage through panel joints)
- Irrigation pipes
- Vegetation (evidence of drainage, can disrupt components of MSE wall)
- Irregular panel spacing and movement
- Popped panel corners
- Cracked panels
- Bowing in wall
- Tears in fabric
- Cracks in coping (indicative of wall movement)
- Cracks in parapets
- Leveling pad issues (exposure, cracking
- Corrosion/Erosion of panels
- Salt deposits

- Corrosion testing coupons
- Reverse batter
- Large deformations/Steep slopes at the wall base
- Presence or absence of bench at base of wall founded on a slope (reduces erosion)
- Presence of adjacent structures (can move along with the wall)
- Block wall issues (spacing, leakage, exposure of reinforcement, corrosion)

Walls are numbered with R-numbers (e.g. R-123) and tracked in a spreadsheet. A software Transportation Asset Management System (TAMS) with an open-source GIS-based interface between MapWindow and Microsoft Access was implemented to create an electronic database from field observations. The walls are identified by the symbol "1", "2" or "B" for one-stage, two-stage and modular block wall, respectively. A unique spatial identifier, usually a GPS point, is entered for each wall record.

## Issues of concern identified by panel of experts

Short-term performance of MSE walls

- Wall drainage
- Wall design, details and specifications
- Retention of wall information

Long-term performance of MSE walls

• Corrosion of the retaining wall systems

A brief overview of some case studies and the mechanisms that had as a result the wall failure are provided below:

## 1. Soda Springs, Idaho

The wall was built in 1978 and was considered the first true MSE wall abutment in the United States. Failure occurred in 2002; six pre-cast concrete panels popped out.

<u>Cause of failure:</u> corrosion of the metallic soil reinforcing strips attached to the lower concrete facing panels. A chemically aggressive slag waste was used as backfill.

### 2. Susquehanna County, Pennsylvania

Built in 1985, failure in 1989. Cavity of 10  $\text{ft}^2$  opened up in the lower right quadrant of east abutment, loss of 70 yd<sup>3</sup> of backfill and pre-cast concrete panels toppled away from the wall.

Cause of failure: overstress of the reduced section in the critical bottom third of the wall.

### 3. Rockville, Maryland

Built in 1996, failure in 2003; geosynthetic reinforcement was used. Large gaps and separation in the wall facing blocks started in 2002. At failure, leaning and bulging of the wall face was observed and the top of wall moved 12 to 18 in. Portions of the wall face were wet even after periods of dry weather; indicative of water flow through backfill. Excavation revealed that the geogrid reinforcement was not horizontal. The compaction of the upper layer of fill was not sufficient (81% of  $\gamma_{dmax}$ ).

<u>Cause of failure:</u> improper installation of the geogrid, inadequate internal drainage of the backfill soil and inadequate compaction of the reinforced soils.

#### 4. Clearfield, Utah

Built in 2001, distress observed same month the bridge was open to traffic. Vertical separation between wall panels was reported along with undermining and displacement of MSE wall leveling pads, rotation of wall panels with accompanying spalling of several panel corners, outward rotation of the top MSE walls panels and overlying coping, cracking of the MSE wall coping, outward rotation/translation of barrier sections, cracking and displacement of the roadway pavement section near the southwest corner of the bridge, horizontal and vertical displacement of the sidewalk along the north side of the west approach embankment and development of voids beneath the sidewalk, and beneath the roadway slabs. Slope movement was in the order of 1.5 ft.

<u>Cause of failure:</u> the bench was not constructed with a full 4-ft width in all locations as per AASHTO specifications. There were changes in the embankment material and insufficient

compaction. Erosion from runoff through cracks in the pavement or around the end of the barrier.

### 5. Salt Lake City, Utah

Built in 1997, failure in 2005. Buckling of MSE wall panels was observed. The panels were repaired but movement continued within the reinforced backfill zone.

<u>Cause of failure:</u> internal settlement of the wall backfill immediately next to the face of the wall and insufficient compaction of the soil.

### 6. Orem, Utah

Built in 2001, failure in 2003. Large void developed behind the back of the MSE wall and exposed the flat reinforced straps and their connection to the panels.

<u>Potential Cause of failure:</u> poor drainage and subsequent erosion or internal erosion and settlement behind the wall panel.

### 7. Northeastern Tennessee

Built in 1999, failure 9 months later. Geogrid reinforcement was used. Movement and deformations in spread footings and nearby electrical duct manholes. No global stability analysis was performed prior to construction of the wall, which would have revealed the necessity of more and elongated geogrid reinforcement.

Cause of failure: inadequate reinforcement and use of clayey backfill.

### 8. Southwestern Virginia

Built in 1999, failure 1 year later. Geogrid reinforcement with clayey backfill was used. The reinforced soil remained intact, while the geogrid reinforcement had pulled from between the masonry blocks. A car wash was located directly above the wall. Cracking in the ground surface behind and parallel to the wall. Compaction of reinforced soil was insufficient.

<u>Cause of failure</u>: poor drainage led to buildup of hydrostatic pressure behind the wall. Erratic compaction of fill soils, increased height of wall and use of clayey backfill within reinforced zone.

A team from Utah State University and another one from Brigham Young University inspected MSE walls in the State of Utah. The inspection form that was developed is separated into: 1) inspector and general information, 2) summary of key observations, 3) plan/drainage view, 4) cross-sections and 5) specific wall characteristics. The specific wall characteristics include eight categories: water and drainage, wall joints, wall facing, conditions at top of wall, foundation conditions and external stability, corrosion, impact and collision, and miscellaneous. Questions are answered as yes, no or unknown, while there is a choice of percentage of the wall exhibiting a characteristic. Inspection forms are provided in Appendix A.

## Observations

- Drainage at top/bottom of the wall
- Blocked drains
- Permanent water flows and runoff
- Internal drainage (leakage through panel joints)
- Irrigation pipes
- Vegetation (evidence of drainage, can disrupt components of MSE wall)
- Irregular panel spacing and movement
- Popped panel corners
- Cracked panels
- Bowing in wall
- Tears in fabric
- Cracks in coping (indicative of wall movement)
- Cracks in parapets
- Leveling pad issues (exposure, cracking
- Corrosion/Erosion of panels
- Salt deposits
- Corrosion testing coupons

- Reverse batter
- Large deformations/Steep slopes at the wall base
- Presence or absence of bench at base of wall founded on a slope (reduces erosion)
- Presence of adjacent structures (can move along with the wall)
- Block wall issues (spacing, leakage, exposure of reinforcement, corrosion)

Walls are numbered with R-numbers (e.g. R-123) and tracked in a spreadsheet. A software Transportation Asset Management System (TAMS) with an open-source GIS-based interface between MapWindow and Microsoft Access was implemented to create an electronic database from field observations. The walls are identified by the symbol "1", "2" or "B" for one-stage, two-stage and modular block wall, respectively. A unique spatial identifier, usually a GPS point, is entered for each wall record.

## Issues of concern identified by panel of experts

Short-term performance of MSE walls

- Wall drainage
- Wall design, details and specifications
- Retention of wall information

Long-term performance of MSE walls

• Corrosion of the retaining wall systems

### **1.5 Retaining Wall Monitoring and Health Assessment Technologies**

There are several approaches to monitor and inspect retaining wall structures. Instrumentation and remote sensing methods are both used for data collection. Examples of instruments that are used for monitoring include strain gages, tiltmeters, extensometers, inductance coil pairs, fiber optic sensors, crack gauges, piezometers, pressure transducers and slope indicators. The two more common remote sensing techniques are the optical photogrammetry and 3D laser scanning. Table 1.23 presents different monitoring techniques and the measuring parameters.

Agencies follow different inspection schemes for data collection of retaining walls. The City of Seattle, for example, is using a digital protractor for wall tilt measurements, while measuring stations are permanently established on many walls (Molla 2014). The Pennsylvania Department of Transportation has implemented new technology as part of its data collection efforts in 2008 and 2009. LiDAR using a fixed-winged aircraft was used to assess the amount of creep that the Lewiston Narrows wall (the second-longest MSE wall in the world with a length of 2.5 miles). The goal of 0.10 ft proved difficult to confirm because of the low altitude required within the canyon. Use of a helicopter might be tried instead. Down-hole cameras in pipes and culverts are also used to inspect the drainage system (Pennsylvania DOT 2010). Arizona DOT used laser scanning and panoramic photography in "inaccessible terrain" (Priznar et al. 2010). British Columbia Ministry of Transportation and Infrastructure monitors walls with slope indicators, tell-tales, surface monitoring hubs, piezometers, tilt meters, etc. (British Columbia Ministry of Transportation and Infrastructure 2013). In a study conducted in Northern Ireland (Gridpoint Solutions Ltd. 2005) laser scanning was used at 6-month intervals to monitor movement in a 5 m high retaining wall along a railroad cut, to detect changes greater than 5 mm. It was found that this method was faster, cheaper and safer than conventional techniques. Table 1.24 shows some of the monitoring technologies and corresponding references.

Technique	Measurement
LiDAR	Wall identification and condition assessment
Photogrammetry	Condition assessment
Ground Penetrating Radar (GPR)	Object location in soil or concrete
Piezometers & Pressure Transducers	Measurement of stream water elevation
Thermal scanning	Identify water accumulation
Down-hole cameras	Inspection of drainage system
Slope indicators	Monitoring of lateral deformations
Crack Gauges	Monitoring cracks in walls
Surface monitoring hubs	Surface movement
Digital protractor	Lateral movement
Tilt meters	Lateral movement
Strain gages	Measurement of stress reinforcement
Extensometers	Horizontal and Vertical movement
Inductance coil pairs	Measurement of geosynthetic strain
Fiber optic sensors	Strain and temperature measurement
Half-cell potential (Ecorr)	Corrosion monitoring of steel reinforcement
Coupon testing	Corrosion monitoring of steel reinforcement
Linear Polarization Resistance (LPR)	Corrosion monitoring of steel reinforcement

## Table 1.23. Monitoring techniques and measuring parameters for Earth Retaining Structures

Technique	References
LiDAR	Kemeny and Turner (2008), Kim et al. (2009), Oskouie et al. (2016), Priznar et al. (2010), Yen et al. (2011)
Photogrammetry	Cerminaro (2014), Wolf et al. (2016)
Ground Penetrating Radar (GPR)	Hugenschmidt and Kalogeropoulos (2009), Huston et al. (2001)
Destructive techniques for MSE walls	Elias et al. (2009), Fishman et al. (2009), Gladstone et al. 2006, Raeburn et al. (2008)
Nondestructive techniques for MSE walls	Elias et al. (2009), Fishman et al. (2009), Koerner and Koerner (2011), Lostumbo and Artieres (2011), Raeburn et al. (2008)
Digital protractor	Molla (2014)
Down-hole cameras	Pennsylvania DOT (2010)
Fiber Optics	Lostumbo and Artieres (2011)

## Table 1.24. Monitoring technologies and references
#### 1.5.1. LiDAR Mapping and Assessment

Light Detection and Ranging (LiDAR), also often referred to as "3D laser scanning", is an emerging three-dimensional mapping technology that employs a laser and a rotating mirror or housing to rapidly scan and image volumes and surficial areas such as rock slopes and outcrops, buildings, bridges and other natural and man-made objects. Ground-based or terrestrial LiDAR refers to tripod-based measurements, as opposed to airborne LiDAR measurements made from airplanes or helicopters (Kemeny and Turner 2008).

The output from ground-based LiDAR is a point cloud consisting of millions of laser distance measurements representing the three-dimensional scanned scene. The point clouds are then processed to extract geotechnical information, which includes discontinuity orientation, length, spacing, roughness, and block size. High-resolution digital images are also taken of the scanned scene, and these images can be "draped" onto the point cloud using texture-mapping techniques (Blythe 1999) to provide a 3D color Digital Terrain Model (DTM) of the scanned scene. Additional geological and geotechnical information can be extracted from the DTM that would be difficult to observe in the point cloud.

Distance values for millions of points on a reflected surface are generated. From the distance and the orientation of the laser pulse, the xyz coordinates associated with each reflected pulse can be determined. In addition, the intensity of the returned pulse is determined. In general, light colored objects and closer objects give a higher reflection compared with darker objects and objects farther away. Together, the xyz coordinates and associated intensity values for millions of data points outputted by the laser make up the "point cloud".

There are two primary types of 3D laser scanners: time-of-flight scanners and phase-shift scanners. The time-of-flight scanners are capable of a much larger range compared with the phase shift scanners. Thus, time-of-flight scanners would be preferred for large highway slopes and cliffs, while phase shift scanners would be preferred for small underground tunnels, for example. Also, the phase shift scanners have a much higher average data acquisition rate compared with the time-of-flight scanners. In terms of distance and position accuracies, the phase shift scanners have a slightly higher accuracy compared with the time-of-flight scanners.

In the case of the conditions assessment of retaining walls, surveying technology such a LiDAR can be deployed efficiently for providing a 3D data profile of the wall surface. While at present this approach can be effectively deployed on a project level, it is not far in the future when such data can be collected on a network level. Using the data collected over the years, an inventory can be developed for management of walls as a highway asset (Kim et al. 2009).

Figure 1.10 shows an example of LiDAR survey results of an MSE wall. Each pixel in the image represents a set of data points that is tied to x, y, and z coordinates. Linking the output of the LiDAR survey with the database of the walls coordinates can automate the process of condition assessment, and provides synoptic approach to asset management of walls. Survey year and survey results can be linked to the cross-section views and locations of the wall. The survey of wall facing with time can provide important information regarding the condition of the wall since perceptive deformation can be detected by comparing the consecutive scans. A wall inventory database clearly needs to include static data such as wall latitude, longitude, and implement such data within state coordinate system. In addition, foundation type, depth of embedment, and drainage measures should be included.



Figure 1.10. LiDAR mapping of MSE wall (from Kim et al. 2009)

Yen et al. (2011) evaluated the feasibility of using the mobile LiDAR technology to capture geospatial data of highways and use it for surveying, asset management, as-built documentation

and maintenance operations. Mobile LiDAR systems collect field data of up to 150 miles a day, removing the need for lane closures while increasing productivity.

Based on discussions with service providers, a few state DOTs, such as Tennessee DOT, Hawaii DOT, Nevada DOT, and Texas DOT, have contracted with mobile LiDAR service providers for asset management. Caltrans has contracted with a mobile LiDAR survey firm to perform bridge clearance measurements and pavement surveys. Recently, Oregon DOT has purchased a mapping grade mobile LiDAR system. Survey service providers have been using survey grade mobile LiDAR systems to collect data for railroad and power transmission line management.

The pilot participants took about 2 hours with the vehicle speed mostly at 55 mph for data collection of the test section, a 5-mile divided 4-lane highway. This includes time that was taken to complete multiple passes, resulting in redundant test area data as well as capturing data of the two intersections. The multiple passes were valuable in examining system repeatability as well as filling in shadows created by large trucks blocking the LiDAR sensor's Field-of-View. The short data collection time confirms high productivity of mobile LiDAR systems. In addition, the data collection personnel were safely protected inside the data collection vehicle.

Oskouie et al. (2016) used Terrestrial Laser Scanners (TLS) to measure retaining wall displacements. The horizontal joints between the wall panels were used as the benchmarks providing displacement data for all the panels on the wall through a fully automated framework; MSE walls were studied. First, an algorithm was introduced for extracting MSE wall facing panels' horizontal joints from a TLS generated point cloud. Next, the displacement of the wall was determined by comparing the extracted joint displacements. The method's accuracy was evaluated using simulations and real-life data sets with an average accuracy of 94.72 %.

#### 1.5.2. Photogrammetry for condition assessment of ERS

In photogrammetry, the 3D coordinates of a scene are determined from digital images taken of the same scene from different directions. In particular, information on the 3D coordinates is determined from the parallax, which is the change of angular position of two observations of a single object relative to each other. Today it is more common to use a standard digital camera and take multiple images of a scene from arbitrary directions and positions. The multiple camera positions are then determined using a technique called bundle adjustment that involves "feature matching" in overlapping areas of the images (Kemeny and Turner 2008). Digital photogrammetry and 3D modelling software are used to generate 3D models and inspect structures and structural elements.

Cerminaro (2014) investigated the use of photogrammetry to quantitatively assess the condition of retaining walls. 3D models of retaining walls were developed and offset displacements were measured to assess their condition. A case study from a site along M-10 highway in Detroit, MI (Figure 1.11) where several sections of retaining walls had experienced horizontal displacement towards the highway was presented and results were validated with field observations and measurements. A small-scale model was also built in the laboratory. The analysis showed that the accuracy of the offset displacement measurements is dependent on the distance between the retaining wall and the sensor, location of the reference points in 3D space, and the focal length of the lenses used by the camera.

The following steps were followed to obtain deflection measurements:

- Creating reference points on the retaining walls which were used to establish a scale and georeferencing for the 3D models
- Collecting photographs from the retaining wall using an optical camera
- 3D models are created using 3D modelling software and the 2D photographs. The software uses the photographs and reference points to extract the location of each point on the surface of the wall in 3D space, it then uses this information to create a 3D model
- Comparing two 3D models from two points in time and analyzing the changes to obtain deflection measurements

The process involved taking photographs of the object of interest from at least two different locations. From each location there is a line of sight that runs from each point on the object to the perspective center of the camera. The images from two consecutive locations need to have a certain overlap and typically 60 percent overlap is used. In this study a Nikon D5100 was used for data collection. The method used in this study which is the most common method to acquire images for 3D modeling is the Pinhole Camera Model. This camera allows 16.2 megapixels pictures, and has digital single lens reflex (D-SLR), and AFS DX 18-55mm with vibration reduction (VR) lens. For the processing conducted in this study, Agisoft Photoscan Professional was used.

Results from the small scale model in the laboratory showed that the deflection measurements can be accurate when the data is captured within a distance of 40 ft. Three field datasets were taken in this study on the wall on M-10 highway. The distance between the surface of the wall and the camera was approximately 100 ft for each data set. The movement was measured in terms of offset relative to the adjacent walls at the expansion joints using a measuring tape and was confirmed by tilt-meter monitoring. The final results from the 3D model comparison from the case study, however, did not provide reliable deflection measurements. The unreliability of the results is due the actual deflection was not within the accuracy range of the models which were collected at a distance of 100 ft.



Figure 1.11. Retaining wall on M-10 (from Cerminaro 2014)

#### 1.5.3. Inspection of retaining walls using Ground Penetrating Radar (GPR)

The technology of Ground Penetrating Radar (GPR) is used to determine the integrity of concrete bridge columns, retaining walls and roadways. GPR is a geophysical method that uses high frequency electromagnetic waves and detects the reflected signals from structural elements.

Hugenschmidt and Kalogeropoulos (2009) presented a study with the application of GPR for the inspection of retaining walls. The main objective was to locate anchor heads and to gain information on the general structure of the reinforced concrete wall. A semi-automated survey apparatus was developed, which consisted of a railing system, an antenna box, a ladder-like guiding system for the antenna box, an electric motor for moving the box up and down the face of the wall, a survey wheel for controlling the vertical position of the box and triggering the data acquisition and an electronic protractor for monitoring the angle between the guiding system and

the vertical line thus controlling the lateral position of the antenna (Figure 1.12). Data was acquired along the face of the wall using three different antennas; up to 30 vertical lines were acquired per hour. This pilot study showed that data acquisition on large retaining walls can be carried out economically and with high precision.

Huston et al. (2001) developed a prototype handheld system in an attempt to use GPR technology on non-horizontal surfaces, such as columns and walls with higher frequencies than those used to examine roadways. The system performed moderately well in identifying rebars and concrete joints for reinforced concrete columns and walls. It was suggested that more development is needed before the system can be a practical tool for routine inspection of columns and walls.



Figure 1.12. Semi-automated survey apparatus (from Hugenschmidt and Kalogeropoulos 2009)

#### 1.5.4. Condition Assessment of Mechanically Stabilized Earth (MSE) Walls

Elias et al. (2001) reported that the majority of the MSE walls for permanent applications are constructed to date with galvanized steel reinforcements. The galvanized steel, either in strip or grid configuration (95% of applications), is connected to a precast concrete facing. The advantages of galvanization steel reinforcement were listed by Gladstone et al. (2006) as: (1) minimizing the surface irregularities and their contributions to corrosion, (2) lowering consumption rate of zinc compared to steel, and (3) "passivation" of steel due to zinc oxides which lowers the rate of steel consumption compared to non-galvanized steel.

The Association for Metallically Stabilized Earth (AMSE) has compiled an inventory documenting details of 780 MSE walls constructed in the United States since 1972 (AMSE 2006). Approximately half of the walls in the AMSE inventory are located in the Western Region of the United States, within an arid climate where backfill sources are alkaline. Compared to steel grid type reinforcements, which are used predominantly within the Western Region, use of strip reinforcements is more uniformly distributed geographically. Approximately 40 percent of the walls constructed with strip reinforcements are located in the more temperate Southern climates, where soils are normally slightly acidic.

Corrosion of the tensile elements due to the chemical hardness of the soil-water is a major concern for the long term durability of MSE walls. The choice of backfill material and reinforcing material are two key issues to address in attempting to mitigate corrosion of MSE wall. Assessing the corrosion state of metal strips reinforcing highway retaining structures is one of the important asset management tasks for departments of transportation across the country.

Controlling factors of corrosion rates include (Raeburn et al. 2008):

- Water content soil water contains the salts and constitutes the electrolyte necessary for corrosion
- Soil resistivity, when measured at saturation, gives a figure related to the total amount of salts present in the soil 153
- pH (potential of hydrogen), that governs the solubility of corrosion by-products and thus the buildup of protective layers around the buried metal
- Chloride content-chloride is the most common aggressive salt

#### • Sulfate content

Destructive and nondestructive techniques are available for corrosion detection. Measuring metal loss data from the exhumation of a wall is a common destructive technique for corrosion detection and measurement. Due to process of excavation while maintaining the integrity of the wall, this method is limited to reinforcement elements near the surface. Such a limitation may provide results that are not representative of the most corrosive area of the wall. Corrosion rates are established through weight loss and thickness measurements, and usually multiple measurements are made at different times to assess the effect of time on the rate of metal loss (Gladstone et al. 2006). The method is expensive since it is labor intensive, and requires caution in order to ensure that the stability of the wall will not be compromised during sampling.

Popular nondestructive methods for assessment of corrosion are polarization resistance measurements, linear polarization resistance (LPR) measurements, coupon testing and half-cell potential measurements of reinforcement (Ecorr). In polarization resistance measurements, composition and geometry of the surface of reinforcement should be known. The approach is based on converting the polarization resistance to a corrosion rate. For LPR, the potential is varied from "-5 to -20 mV" to "+5 to +20 mV" around the free corrosion potential while simultaneously measuring the applied current. Polarization resistance is determined from the slope current and potential. Since corrosion rates vary throughout the year, measurements should be taken during different seasons to attain an average corrosion rate for the structure (Gladstone et al. 2006). Coupon testing and half-cell potential measurements of reinforcements are installed at regular intervals during MSE wall construction. Zinc bars and steel plate coupons are installed, and reinforcements are wired for half-cell potential measurements at each monitoring station along the wall. Elias et al. (2009) advise that "given the advantages, utilization of remote electrochemical methods is highly recommended with at least some coupons buried for retrievals to confirm results." Their provided rule of thumb regarding installation is two locations spaced at least 200 ft (60 m) apart for MSE walls 800 ft (250 m) or less in length and three locations for longer walls. At each location, corrosion should be monitored at a minimum of two depths. Caltrans has developed a typical layout of 18 clustered coupons to be periodically extracted. With respect to frequency of assessing corrosion, Elias et al. (2009) recommend that potential and polarization resistance measurements (owing to their sensitive nature) be made monthly for the first three months, bi-monthly for the next nine months, and annually thereafter.

Koerner and Koerner (2011) presented instrumentation devices and strategies to monitor MSE walls with geosynthetic reinforcement during and soon after construction. The most common instruments are strain gages, fiber optic sensors, extensometers, piezometers, inductance coil pairs and slope indicators. It is recommended, that sampling and testing occur at five to seven-year intervals for a minimum of four retrievals, or one-third the expected life of the facility (Elias et al. 2009).

# 2. DESIGN AND DEPLOYMENT OF WIRELESS MONITORING SYSTEMS FOR THE I-696 AND M-10 RETAINING WALLS

### 2.1. Introduction

Sensors can be installed on retaining walls whenever an asset manager is seeking quantitative data related to how a wall is performing. Specifically, instrumentation provides managers with insight to the behavior of the structural and geotechnical subsystems of the overall wall system. To understand the behavior of retaining walls, performance parameters of interest include: horizontal and vertical wall movements, deterioration of wall elements, drainage behavior of the backfill, lateral earth pressures, vertical stress distributions at the wall base, corrosion state of steel reinforcement, pore pressures below the structure, and other environmental parameters that vary temporally (*e.g.* temperature, rainfall) (FHWA, 2009). Sensors can be used to measure some of these performance parameters. Instrumentation can be installed during wall construction, but the cost of instrumentation makes this a rare choice. Rather, instrumentation is typically installed after construction and only if there are some concerns about wall performance that warrant the cost and effort of instrumentation (Koerner and Koerner, 2011). When instrumentation is selected, many performance parameters are difficult to measure, especially those associated with the soil system including earth pressures on the back-wall surface.

In general, the most common measurement of retaining walls is wall movement including tilting and relative displacements between wall panels. While all walls exhibit some cyclic movement based on seasonal variations, progressive wall tilt is a serious issue and if left unmitigated, can lead to instability of the system. Measurements are commonly taken by tiltmeters (also termed inclinometers) which provide a measure of the rotation of the wall away from the system backfill (WSDOT, 2011). Tiltmeters can be installed permanently or they can be used intermittently. When used intermittently, mounting plates are installed on the wall with tiltmeters manually applied to these plates when measurements need to be taken (*e.g.* weekly, monthly). Linear variable displacement transformers (LVDT) can serve as another sensor-type useful for measuring the movement of wall panels relative to one another. LVDTs require the mounting of the two ends of the sensor to two adjacent wall panels to measure relative options for relative displacement measurements between panels. Strain gages can also be used

to measure the strain response of the wall over a defined gage length. Strain gages have been used widely on sensing



Figure 2.1. (a) Retaining wall site locations (Source: Google 2019); (b) I-696 Wall

strain in steel sheet pile walls; however, in some instances they have also been used on concrete walls. Strain can be highly valuable for measuring the deformable body response of the retaining wall under load (*e.g.* active earth pressure, thermal loads) including assessing the evolution of cracks that jeopardize the long-term durability of the wall. Strain gages are typically metal foil gages embedded in a thin polyimide carrier film, but alternative strain sensors such as BDI strain transducers have found use in walls in the past (BDI, 2006).

In this project, an instrumentation strategy is proposed for monitoring highway retaining wall systems. Emphasis is placed on a strategy suitable for rapid installation after wall construction to provide data on wall behavior. The project considers instrumentation that can be installed on exposed wall surfaces and that is easy to deploy. The instrumentation strategy adopts tiltmeters to measure wall tilt, long-gage strain gages to measure thermal and flexural strains, and thermistors to measure wall temperatures. These measurements offer insight to the wall response to its loading environment. The instrumentation is also intended to assess if a wall is behaving as designed aiding in an assessment of its capacity. To collect measurements from these sensors, a wireless monitoring system is adopted. The wireless sensor nodes are designed to use solar power for their operation and cellular modems for their communication, making them easy to install. The proposed wireless instrumentation strategy is validated using two retaining wall systems in southeast Michigan. The first retaining wall system is a reinforced concrete cantilever wall along I-696 while the second is a caisson-supported reinforced

concrete wall along M-10. The M-10 wall system has a history of failures associated with corrosion-induced failure of the steel tendon anchoring the wall to the caisson (MDOT, 2013; AECOM, 2016).

#### 2.2. Retaining Wall Systems

#### 2.2.1. Overview of the I-696 and M-10 Retaining Wall Systems

In Michigan, common retaining wall systems along the state highways include: cantilever reinforced concrete (RC) walls, RC walls supported by caisson tiebacks, mechanically stabilized earth (MSE) walls, sheet pile walls and soldier pile with legging walls. The cantilever RC wall is the most common wall type in the southeast Michigan region. The I-696 corridor in the Detroit metropolitan area has a large number of RC cantilever walls. Two wall panels located near Central Park Boulevard supporting West Eleven Mile Road on the south side of I-696 in Southfield, MI are selected (Figure 2.1). These walls were constructed in the mid-1980's and are more than thirty years old. These retaining wall locations are selected due to observed relative displacement between adjacent wall panels and due to a high amount of drainage emanating from the lower portions of the wall surface. In addition to the I-696 wall, one wall panel of the M-10 retaining wall located near Schaefer Highway in Detroit, MI (Figure 2.1) is also selected for this project. The M-10 wall system is supported by RC caissons with the wall anchored to the caissons using post-tensioned tendons at an angle of 30 degrees from the vertical wall face. The M-10 wall is selected due to the history of corrosion failure of the post-tension anchor rod (MDOT, 2013; AECOM, 2013).

#### 2.2.2. On-Site Visual Assessment

During on-site visual inspection in June 2017, the M-10 retaining wall system was found to be in very good condition overall. Due to the history of failing anchor rods, the on-site inspection specifically sought out visual signs of major wall deformations that might indicate distress. Even though the M-10 walls were in a good structural condition, evidence of minor distress was observed. At the top of the wall, pavement deformation of the supported service road indicated movement and compaction of the backfill system. Differential displacement (in plane and out of plane) between wall panels was also evident. In addition, mild leakage stains, small areas of concrete spalling, and vertical cracking were evident on the face of the wall. Photos from the on-site inspection are provided in Figure 2.2. Given the observations made, the wall was deemed to be an excellent candidate for monitoring. Specifically, monitoring could be used to assess if wall movement is the cause of the distressed pavement observed on the upper service road supported by the wall system.

The I-696 retaining wall system was also inspected in June 2017. While the wall system was found to be in very good condition, several concerns were raised. First, there appeared to be



Figure 2.2. M-10 retaining wall system distress (June 2017): (a) wall tilting observed at the wall top; (b) severe lateral cracking of the service road pavement; (c) leakage stains on the wall face at the top of the wall; (d) vertical cracking of the wall face; (e) joint movement between wall panels; (f) distress of storm water pipe behind the wall; and (g) backfill soil evident at the wall joints.

excessive leakage of water through holes at the lower portions of the wall panels and at the vertical joints between panels (Figures 2.3a and 2.3b). The weather at the time of the site visit was dry and warm suggesting the weeping evident was associated with either a drainage system behind the wall in bad condition or due to a saturated backfill. The uniform nature of the weeping along the wall length suggested the later. The wall panels appeared to have minor levels of tilt and displacements between them. Given the observations in the field, the I-696 wall system also proved to be a good candidate for long-term monitoring in this project.

# 2.3. Long-Term Monitoring System Design and Installation

One panel of the M-10 retaining wall system and two panels of the I-696 retaining wall were selected for instrumentation. A permanent monitoring strategy that focused on instrumentation installed on the front face of the retaining walls was sought. The monitoring strategy considered for both walls included the measurement of tilt, strain and temperature.



Figure 2.3. Issues observed on I-696 wall panels (June 2017): (a) water leakage via the lower portion of the wall expansion joint; (b) walkway failure on the top of the wall back fill; (c) prolonged and severe weeping through drainage holes; (d) moderate levels of vertical cracking; (e) excessive wall expansion at panel joints; (f) differential tilt of adjacent panels.

A triaxial accelerometer well suited for the measurement of tilt was adopted to measure the pitch, roll and yaw of the wall. Strain gages bonded to an aluminum plate were used to provide a long-gage strain sensor for measuring thermal and flexural strain in the walls. Finally, thermistors were adopted to measure wall temperatures. The motivation for the selection of these sensors was that they are relatively low cost yet provide insightful measurement of the wall response to its loading environment.

#### 2.3.1. Sensor Placement on the Wall Panels

Unlike bridges that require dozens of sensors on many of the main structural elements, retaining walls have expansion joints between wall panels that enable them to act with some degree of independence. A uniform instrumentation strategy was sought for measuring the panels in the M-10 and I-696 wall systems. For the M-10 wall system, the panel dimensions on the wall were measured during the on-site inspection to be 25'-0" wide and 22'-6" high from the front grade level, on average. The M-10 wall system is stabilized by a caisson with a tie rod anchoring the



Figure 2.4. Possible vertical wall deformations of the M-10 wall

wall approximately 15'-9" from the front grade level. The history of tieback failure in the M-10 wall systems informed the instrumentation strategy. When the tieback is engaged, the wall deflection is restrained at the tieback with lateral earth pressures inducing a bulging deflected shape below the tieback point. Figure 2.4 shows the hypothesized deflected shape of the wall when the tieback is properly engaged as intended (*i.e.* the curve denoted as (a)-(c) in Figure 2.4). In this scenario, the maximum flexural moment in the wall would be below the tieback. If the tieback fails, the wall would then begin to respond like a cantilever wall with a deflected shape defined by the curve denoted as (a)-(b) in Figure 2.4; in this case, the maximum tilt would be measured at the top of the wall. The M-10 wall instrumentation places two tiltmeters on the same wall panel to assess the wall deflected shape: one at the top of the wall (19'-0")

from grade) and another at mid-height (10'-6" from grade). Additionally, two long gage strain sensors were installed: one at the base (1'-0" from grade) and the other at mid-height (10'-6" from grade). Thermistors were also installed on the wall panel adjacent to each strain sensor to measure the wall temperature. All of the sensors were installed at the center of the panel along the same vertical line. Figure 2.5 summarizes the location of the sensors installed on the M-10 wall panel.



Figure 2.5. M-10 instrumentation: two tilt-meters, two strain gages & two thermistors on a single panel.



Figure 2.6. I-696 instrumentation on two separate panels 40 feet apart (so called "tall" and "short" walls)

For the I-696 wall system, two panels were selected as previously described. The panel at the 0 ft. reference vertical line (see Figure 2.6) is roughly 20'-0" wide and 26'-0" high above the front grade line while the panel at the -40 ft. vertical reference line is 20'-0" wide and 23'-0" high above the front grade line. The different heights are due to an inclined exit ramp for I-696 at the base of the wall system. The wall panels are hypothesized to respond with either flexural bending (with maximum tilt at the top of the wall) or through rigid body rotation. The instrumentation strategy was nearly identical to that of the M-10 wall panel with tiltmeters placed at the wall top and mid-height, and strain gages placed at the panel mid-height and base (Figure 2.6). The taller panel at the 0 ft. reference had tiltmeters installed at 13'-6" and 24'-0" above grade while the shorter panel at the -40 ft. reference had tiltmeters installed 11'-0" and 21'-0" above grade. The tall wall had two strain sensors installed: one at the base 3'-6" above grade and the other at mid-height at 13'-6" above grade. Similar to the M-10 wall, thermistors were installed adjacent to each strain sensor to measure wall temperature.

#### 2.3.2. Sensing Transducers

Based on previous field measurements collected by MDOT on the M-10 wall system, the wall was measured to tilt +/- 0.05 degrees on a weekly basis. Based on this, a tilt-meter with a resolution of 0.01 degrees was sought. An orientation sensor with a triaxial accelerometer included was selected: Bosch BNO055 (Bosch, 2016). The BNO055 is a complete inertial measurement system including a triaxial accelerometer (14-bit resolution), triaxial gyroscope (16-bit resolution) and triaxial magnetometer. Only the internal accelerometer of the BNO055 sensor was used in this project to assess wall tilt. The BNO055 accelerometer has an acceleration measurement range of +/- 2g with a resolution of 1 mg. An additional feature of the sensor is that it outputs its measurements using a digital communication protocol (*e.g.* SPI, UART, I2C) allowing it to be easily interfaced to a microprocessor. Accelerations are used to estimate the rotation of the sensor; given the 1 mg acceleration resolution, this yields a tilt resolution lower than the desired 0.01 degrees sought. An additional feature of the BNO055 is that it includes temperature compensation providing thermally stable measurements between - 40 °C (-40 °F) and 125 °C (257 °F).

Strain sensors were needed to measure thermal strain and strain associated with the flexural response of the wall to lateral earth pressures. Accurate measurement of strain in concrete

structures typically require a long gage length (in order for strain measurements to be immune to



Figure 2.7. Strain gage set-up: (a) long-gage aluminum plate with four active gages attached; (b) close-up of bonded gages; (c) full-bridge circuit.

localized cracking). A strain measurement system based on metal foil gages was custom designed for the M-10 and I-696 walls. The approach adopted an aluminum plate (2" wide, 1' long and 0.25" thick) upon which four metal foil strain gages were attached (Figure 2.7b). The plate has two holes along its longitudinal axis for mounting to the wall panels with a gage length of 10" between the mounting holes. Small metal foil gages (Omega KFH-6-350-C1-11L1M2R) were bonded to one side of the aluminum plate using cyanoacrylate glue. The gages had 350  $\Omega$  nominal resistances with 2.05 gage factors and 2% transverse sensitivity (Omega, 2018). The four gages were configured in a full bridge circuit with two gages measuring axial strain and two measuring transverse strain. The full bridge configuration (Figure 2.7c) has the benefit of minimizing the thermal sensitivity of the strain set-up. The two longitudinally aligned gages were placed on opposite sides of the bridge and are denoted as R<sub>F1</sub> and R<sub>F4</sub> in Figure 2.7c; the transverse gages are used for thermal compensation and are denoted as R<sub>F2</sub> and R<sub>F3</sub>. The gages were powered using a 3.3V source (V<sub>in</sub>) with the bridge output voltage (V<sub>out</sub>)

fed into a standard instrumentation amplifier with a 1000 times gain. This full bridge set-up allowed longitudinal strain,  $\varepsilon$  in the plate, to be calculated as:

$$\varepsilon = \frac{-2\left(\frac{V_{\text{out}}}{V_{\text{in}}}\right)}{GF\left((1+\nu) + \left(\frac{V_{\text{out}}}{V_{\text{in}}}\right)(1-\nu)\right)}$$
(2.1)

where GF is the gage factor of the metal foil gages and v is the Poisson ratio of aluminum. Prior to field installation, the thermal compensation functionality of the bridge was assessed in the lab. With the plate holding static load, the temperature of the plate was varied from 25 °C (77 °F) to 117 °C (243 °F). The thermal sensitivity of the full gage configuration on the aluminum plate was 0.13  $\mu\epsilon/°C$ . This thermal sensitivity is minimal due to the anticipated magnitude of expected strain (which is expected to be in the 10's of  $\mu\epsilon$ ). Prior to deployment of the plate to the M-10 and I-696 wall systems, a conformal polymeric coating was applied to the gages and lead wires to make them water tight and to protect them from the harsh operational environment.

Measurement of the temperature of the walls is critical to understanding their thermal expansion behavior. As a result, waterproof thermistors were selected for installation on the wall surfaces. The thermistor selected for this project was the TDK Group B57020M2 (TDK, 2018). It is contained in a watertight plastic case which ensured its durability when installed in the field.

#### 2.3.3. Wireless Sensing Node

To collect data from the sensing transducers selected for the retaining wall systems, a wireless sensing node termed *Urbano* (Flanigan and Lynch, 2018) was adopted. *Urbano* (Figure 2.8) was designed at the University of Michigan and is an ultra-low power wireless sensor node that utilizes cellular communications to directly transmit its data to the Internet. This is a very attractive feature because it eliminates the need for an on-site base station that would otherwise be needed to collect data from a network of wireless sensors. Use of cellular telemetry has other advantages including precise time synchronization of the nodes. To transmit and receive data, *Urbano* utilizes a Nimbelink Skywire 4G cellular modem. The radio consumes 616 mA (referenced at 3.3V) when transmitting, 48 mA when idle, and 8.6 mA when in low-power mode; these seemingly high-power numbers are offset by the high data rates supported by the radio including a 5 Mbps upload rate. When the radio is needed, *Urbano* is designed to turn the radio on for bursting out data and then turning it back into sleep mode in order to minimize consumption of the battery energy.

Characteristic	Specification		
Computational Core	8-bit RISC Atmel 2561V µC at 16 MHz		
Memory	256 kB Flash; 512 kB SRAM		
Sensor Interface	10-bit ADC with 8 differential or 16 single-ended channels		
Base Power without Cell	75 mW (Active); 21 uW (Sleep)		
Cellular Communications	Verizon 4G Cellular Modem (2W power)		

# Table 2.1. Urbano performance specificationscSpecification



Figure 2.8. Urbano wireless sensor node for data collection

At the core of the *Urbano* node is an 8-bit microcontroller (Atmel Atmega2561V) clocked at 8 MHz. The microcontroller has 256 kB of flash memory for program storage and 8 kB of SRAM for data storage on-chip. Memory is expanded with an addition 512 kB of SRAM using an off-chip memory chip (Cypress CY62148EV30). The 8-bit microcontroller includes a multichannel 10-bit analog-to-digital converter (ADC) capable of a maximum sample rate of 200 kHz. In this study, the on-chip ADC was used to measure temperature and strain. The strain gage and thermistor analog outputs were interfaced to the ADC through amplified bridge circuits as previously described. The microcontroller also has traditional serial communication ports (including UART and SPI) to which digital sensors can be attached. The tiltmeter (Bosch BN0055) was interfaced to *Urbano* using the UART serial communication port. The Atmega2561V is powered by a 3.3V source and draws 7.3 mA when active, but 4.5 µA when in power-save mode.



Figure 2.9. Two axes used for dual-axis tilt sensing

The BNO055 tilt-meter has its own microprocessor through which the device is operated by writing and reading on-chip registers. Acquiring acceleration data from BNO055 requires the *Urbano* microcontroller to write commands to registers on the BNO055 and then reading the measurement data from read registers. The Atmega2561V was configured to take 100 acceleration measurements on each of three axes of the BNO055; the accelerations were averaged on the *Urbano* microcontroller for each axis before pitch and roll tilt was calculated by the *Urbano* node.

#### 2.3.4. Dual-Axis Tilt Calculation

The project selected the BNO055 accelerometer to serve as the basis for measuring wall tilts. The sensor was selected due to its performance attributes (namely, low noise floors) that allow tilts to be precisely calculated from the acceleration readings. The internal tri-axial accelerometer in the BNO055 is capable of measuring accelerations across a wide vibration range all the way down to static accelerations (*i.e.* can measure orientation relative to gravity). As a result, when the accelerometer is rotated in three-dimensional space and held stationary, the readings of acceleration on three axes can be transposed to measure tilt on three axes. In the case of the retaining wall systems monitored in this project, the out of plan tilt of the wall was sought. As shown in Figure 2.9, to measure out of plan tile, static acceleration relative to gravity (denoted as 1g axis) on two internal axes (x and y in Figure 2.9) are needed to estimate the angle,  $\theta$ . For example, when the tilt is 0° as shown on the left of Figure 2.9, the x-axis acceleration would be 0 g and the y-axis acceleration would read 1 g. When rotated on the zaxis by the angle,  $\theta$ , as shown on the right side of Figure 2.9, the x-axis acceleration will be a small number greater than 0 g while the y-axis will report an acceleration less than 1 g. While one axis is sufficient to estimate tilt (for example, reliance only on the y-axis acceleration measurement), there are three major benefits to including a second axis in determining the angle of inclination. First, the tilt measurement is less noisy with constant sensitivity. Second, it does not require alignment of the sensor package with a precise orientation of gravity before measurements can be taken. Finally, it can measure tilts in 360°. The constant sensor sensitivity is derived from reliance on two measurement axes. Whenever the incremental sensitivity of one axis is reduced (such as when the acceleration on that axis approaches +1 g or -1 g), the incremental sensitivity of the other axis increases. For a dual-axis tilt calculation (Figure 2.9), an easier and more efficient approach is to use the ratio of the two values, which results in the following calculation for tilt,

$$\theta = \tan^{-1} \left( \frac{A_{X,OUT}}{A_{Y,OUT}} \right)$$
(2.2)

where the inclination angle,  $\theta$ , is in radians.

#### 2.3.5. Packaging Wireless Sensor Nodes

The *Urbano* wireless sensor node was packaged in a water-tight NEMA-rated enclosure (8  $\frac{3}{4}$  x 5  $\frac{3}{4}$  x 3 in<sup>3</sup>) prior to deployment on the M-10 and I-696 retaining wall systems. A picture of the wireless sensor enclosure is shown in Figure 2.10. The enclosure was selected to comply with the Michigan Department of Transportation restrictions on protrusions from the wall entering the roadway space to be less than 4" (this package extended only 3" into the roadway space).

Inside each enclosure was an *Urbano* node, signal conditioning circuits (*e.g.* amplified bridge circuit for the full-bridge strain set-up of Figure 2.7), and a 12V (2.9 A-hr) sealed lead acid battery (Powersonic PS-1229). Lead acid batteries are excellent battery solutions for recharging in extreme cold environments. To charge the battery while utilizing it to power the node, a charge controller was included in the enclosure. Each node was powered by a 12V (10 W) solar panel (Acopower HY010-12M) housed outside the enclosure. The tilt-meter nodes also included the Bosch BNO055 inside the enclosure bonded to the enclosure's bottom surface. The temperature and strain sensor nodes had both of these transducers installed outside of the enclosure for direct installation to the surface of the wall panels.

#### 2.3.6. Field Installation on the Wall Systems

A total of 11 wireless sensor units were installed at the two wall sites (M-10 and I-696) in August and November 2018. Seven units were installed on the I-696 wall panels on August 25, 2018 during a warm and dry day. Four additional wireless sensors were installed on the M-10 wall panel on November 27, 2018 during a dry, but cold day. First, solar panels were mounted to light poles at the top of the walls (Figure 2.11a) with one panel used per wireless sensor node. To ease



Figure 2.10. Inside the wireless sensor enclosure showing the Urbano node, solar charge controller, tilt sensor (Bosch BNO055), and lead acid rechargeable battery

their installation, the panels were pre-mounted to an aluminum frame system with a predefined angle optimized to maximize solar energy capture. Next, the wireless sensor enclosures were installed on the wall surfaces. To mount each enclosure, a 12-gage aluminum plate (14 by 14 in<sup>2</sup>) was bolted to threaded rods pre-anchored into the wall by a crew from the Michigan Department of Transportation (Figure 2.11b). The aluminum plates had holes drilled in them to allow the wireless sensor enclosures to be mounted to the aluminum plates using threaded screws; the enclosures were installed after the plates had been attached to the wall surface. To reinforce the enclosure mounting, quick setting epoxy was also used between the back of the enclosure and the aluminum plate. The long-gage strain sensor was similarly installed by bolting the plate to threaded rods pre-anchored into the walls. Figure 2.11c shows tilt and strain enclosures mounted to the I-696 walls. Installation of the nodes took approximately one day per wall panel. Required for installation was a road/lane closure to ensure the safety of project personnel installing the nodes. MDOT provided a lift truck with a two-person crew to assist the team.

#### 2.4. Data Acquisition Process

An automated data acquisition architecture (Figure 2.12) was created to collect data from the wireless sensors installed on the M-10 and I-696 wall systems. The system was designed to operate on a schedule with each sensor node programmed to collect data every one or two hours. The nodes on the I-696 wall panels collected data every hour while those on the M-10 wall were programmed to collect every two hours. The change in sampling frequency for the M-10 panel (which was installed in November 2018) was due to observing that the tilt and strain data on the I-696 panels (which were installed in August 2018) had less hourly variation than originally anticipated. When not sensing, the nodes remained in a sleep state to preserve battery energy. The tilt measuring nodes were scheduled to collect 100 acceleration samples (at 100 Hz) on each axis, average the samples, and report mean accelerations for each of the three axes. The strain and temperature units were scheduled to only take one sample at each hourly (or bi-hourly in the case of the M-10 panel) sampling cycle. After the data was sampled locally by each Urbano node, the nodes were programmed to communicate data to a cloud server using the cellular modem integrated with each node. The retaining wall monitoring systems used a commercial cloud data portal called Exosite for data management (Exosite, 2018). The project team selected a commercial data platform so that MDOT could continue to use the portal to collect wall data after the project officially ended.

Exosite hosted the monitoring data with a web portal used for data visualization. Strain and temperature were displayed in the portal as measured. The three axes of acceleration from each tiltmeter were processed on the Exosite server to measure tilt by using a real-time program coded in the high-level programming language called Lua. The Exosite web portal offered views of the measurement data in real-time using tabular and graphical displays (for example, see Figure 2.13). It also offered the ability for system end users to download the data for offline analysis.



Figure 2.11. Installation of instrumentation on the I-696 wall: (a) seven solar panels to power seven wireless sensor nodes; (b) aluminum mounting plate bolted to the wall surface; (c) installed wireless sensor enclosures with strain sensor evident.



Figure 2.12. Data acquisition architecture for the I-696 and M-10 retaining wall monitoring systems (Admassu et al, 2019)

← → C ▲ https://portals.exosite.com/manage/data						
<b>%</b> E)		Kidus Admassu PORTAL 💌		Kidus Admassu 💌   Support		
	_					
MANAGE	Data	Data Information	C 🛛	+ Add Data		
Home	Name	Data Update	Data Graph	t Reported Time		
Data	Portal: Kidus #		-85.5			
Devices	Device: <u>1-696</u>	Name: Top_Tilt_Node_0m_Y_Dec				
Events	Bottom_Str	Current -87.313564529815		3:56:29 Aug 4, 19 America/New_York		
Dashboards	Bottom_Th	Value:	-86.5	3:56:29 Aug 4, 19 America/New_York		
Scripts	Device: M-10	Units: degrees	-87.0			
Settings	Bottom_Sti	Format: float		3:12:49 Aug 4, 19 America/New_York		
Start	Bottom_Th	Storage: 42.08 KB	-87.5	3:12:49 Aug 4, 19 America/New_York		
map_domain_dashl	Mid-height	Source Info	-88.0	9:19:03 Dec 17, 18 marica/New York		
		Device: I-696_Tilt_Nodes_at_0m		2:19:03 Dec 17, 18		
	Mid-height	Alias: ay_deg1	7/20 7/23 7/26 7/29 8/1 8/4	merica/New_York		
	Mid-height	RID: fad1a52cd43a794b4c5a69d3d255e1688d0e762b	0.00 0.00 0.00 0.00 0.00	I:16:14 Aug 4, 19 America/New_York		
	Mid-height		<	I:16:14 Aug 4, 19 America/New_York		
Mid-height			Write Data	I:16:14 Aug 4, 19 America/New_York		
	Mid-height	Retention	whe bata	1:16:14 Aug 4, 19 America/New_York		
	Mid-height		Data value: UPDATE	I:16:14 Aug 4, 19 America/New_York		
	Mid-height Duration:	infinity custom		1:16:14 Aug 4, 19 America/New_York		
		Duration:	Data Log (last 200 records) <u>View Detail</u> <u>Export Data</u>	37:19 Jul 14, 19		
	.op_m_r	Count:	Time Value	merica/New_York		

Figure 2.13. Screenshot of the online data management portal (Exosite)

# 3. RESPONSE DATA ANALYSIS OF THE I-696 AND M-10 RETAINING WALL SYSTEMS

#### 3.1. Introduction

As previously described, this project deployed a permanent wireless monitoring system on three retaining wall panels in southeast Michigan to measure their response to operational and environmental load conditions. The data collected was intended to serve as the basis for assessing the performance of the walls within a risk management framework. In total, 11 wireless sensor nodes were installed on three separate retaining wall panels: two wall panels were located on the southside of the I-696 freeway in Southfield, Michigan near Central Park Boulevard and West Eleven Mile Road, while the third panel was located on the southside of M-10 freeway in Detroit, Michigan near the intersection of Schaefer Highway and James Couzens Highway. The wireless sensors deployed as part of the monitoring systems measured the tilt, strain and temperature of the wall panels. Data from the I-696 wall panels has been collected hourly since August 25, 2018 while data from the M-10 wall panel has been collected every two hours since November 27, 2018.

This chapter will provide a detailed overview of how the wireless monitoring systems performed over the one year monitoring period. Lessons learned from the installation of the monitoring systems will be presented. Using the data collected, a detailed analysis of the three retaining wall panels will be introduced by analyzing the different wall responses collected since August 2018. In addition, meteorological data (*i.e.* precipitation) was acquired from meteorological measurement stations located within 10 miles of the retaining wall sites. Time synchronized wall response and meteorological data was used to explore how the behavior of the wall panels change as a function of environmental conditions. The primary goal of the analysis was to correlate wall responses to environmental loads with the aim of identifying load conditions leading to maximum wall responses (*e.g.*, maximum tilt, maximum strain). Specifically, the analysis attempted to understand the thermal response of the wall panels under diurnal and seasonal variations in addition to changes associated with precipitation. The presentation of the

response data collected from the three wall panels are presented in order: tall I-696 wall panel, short I-696 wall panel, and M-10 wall panel.

# 3.2. Performance of the Wireless Sensor Units

An objective of the project was to assess the performance of the wireless monitoring systems on the retaining wall systems. As previously described, three wall panels were monitored with 11 wireless sensor nodes that had been designed and assembled at the University of Michigan. A total of 16 channels of data were collected (6 channels on the M-10 wall panel, 4 channels on the I-696 short wall panel, and 6 channels on the I-696 tall wall channel). Figure 3.1 presents the performance of the 11 wireless sensor nodes over the course of the project. The green markers in the figure correspond to a valid measurement communicated to the cloud database server; the absence of a marker suggests the unit did not communicate data.

The wireless sensor nodes on the I-696 wall panels were installed on August 25, 2018. The nodes on both the tall and short wall panels worked properly from their initial installation. However, the sensors installed at the bottom of both wall panels had issues immediately after installation. The nodes collecting the bottom flexural strain from both wall panels and temperature on the tall wall



Figure 3.1. Performance of the wireless sensor nodes on the M-10 and I-696 panels.

panel did not operate after installation. On October 9, 2018, the team returned to the site and repaired the units after which they worked properly. During site visit, there was no apparent reason for the units not working. To avoid debugging the units at the site, the hardware was quickly replaced with new hardware after which the nodes properly collected data. The interface circuits for the strain sensors were also adjusted during the October 9, 2018 site visit to ensure all of the strain measurements would fall within the wireless node measurement range. The sensors operated well for the majority of the time thereafter on both wall panels until the winter season. By mid-winter a number of units struggle to remain in operation reporting their data. As shown in Figure 3.1, the following channels eventually stopped reporting data: bottom strain on the I-696 tall wall panel (stopped July 2019), mid-height tilt on the 1-696 short wall panel (stopped December 2019), mid-height temperature on the I-696 tall wall panel (stopped September 2019), and bottom temperature on the I-696 tall wall panel (stopped July 2019). The remaining channels on the I-696 panels continue to work and report data. The precise reason the sensor not working remains unclear and should be investigated further. Nonetheless, a sufficient level of data was collected by the wireless monitoring systems on the short and tall wall panels of the I-696 systems.

The wireless sensing nodes on the M-10 wall panel were installed on November 27, 2018 during inclement weather conditions. A total of 4 sensor nodes were deployed all of which worked properly upon installation. Shortly after installation, two of the nodes measuring strain and temperature at the mid-height and bottom of the wall stopped operating (the node measuring mid-height strain and temperature stopped in December 2018 and the node measuring bottom strain and temperature stopped in January 2019). It was not until after the harsh winter could the units be inspected and repaired. The units on the M-10 wall inspected on April 21, 2019 revealed the nodes had low nominal voltages, suggesting the batteries installed in November 2018 had been run down during the winter during cold temperatures and days with limited sun. The battery was replaced on the node measuring bottom wall strain and temperature; this unit has operated without incident since. During the site visit on April 21, 2019, the team did not have time to investigate the wireless node measuring mid-height strain and temperature. Later, these measurements were deemed not critical and the unit was never repaired.

In late May 2019, the wireless sensor node monitoring the top tilt of the M-10 wall panel went down. Given the critical nature of the tilt measurement, a site visit was performed on May 29, 2019 to repair the unit. During the repair, it was evident that the wire connecting the solar panel to the nodes had corroded due to moisture running down the solar panel wire and pooling at a connection where copper wires were exposed. This finding may explain why some of the I-696 sensors have gone down with time. In the future, extra care will need to be made to make the wire connection between the solar panels and the units is waterproof by selecting a more robust connector.

The performance issues encountered by the wireless sensors on the I-696 and M-10 wall panels provides insight to how the node designs could be improved for future deployments. A summary of the key findings are:

- The connection between the solar panel and the wireless sensor nodes is prone to corrosion; in the future, a more robust connection should be used less vulnerable to water and corrosion issues.
- The wires connecting the solar panels to the wireless sensor nodes provide a pathway for water to drip down to and pool at the connections; installation methods should avoid

installing the solar panel wires in a manner where gravity will naturally pull the water down to the connection.

- The wire between the solar panel and the node could be removed by mounting the solar panel directly to the enclosure; this was infeasible in this study because the units faced north and would not have direct exposure to the sun (and hence the need for separate solar panels installed on light poles above the walls).
- The winter season was harsher than anticipated resulting in wear down of the lead acid batteries. Larger batteries with a higher energy capacity should be considered. Also, the design of the wireless sensor node could be revised to lower its power requirements so as to alleviate the demands on the battery recharged by limited solar energy in the winter.

## 3.3. Tall I-696 Retaining Wall Panel

#### 3.3.1. Description of the Wall System

The tall wall panel of the I-696 wall system is more than 30 years old and was designed as a classical reinforced concrete (RC) cantilever retaining wall system. The wall was measured during the site visit to be roughly 26' above the grade line. However, the engineering drawings of the wall system reveal the wall sits on a 3'-0" thick RC footing system and has a height of 28'-5" above the top surface of the footing (Figure 3.2a). On the wall far side (F.S.), there exists a two layer backfill soil system that consists of a 13'deep medium compacted sandy soil stratum resting upon a 12'-7" deep medium compacted silty-sand soil stratum. On the wall near side (N.S.), the wall has three horizontal strips of 3" wide corrugated indentations. The strip vertical widths are (from top to bottom): 4'-7", 2'-0" and 4'6". The corrugated indentations have aesthetic value to the I-696 freeway corridor but play no real role in the structural behavior of the retaining wall system (MDOT, 1986). The construction of the wall panel occurred in multiple stages. First the footing was cast after which the primary retaining wall was cast in two stages to a height of 18'7". The concrete retaining wall is tapered with a thickness of 3'-0" at the footing and 1'-10" at the top. The last stage of construction was the placement of a 9'-11" tall parapet wall 1'-5" thick. A cross section view of the wall system is shown in Figure 3.2b. At the street level, the F.S. surface of the parapet wall is later bricked to enhance the aesthetic of the wall.

Both horizontal and vertical steel reinforcement system is placed in the poured concrete wall as shown in Figure 3.2c. To accommodate the high tensile response of the F.S. of the wall, the vertical steel reinforcement is denser on the F.S. of the wall. Furthermore, the vertical steel reinforcement runs continuously through the vertical height of the wall and ensures full compatibility of the three wall segments. The wall reinforcement is further summarized in Table 3.1.

Understanding the structural design and geotechnical conditions of the tall wall panel is critical to interpreting the structural response of the wall. First, the soil conditions of the lower portion of the backfill soil at the bottom of the wall is less pervious than the top layer; this will strongly influence

<b>Rebar Name</b>	Size (or Diameter)	Length	Spacing	<b>Rebar Shape</b>
		Bottom		
A062100	#6 (or <sup>3</sup> / <sub>4</sub> ")	21'-0"	18" (F.S.)	Straight
B092006	#9 (or $1^{1/8}$ ")	20'-6"	18" (F.S.)	L-shaped
B091109	$\#9 \text{ (or } 1^{1/8})$	11'-9"	18" (F.S.)	L-shaped
		Mid		
EA062309	#6 (or <sup>3</sup> /4'')	23'-9"	18" (N.S.)	Straight, Epoxy Coated
		Тор		
EA060609	#6 (or <sup>3</sup> /4'')	6'-9"	9" (F.S.)	Straight, Epoxy Coated
EA060900	#6 (or <sup>3</sup> /4'')	9'-0"	18" (N.S.)	Straight, Epoxy Coated

Table 3.1. Reinforcement bars of the tall panel of the I-696 wall system.



Figure 3.2. Dimensions and structural details of the 'tall' wall panel of the I-696 retaining wall site: (a) front side elevation showing sensor locations; (b) vertical sectional profile; (c) horizontal sectional plan.

the hydrostatic pressures on the wall back surface. Also, the lower portion of the wall has a much higher flexural rigidity than the upper portions to accommodate the higher flexural moments. In contrast, the top parapet wall is thinner and more lightly reinforced.

#### 3.3.2. Daily Mean Responses

As previously discussed, the tall I-696 wall panel was monitored continuously from August 2018 to October 2019 with the wall tilt at its top and mid-height (collected by wireless sensor nodes 1 and 2, respectively, in Figure 3.2a), wall flexural strain at the bottom and mid-height (collected by wireless sensor nodes 3 and 4, respectively, in Figure 3.2a), and wall surface temperature at the bottom and mid-height (collected by wireless sensor nodes 3 and 4, respectively, in Figure 3.2a). The tilt response (using equal scales on tilt magnitude) is plotted in Figure 3.3, strain (on equal scales) in Figure 3.4, and wall temperature in Figure 3.3 and Figure 3.4. The daily means were calculated for each day by averaging the response data received over a 24 hour period from 12:00am to 11:59pm. To provide context to the data, the daily average precipitation was also extracted from an online weather database from a weather station in close proximity to the walls (Weather Underground, 2019). The daily total precipitation is plotted in Figure 3.3 and Figure 3.4. It should be noted that the strain measurements in Figure 3.4 are the raw strain readings from the strain sensors. With the residual strain in the wall at the start of data collection unknown, the strain was taken as zero at the start of data collection.

The tilt history of the top portion of the wall system demonstrated much greater variation in daily mean tilt as compared to the mid-height tilt. The top tilt varied from 0.5 to  $3.5^{\circ}$  while the mid-height tilt had a much smaller variation between 1.0 to  $1.45^{\circ}$ . The top-level tilt node was installed on the top of the thin parapet wall just 2' below the wall top (Figure 3.2a). The wide variation in tilt of the parapet wall was attributed to both thermal variations (*e.g.*, thermal expansion of the soil backfill) and varying hydrostatic pressures associated with precipitation and other factors. Specifically with respect to precipitation, the wall appeared to be sensitive to repeated days of precipitation resulting in the build-up of hydraulic pressure in the top stratum of soil and corresponding higher tilts on the top of the wall. For example, continuous days of rain in late September 2018 into early October 2018 induced a noticeable upper tilt of the top portion of the wall (going from 1.0 to 2.5°); after rain ceased, the wall returned back to 1.0°. The daily mean tilt of the lower portion of the tall wall was less sensitive to precipitation with little variations in daily mean wall tilts during periods of rain. This may be attributed to the high flexural rigidity of the wall; it may also be explained by the lack of variation in the hydrostatic pressures in the lower soil stratum behind the wall.


Figure 3.3. The daily mean response of the I-696 tall wall panel: top tilt, mid-height tilt, precipitation, and wall temperatures (August 2018-October 2019)

From late November 2018 to January 2019, the top tilt had a high level of day-to-day variation as the trendline mean of the tilt time history increased slowly. It is also noted that the wall daily mean top tilt dramatically varied in mid-January 2019 to mid-February 2019 when the wall temperature was near or below freezing (32 °F). In fact, the last few days of January 2019 saw the wall achieve a temperature of 0 °F after which a few days later the temperature was 42 °F; during this period the daily mean top wall tilt varied from 0.5 to 3.5°. By May 2019, the wall reached a maximum daily mean top tilt of 3.5°. After May 2019, the tilt at the top of the wall had less day-to-day variations and the mean trendline of the time history decreased to about 1.5° by July 2019. It was hypothesized that the daily mean top wall tilt trendline slowly increased from November to May due to lowering ambient temperatures and their effects on the backfill soil. Comparing the daily mean top tilt trendline with the wall temperature time history, the two appear to be correlated with a 30 to 45 day lag; this may be attributed to thermal inertia of the backfill.



Figure 3.4. The daily mean response of the I-696 tall wall panel: mid-height strain, bottom strain, precipitation, and wall temperatures (August 2018-October 2019)

Overall, when the full 15 months of monitoring are considered, the parapet wall tilt exhibited sensitivity to precipitation in the late-Spring to mid-Fall season. During the Winter and early-Spring, the wall is less sensitive to precipitation and has a trend line correlated to wall temperature with a lag.

The daily mean strain responses at the wall mid-height and bottom captured changes in strain of the wall façade on the wall front side; the absolute state of strain was unknown. Taking compressive strain to be of negative magnitude and tensile strain to be of positive magnitude, the daily mean wall strain histories (Figure 3.4) exhibited a trend correlated to the wall temperature. The mid-height strain varied over the 15 month period a total of 125  $\mu$ s while the bottom strain varied only 75  $\mu$ s. Maximum compressive flexural strain (which would correspond to maximum tensile strain on the wall far side was during the winter).



Figure 3.5. Night and day daily mean response of the I-696 tall wall panel: top tilt and mid-height tilt (August 2018-October 2019)



Figure 3.6. Night and day daily mean response of the I-696 tall wall panel: mid-height strain and bottom strain (August 2018-October 2019)

Figures 3.3 and 3.4 plotted the daily mean response of the tall wall panel. The daily variations were also investigated. It was observed that there was an interesting difference between the behavior of the wall during the night (11:59pm to 6:00am) and day (11:00am to 5:00pm). Figure 3.5 plots the top and mid-height average tilts during the night and day periods. Similarly, Figure 3.6 plots the mid-height and bottom average strains during night and day periods. As is evident from Figure 3.5, the evening tilt at the wall top and mid-height is noticeably higher than the average day tilt throughout the observation period. However, the difference in average night and day top tilt is especially pronounced from late March 2019 to October 2019. The strain

measurements also had a noticeable difference between night and day with higher compressive strain in the night. The cause for this discrepancy is unknown. One hypothesis is that water is present in the night that is not there in the day adding temporary hydraulic pressure on the wall backside; the source of this water is unknown and could be from a failed water utility that is buried behind the wall.

### 3.3.3. Response Scatter Plots

In the previous section, the behavior of the tall I-696 wall was described over the 15 month monitoring period. In this section, the causality between environmental parameters and the wall behavior are studied using response scatter plots with linear regressed behavioral models fit. As previously described, the lower portion of the tall wall panel appeared to be sensitive to temperature. Plotted in Figure 3.7 is a scatter plot of lower wall responses (i.e., bottom strain, mid-height strain, and mid-height tilt) as a function of measured wall temperature. As shown, a strong linear relationship exists between these two measurands. Using linear regression to model the relationship, it is evident that the mid-height tilt of the lower portion of the wall varies roughly  $0.004^{\circ}$  per degree F of wall temperature. The greatest tilt is experienced during the colder months with maximum tilt observed at  $1.4^{\circ}$  when the wall temperature is 0 °F. Similarly, the wall strain at the bottom and mid-height was dependent on temperature. Based on linear regression, the mid-height and bottom strain varied 0.92 µ $\epsilon$  and 0.45 µ $\epsilon$  per degree F, respectively.

The lower portion of the retaining wall was relatively insensitive to precipitation supporting the hypothesis that the lower portions of the wall backfill are saturated. Recall, this hypothesis was supported by visual observation of steady weeping in the wall panels in their lower sections. Shown in Figure 3.8, the mid-height tilt and bottom strain of the tall wall panel were plotted as a function of cumulative precipitation. Cumulative precipitation is taken as a weighted sum of prior precipitation that has been designed to model the time delay for rain to permeate in the soil and to develop hydrostatic pressure in the wall system. After periods of no precipitation, the cumulative precipitation model also assumes drying of the soil resulting in the alleviation of hydrostatic pressure. In this study, it is assumed the cumulative rain,  $C_{rain}$ , is:

$$C_{rain} = \sum_{j=0}^{i-1} \frac{j}{i} D(j) + \sum_{j=i}^{T} e^{\alpha(i-j)} D(j)$$
(3.1)

where *D* is a daily time series of precipitation with *j* serving as an index that begins at 0 and marches backwards (*e.g.* 3 days prior is j=3), *i* is a constant that reflects the time it takes (in days)



Figure 3.7. Relationships between I-696 tall wall mid-height tilt, mid-height strain, and bottom strain as a function of wall temperature (August 2018-October 2019)



Figure 3.8. Relationships between I-696 tall wall mid-height tilt and bottom strain as a function of cumulative precipitation (August 2018-October 2019)

for soil to saturate the backfill, *T* is a constant that reflects the time it takes (in days) for the soil to dry after being saturated, and  $\alpha$  is a time constant on the tail portion of the weighted sum. Using the wall top tilt and precipitation measurements collected during period of heavy precipitation (*e.g.* late September 2018 into early October 2018), the cumulative rain function is emperically determined with *i*=3 days, *T*=18 days and  $\alpha$ =0.1. As shown in Figure 3.8, the mid-

height tilt and bottom strain of the tall wall panel was insensitive to cumulative rain, reinforcing the hypothesis of a saturated back fill.

The behavior of the top portion of the tall wall panel, especially the parapet portion of the wall system, was observed to be sensitive to precipitation in the non-winter months while sensitive to temperature in the winter when there is less precipitation. Figure 3.9 plots the tall wall top tilt as a function of precipitation from August 25 to November 30, 2018 and April 1 to October 11, 2019



Figure 3.9. Relationships of the I-696 tall wall top tilt to cumulative precipitation (August 25 to November 30, 2018), temperature (January 1 to March 30, 2019) and cumulative precipitation (April 1 to October 11, 2019)

revealing a linear relationship between top tilt and cumulative precipitation during the nonwinter observation period. The figure also plots the top wall tilt as a function of wall temperature in the winter (January 1 to March 30, 2019) revealing a fairly strong linear relationship of 0.05° per degree F.

## 3.3.4. Wall Deflection Curves

The prior sections explored analyses of the tall I-696 retaining wall panel responses. Both response time histories and response scatter plots were presented and discussed. This section considers an analysis of the wall deflections based on the top and mid-height daily mean tilt measurements. The deflected shape of the wall can be calculated from the top and mid-height tilt measurements under an assumed lateral earth pressure profile and structural properties. Structural properties including wall geometries, steel reinforcement locations, and material properties can be reasonably estimated based on the engineering drawings of the wall. The shape of the lateral earth pressures on the wall backside can also be estimated. In this analysis, three

pressure profiles were assumed: lateral earth pressures from surcharge loads on the top surface (rectangular), backfill pressure (triangular), and hydrostatic pressure (triangular) as shown in Figure 3.10. Estimating the deflected shapes of the wall can aid in understanding the seasonal variations observed in the data, especially the wall responses to hydrostatic pressures associated with precipitation.

The lateral earth pressures from the backfill soil induce a flexural response of the wall. The top wall rotation,  $\theta$ , due to the backfill earth pressures is static. However, the surcharge and hydrostatic lateral earth pressures vary depending on the surface gravity loads on the top of the backfill and saturation of the soil, respectively. Variations in these loads result in a variation in the top tilt,  $\Delta\theta$ .



Figure 3.10. Cross-sectional view of the I-696 tall wall and its backfill lateral loads

In this section, the theoretical deflected shape of the wall panel are presented for each of the three lateral earth pressures: backfill pressure, hydrostatic pressure, and surcharge pressure.

Using the tilt measurements collected, the wall deflected shapes were estimated using the analytical relationship between rotation and deflection.

To compute,  $\theta$ , the backfill information from soil boring elevation drawings was utilized. According to MDOT (1986), the backfill was a two-layer system with a 13'-11" thick medium compacted sand layered on top of a 9'-7" thick medium compacted gray silt and fine sand. The soil within each layer was assumed to have a uniform/homogeneous material distribution. The range of bulk and submerged densities of the two soil types are summarized in Table 3.2 (Yu et al, 1993). For analysis, the following properties are selected:

- Sand with medium compaction:
  - Bulk density =  $0.069 \text{ lb/in}^3$  (submerged density =  $0.038 \text{ lb/in}^3$ )
- Sandy silt with medium compaction:
  - Bulk density =  $0.073 \text{ lb/in}^3$  (submerged density =  $0.042 \text{ lb/in}^3$ )

Table 3.2. Typical bulk and submerged densities of selected soils (Based Coduto, 2001)

Soil Type	Bulk Density (lb/in <sup>3</sup> )	Submerged Density (lb/in <sup>3</sup> )
Firm silty sandy clay	0.058 to 0.085	0.022 to 0.049
Medium compacted sand	0.049 to 0.079	0.030 to 0.042
Medium compacted sandy silt	0.051 to 0.082	0.031 to 0.046

To estimate the backfill earth pressures, Coulomb or Rankin theory can be used. Due to the unknown frictional properties of the soil on the wall surface, Rankine theory was used. Rankine theory is also favored by state transportation agencies including AASHTO and FHWA. The Rankine active earth pressure coefficient,  $k_a$ , is given by (Das, 2011):

$$k_a = \frac{\cos(\beta - \theta)\sqrt{1 + \sin^2 \phi - 2\sin \phi \cos \psi}}{\cos^2 \theta (\cos \beta + \sqrt{\sin^2 \phi - \sin^2 \beta}}$$
(3.2)

$$\psi = \sin^{-1} \left( \frac{\sin \beta}{\sin \phi} \right) - \beta + 2\theta \tag{3.3}$$

where  $\beta$  is the slope of the top surface of backfill relative to the horizon (in this case  $\beta=0^{\circ}$ ),  $\phi$  is the internal soil friction angle, and  $\theta$  is the slope of the backwall incline (relative to vertical). Typical angles of internal soil friction were acquired from (Koloski, 1989):

- Sand with medium compaction:  $\phi = 33^{\circ}$
- Sandy silt with medium compaction:  $\phi = 29^{\circ}$

The angle of the back surface of the I-696 walls,  $\theta$ , was computed as to be 3.6°. Based on Equation 3.2 and 3.3, the active earth pressure coefficients were computed as:  $k_{a1}$  =0.30 and  $k_{a2}$  =0.35 for the upper and lower soil layers, respectively. Using the active earth pressure coefficients, depths of the backfill layers ( $h_1$  and  $h_2$  referenced from the top surface of each layer), and bulk soil density,  $\rho$ , the backfill earth pressure profile (Figure 3.10) was calculated as shown in Table 3.3.

The calculations presented in Table 3.3 apply when the backfill soil is acting on the wall in an unsaturated state and with no acting surcharge load. When the backfill has water to a specified saturation level, the active lateral pressure from the backfill soil will need to consider the submerged density of the soil in lieu of the bulk density for the portions of the backfill that are saturated (i.e., below the ground water table or GWT level) (Figure 3.11 and Table 3.4).

Equation	Calculation	<b>Pressure Value</b>
$w_1 = k_{a_1} \rho_{bulk_1} h_1$	$w_1 = 0.3 * 0.069 \frac{lb}{in^3} * \left(13.95  ft. *  12 \frac{in}{ft.}\right)$	$w_1 = 3.47  psi$
$w_2 = \frac{k_{a_2}}{k_{a_1}} * w_1$	$w_2 = \left(\frac{0.35}{0.30}\right) * 3.47 \ psi$	$w_2 = 4.04  psi$
$w_3 = w_2 + k_{a_2} \rho_{bulk_2} h_2$	$w_3 = w_2 + 0.35 * 0.073 \frac{lb}{in^3} * \left(9.55  ft. *  12 \frac{in}{ft.}\right)$	$w_3 = 6.97  psi$

 Table 3.3. Static backfill earth pressures (no surcharge or hydrostatic pressure)

The flexural rigidity of the wall was next determined assuming the wall acts in flexural response as an ideal vertical cantilever. The horizontal flexure of the wall panels between their vertical joints was ignored to simplify the analysis. The vertical flexural rigidity of the wall is dependent on the wall moment of inertia, *I*, and the elastic modulus, *E*. Given the age of the I-696 retaining wall system, the material properties need to account for age, especially the modulus of elasticity for the concrete used in the wall construction. The effective modulus of the concrete was determined based on the documented 28-day compressive strength of the concrete ( $f'_c$ =4,000 psi) and consideration of creep and shrinkage (ACI 1997). The effective modulus, *E<sub>e</sub>*, is based on the initial modulus,  $E_i = 57,000\sqrt{f'_c}$  (in psi) and the creep coefficient,  $\vartheta_t$ , with *t* specified in days:

$$E_e = \frac{E_{ci}}{1 + \vartheta_t} \tag{3.4}$$

$$\vartheta_t = 2.35 * \frac{t^{0.6}}{1 + t^{0.6}} \tag{3.5}$$

Given the age of the wall (33 years), the effective modulus,  $E_e$ , was determined to be  $1.1 \times 10^6$  psi based on Equation 3.4.

The section moment of inertia,  $I_{cr}$ , was based on the assumption of a cracked section with the tensile response of the wall entirely taken by the steel reinforcement. This was a valid assumption given the age of the wall, the large loads present, and visual evidence of flexural cracking in previously excavated retaining walls (such as the M-10 wall system) in the vicinity of the I-696 wall (MDOT, 2013). Figure 3.12(a) highlights the assumption of the cracked section. To determine



Figure 3.11. Static backfill earth pressures, surcharge and hydrostatic pressure: (a) higher versus (b) lower GWT

Equations (GWT in top layer)	Equations (GWT in bottom layer)
$w_1 = k_{a_1} \rho_{bulk_1} * (23'6" - h) + \rho_{sub_1} * (h - 9'7")$	$w_1 = k_{a_1} \rho_{bulk_1} * 13'11"$
$w_2 = \frac{k_{a_2}}{k_{a_1}} * w_1$	$w_2 = \frac{k_{a_2}}{k_{a_1}} * w_1$
$w_3 = w_2 + k_{a_2} \rho_{sub_2} * 9'7"$	$w_3 = w_2 + k_{a_2} \rho_{bulk_1} * (9'7'' - h) + \rho_{sub_1} * h$
$w_4 = \rho_{water}(h - 9'7")$	$w_4 = 0$
$w_5 = \rho_{water}h$	$w_5 = \rho_{water}h$
$w_6 = k_{a_1} * q$	$w_6 = k_{a_1} * q$
$w_7 = \frac{k_{a_2}}{k_{a_1}} * w_6$	$w_7 = \frac{k_{a_2}}{k_{a_1}} * w_6$

Table 3.4. Formulae to calculate backfill, surcharge and hydrostatic pressures

the moment of inertia of the cracked section,  $I_{cr}$ , the location of the neutral axis,  $\bar{y}$ , must be calculated for the tapered section as a function of the wall height, x. Using the structural drawings to identify the wall geometries including reinforcement details, the cracked section was transformed based on the ratio of elastic modulus, n, between the steel ( $E_s = 29 \times 10^6$  psi) and concrete ( $E_e = 1.1 \times 10^6$  psi) (Figure 3.12(b)). The neutral axis,  $\bar{y}$ , was then parameterized as the following polynomial given the wall tapering (which affects the depth to the tensile



Figure 3.12. (a) Cracked section assumption with tensile load taken by vertical steel reinforcement and compressive load taken by concrete; (b) cracked section transformation.

reinforcement, d(x) and the effective area of the steel in the tension zone,  $A_{s,eff}$ , and the compression zone,  $A'_{s,eff}$ :

$$\frac{1}{2}b\bar{y}^2 + (n-1)A'_s(\bar{y} - d') - nA_s(d(x) - \bar{y}) = 0$$
(3.6)

As shown in Figure 3.13, the tall I-696 wall panel has flexural reinforcement that varies over its height. As a result, the wall can be divided into three sections that will have separate formulas that offer the real positive root to Equation 3.6.

The neutral axis,  $\bar{y}(x)$ , was found as a function of height, *x*, by finding the real positive root of the polynomial of Equation 3.6. The cracked moment of inertia about the neutral axis,  $\bar{y}(x)$ , was found to be:

$$I_{cr}(x) = (n-1)A'_{s,eff}(\bar{y}(x) - d')^2 + nA_{s,eff}(d - \bar{y}(x))^2 + b\bar{y}(x)\left(\frac{\bar{y}(x)}{2}\right)^2 + \frac{b\bar{y}(x)^3}{12}$$
(3.7)

As before, Figure 3.13 summarizes the cracked section moment of inertia as a function of wall height, x.



Figure 3.13. (a) Tapering of the I-696 cantilever RC retaining wall and the parapet wall; (b) Depth to Neutral Axis (NA) and Cracked Moment of Inertia (I<sub>Cr</sub>)

The I-696 tall wall was assumed to be in static equilibrium governed by the following moment balance:

$$E_e I_{cr}(x) \left(\frac{d^2 y(x)}{dx^2}\right) = E_e I_{cr}(x) \left(\frac{d\theta(x)}{dx}\right) = M(x)$$
(3.9)

where M(x) is the flexural moment induced by the backfill, y(x) is the displacement of the wall, and  $\theta(x)$  is the wall tilt. Similarly, the wall responded to moments from hydrostatic pressure and top-grade surcharge,  $\Delta M(x)$ , which induced a time-varying displacement,  $\Delta y(x)$ , and tilt,  $\Delta \theta(x)$ :

$$E_e I_{cr}(x) \left(\frac{d^2 \Delta y(x)}{dx^2}\right) = E_e I_{cr}(x) \left(\frac{d \Delta \theta(x)}{dx}\right) = \Delta M(x)$$
(3.10)



Figure 3.14. Lateral earth pressures on tall I-696 wall: backfill, surcharge, hydrostatic

Three load profiles were explored on the I-696 tall wall panel: backfill lateral earth, hydrostatic pressure, and pressure from surface surcharge. The maximum lateral load on the wall is when all the three loads act on the wall, simultaneously (Figure 3.14). With the rise of water in the backfill to the level close to the top surface, the bulk backfill pressures that were calculated in Table 3.3 need to be recalculated by utilizing the submerged unit weight of the soil layers (Table 3.2) and converted to a load diagram for the wall considering the width of the wall (20'). A surcharge pressure of q = 1.7 psi (shown in Figure 3.10) was considered on the top surface of the backfill resulting in a uniformly distributed lateral earth pressure on the wall ( $w = k_a q$ ). This surface surcharge was conservatively obtained from the load assumptions made during the design of retaining walls (FHWA, 2009). Finally, hydrostatic pressure was also considered assuming a fully saturated backfill using the specific weight of water to be 62.43 lb/ft<sup>3</sup> (Figure 3.14).

Using the equilibrium condition of the wall, the wall deflection was theoretically predicted. Equation 3.9 was used to derive the wall deflection, y(x), for four load cases to offer a range of feasible deflection scenarios: 1) backfill only; 2) backfill and surcharge; 3) backfill and

hydrostatic (fully saturated); and, 4) backfill, surcharge and hydrostatic (fully saturated). Depending on the



Figure 3.15. Discretization of the I-696 cantilever wall

level of the GWT delineated in Figure 3.11, there were different moment equations for the different sections of the tall I-696 wall system. Therefore, the analysis of the wall was simplified by discretizing the wall into 1" sections ( $\delta h$ ) with equilibrium applied (Equation 32.9) as shown in Figure 3.15. While the wall could be discretized more finely, the 1 inch discretization size was found to be sufficiently precise. The tilt and displacement of the wall section at the top of the discretized element, *k*, was:

$$\theta(k+1) = \sum_{i=1}^{k} \frac{M(i)}{E_{eff} I_{cr}(i)} = \theta(k) + \delta h \frac{M(k)}{E_{eff} I_{cr}(k)}$$
(3.11)

$$y(k+1) = \sum_{i=1}^{k} \theta(i)\delta h = y(k) + \theta(k)\delta h$$
(3.12)

The total bending moment acting on the reinforced concrete cantilever wall panel at a height of el ement, k, is the superposition of the bending moments from the hydrostatic pressure,  $M_{HS}(k)$ ,

the surcharge,  $M_q(k)$ , and the backfill,  $M_b(k)$  at that level. Therefore, the total bending moment, M(k) on the tall I-696 wall was expressed as,



Figure 3.16. (a) Actual (left) and (b) theoretical (right) deflected shapes of the tall I-696 RC cantilever wall.

$$M(k) = M_{HS}(k) + M_a(k) + M_b(k)$$
(3.13)

To analyze the behavior of the wall, the finite difference method encapsulated in Equation 3.11 was applied for a given assumed surcharge, q, and height of saturated soil,  $h_{sat}$ . Given q and  $h_{sat}$ , the pressure on the wall was formulated by Table 3.4 and the moments determined by Equation 3.13. Equation 3.11 was analyzed using  $E_c I_{cr}(k)$  for each discrete element working from the base to the top of the wall to determine tilt,  $\theta(k)$ . Finally, the displacement, y(k), was calculated by Equation 3.12 using tilt.

The tall I-696 wall was monitored with two measures of tilt (*i.e.*, at the top and mid-height of the wall). These tilt measurements were used to assess the deflection curve of the wall based on the finite difference model developed. The two measurements of tilt were used to estimate two parameters of the lateral pressures on the wall: surcharge, q, and the height of the water

(a + a)

saturation,  $h_{sat}$ . The two unknown parameters, q and  $h_{sat}$ , were determined by searching through a look-up table composed of the mid-height and top tilt for all variations of q and  $h_{sat}$ pre-calculated. For each tilt measurement pair, q and  $h_{sat}$  were identified. The deflected shape, y(k), of the wall using these load parameters was determined. Figure 3.16a shows the actual deflection curve extracted from the measured tilts. The minimum and maximum curves are plotted. Figure 3.16b shows the theoretical deflected shapes of the retaining wall using the flexural rigidity and prescribed loading scenarios. The saturation of the soil is recognized to be a conservative load assumption (fully saturated backfill,  $h_{sat} = 23.5'$  and q = 1.74 psi) and was considered as an upper bound on the wall response (Figure 3.16). For the actual deflection cases, it was discovered that the level of backfill soil saturation,  $h_{sat}$ , was dominant in increasing tilt significantly as compared to the presence of surcharge, q. The minimum deflection profile corresponded to  $h_{sat} = 0'$  and q = 1.74 psi, while the maximum deflection corresponded to  $h_{sat} = 21.5'$  and q = 1.74 psi.

## 3.4. Short I-696 Retaining Wall System

The second wall system instrumented was the "short" I-696 wall; this wall system is 40 feet to the east of the tall I-696 wall. The wall panel is structurally *identical* to the tall I-696 wall system (as described in Section 3.3) including wall dimensions, reinforcement details and backfill soil information. However, the exit ramp coming off of the I-696 freeway is inclined resulting in a higher soil profile on the front face of the wall. As a result, the front grade line of the short I-696 wall panel is higher by 2'-6" compared to the tall wall. The top of the short wall was measured during site visits to be 23'-6" above the grade line.

The tilt response (using equal scales on tilt magnitude) of the short I-696 wall panel is plotted in Figure 3.17, raw strain (on equal scales) in Figure 3.18, and wall temperature in Figure 3.17 and Figure 3.18. As was done before, the daily means were calculated by averaging the measurement data over a one day period from 12:00am to 11:59pm. The daily total precipitation is also plotted over the same period in Figure 3.17 and Figure 3.18 to provide insight to responses associated with precipitation. The raw strains plotted in Figure 3.18 were nulled to be zero at the start of data collection. Only one strain sensor was installed at the bottom of the short I-696 wall panel.

The top and mid-height tilts are plotted in Figure 3.17. As can be seen, the mid-height tilt sensor stopped operating in early December so only three months of data were collected. The top tilt exhibited greater variability as compared to the mid-height tilt sensor during that time, an observation consistent with the tall I-696 wall. However, the mid-height tilt on the short wall was



Figure 3.17. The daily mean response of the I-696 short wall panel: top tilt, mid-height tilt, precipitation, and wall temperature (August 2018-October 2019)

significantly greater than that on the tall wall, suggesting more movement in the short wall panel. Over the one year period, the top tilt varied from 0.3 to 4.1° while the mid-height tilt had a much smaller variation between 0.4 to 2.5° over the three months of measurements. Similar to the tall I-696 panel, the short I-696 wall panel appeared to be sensitive to repeated days of precipitation. Steady precipitation results in the build-up of hydraulic pressure in the top stratum of soil inducing larger tilts on the top of the wall. For example, continuous days of rain in late September 2018 into early October 2018 induced a noticeable upper tilt of the top portion of the wall from 1.5 to 2.5° with the wall returning back to 1.5° after rain stopped. From early November 2018 to January 2019, the top tilt had a high level of day-to-day variation as the

trendline mean of the tilt time history steadily rose. In mid-January to mid-February during a time period of extreme cold temperatures (when the wall surface temperature is below freezing), the daily mean top tilt varied from 0.5 to 3.1°. By mid-March 2019, the top tilt of the short wall had less variation with a mean trendline up near 2.8°. Thereafter, the wall top tilt slowly reduced as the wall temperature steadily increased. By early August 2019, the top tilt has settled to about 1.5° with low day to day variation.



Figure 3.18. The daily mean response of the I-696 short wall panel: bottom strain, precipitation, and wall temperature (August 2018-October 2019)

For the remainder of the measurement period, the wall showed greater variation in top tilt. These observations in the tilt response were similar to those made for the tall I-696 wall panel. The daily mean strain response at the bottom of the short I-696 wall panel captured strain variations and do not represent an absolute state of strain. The daily mean strain (Figure 3.18) exhibited a trend correlated to the wall temperature. The bottom strain of the short wall panel varied only 60  $\mu\epsilon$  which was less than the bottom strain experienced in the tall wall panel (which was 75  $\mu\epsilon$ ).

Due to the observations of difference in the response of the tall I-696 wall panel during night (11:59pm to 6:00am) and day (11:00am to 5:00pm), a similar analysis was performed on the short panel. Figure 3.19 plots the top and mid-height average tilts while Figure 3.20 plots bottom average strain during night and day. As is evident from Figure 3.19, the tilt of the short wall at

average strain during night and day. As is evident from Figure 3.19, the tilt of the short wall at the top and mid-height was generally higher at night than during day. However, the top of the wall had higher tilt during the day from December 2018 to mid-January 2019 and from mid-March to May 2019, which is different than that observed in the tall wall panel.



Figure 3.19. Night and day daily mean response of the I-696 short wall panel: top tilt and mid-height tilt (August 2018-October 2019)



strain and bottom strain (August 2018-October 2019)

The strain measurements also had a slight difference between night and day with higher compressive strain in the night likely due to lower night temperatures.

Scatter plots were studies to understand the causality between environmental parameters and wall behavior. Scatter plots similar to those constructed for the tall I-696 wall panel are done for the short I-696 wall panel. First, the daily mean bottom strain response of the short wall was plotted as a function of temperature. Figure 3.21 plots the bottom strain wall strain as a function of wall temperature. A strong linear relationship was evident with a sensitivity roughly  $0.5 \ \mu\epsilon$ 

to cumulative precipitation,  $C_{rain}$  (Equation 3.1). The scatter plot of strain versus  $C_{rain}$  in Figure 3.21 shows no dependencies.

Top tilt of the short I-696 wall was compared to top tilt of the tall I-696 wall. A strong correlation was evident between the two wall panels as shown in Figure 3.22. Also plotted is the scatter plot



Figure 3.21. Relationships between I-696 short wall bottom strain as a function of temperature and cumulative precipitation (August 2018-October 2019)



Figure 3.22. Relationships between I-696 short and tall wall panels: (a) mid-height tilt (August 2018-December 2018) and (b) top tilt (August 2018-October 2019)

of mid-height tilt of the short and tall wall over the period the short wall mid-height was measured. While less data was collected, there was a mild correlation between the two. These findings were not surprising given the close proximity of the two panels from one another. In Figure 3.23, the theoretical and estimated by measurement deflection curves for the short I-696 wall are presented. The same procedure was followed to compute the actual deflections by looking at minimum and maximum deflection cases based on the top tilt measurements. For the actual deflection cases, same as the tall I-696 wall, it was discovered that the level of backfill soil saturation ( $h_{sat}$ ) was dominant in increasing tilt significantly as compared to the presence of surcharge (q). The minimum deflection case corresponded to  $h_{sat} = 0$ ° and q = 0.87 psi, while the maximum case corresponded with  $h_{sat} = 22.5$ ° and q = 1.74 psi.



Figure 3.23. (a) Actual (left) and (b) theoretical (right) deflected shapes of the short I-696 RC cantilever wall.

## 3.5. M-10 Retaining Wall System

## 3.5.1. Description of the Wall System

The M-10 wall is more than 50 years old and is a reinforced concrete (RC) retaining wall system with panels 25' wide. Construction of the wall was in multiple stages. First, a 14' tall (above foundation pad) and 1'6'' thick reinforced concrete retaining wall was erected upon a 2'-6''

thick and 7'-9" wide foundation pad. Second, a 8'-0" tall wall with a horizontal beam at the bottom portion (which the tie-backs are attached to) was formed in the second cast along with a 2'-0" overhand barrier. The second pour of the wall was also 1'-6" thick except for the horizontal beam section which was 1'-10" wide. Details of the steel reinforcement is summarized in Table 3.5 while Figure 3.24 summarizes the structural details of the wall including its geometry and steel reinforcement. Each panel is stabilized with a post-tensioned (PT) battered caisson tie-backs (30° from vertical). The tiebacks are spaced every 15' to 18' along the M-10 corridor. The instrumented



Figure 3.24. Structural details of the M-10 wall: (a) front elevation; (b) sectional profile; (c) cross-sectional profile detailing reinforcement.

panel shown in Figure 3.24a had two tie-backs stabilizing it; each tieback was 5'-0" from the vertical edge of the panel (15'-0" apart). The wall was measured during a site visit in June 2017 to be 22'-6" tall relative to grade on the front side of the wall. From the working drawings, the

height of wall above the 2'6'' thick foundation pad was 24' (Figure 3.24a). On the wall back side (B.S.),

Rebar Name	Size or diameter	Length	c/c spacing	Rebar shape
Main Wall				
A22	#5 or <sup>5</sup> / <sub>8</sub> "	23'	18" (N.S.)	Straight
A23	#6 or $3/4$ ,	23'	18" (F.S. and N.S.)	Straight
Submerged Beam				
A24	#7 or <sup>7</sup> / <sub>8</sub> "	21'	9" (N.S.)	Straight
A25	#6 or <sup>3</sup> /4''	21'	9" (F.S.)	Straight

Table 3.5. Reinforcement bars of the wall panel of the M-10 wall system.

there exists a silty sandy clay backfill soil system. There were different levels of compaction (*i.e.*, firm to hard) throughout the height of the backfill but the backfill was analyzed as a single homogeneous layer considering the internal soil friction angle  $(34^{\circ})$  (MDOT, 1960).

The sensors were installed along the centerline of the panel centered between the two tiebacks. A summary of the sensors installed on the M-10 panel is shown in Figure 3.24a. There are two tilt units at the top and mid-height levels (*i.e.*, wireless sensor units 8 and 9) of the wall. There are also two strain gage-thermistor units at the wall mid-height and bottom levels (*i.e.*, wireless sensor units 10 and 11). Data was collected from November 2018 to October 2019.

## 3.5.2. Preliminary Observations from 2012 and 2013

The M-10 wall was instrumented due to a past history of structural failures. In 2012, it was reported that wall panels on the northbound side of the M-10 corridor close to Schaefer Highway (roughly



Figure 3.25. Preliminary investigation of the M-10 wall in 2012-2013: (a) tilt measurement during site visit; (b) failure of tie-back behind the wall (MDOT, 2013).

100'-0" north of Schaefer) had moved in mid-July. The initial thoughts as to the cause included the possibility of excessive hydrostatic pressure from an undrained backfill. MDOT performed an excavation of the wall backside and discovered the water main under the service road was leaking water into the backfill soil adding the hydraulic pressure. When the wall was measured, the wall had tilted 4" at its top (Figure 3.25a). Tilt monitoring was performed on the wall panels. Initially, the wall sections did not show significant variations but over a longer period from Fall 2012 to Summer 2013, significant changes in the wall tilt was observed. The monitoring showed a clear trend of movement suggesting to engineers that a progressive failure was underway. Upon excavation, the tieback was found to have been corroded to an extent that it was no longer engaged to the wall panel and restraining its motion (Figure 3.25b). The failed section of the M-10 wall was later replaced with a new wall section (MDOT, 2013).



Figure 3.26. Wall response time histories of the M-10 wall: top-level tilt history; midheight level tilt history; daily precipitation and wall temperature history at the midheight level

## 3.5.3. Daily Mean Responses and Scatter Plots

The time histories of the daily mean wall responses of the M-10 wall panel at Thatcher Avenue near Schaefer Highway are presented in Figure 3.26. The daily mean tilt responses from the two wireless sensor units (i.e., wireless sensor nodes 8 and 9 in Figure 3.24a) at the top and midheight locations are presented in Figure 3.26. Figure 3.26 also presents recording of the precipitation at the site based on data collected from an online weather database (Weather Underground, 2019) and the measured daily mean wall temperature at the mid-height location (i.e., wireless sensor node 10 in Figure 3.24a). Figure 3.27 plots the strain response of the M-10 wall panel along with precipitation and temperature.

The top tilt of the M-10 wall panel showed a high level of daily variation (similar to that observed for the I-696 walls) with tilt measuring from 1.1 to 2.0° with a mean trendline that appeared to be seasonally dependent (i.e., greater tilt in the winter and less tilt in the summer). The mid-height tilt was less variable measuring from 0.2 to 0.7° with a seasonal trend similar to the top tilt. However, the top and mid-height tilts of the wall were out of phase: when the top leaned out the mid-height pulled back in toward the backfill. It was hypothesized that the cold temperature contracts the tieback resulting in the tieback pulling the lower portion of the wall back toward the backfill. In such a scenario, the top which is not restrained would lean out as the tieback axial force pulled the panel back. Additionally, expansion of the top layer of the backfill due to freezing would push the top portion of the wall above the tieback out. As shown in Figure 3.28, the top and mid-height tilts exhibited a strong linear relationship with temperature. Strain also appeared to have a trendline with a strong season dependence as shown in Figure 3.27. In total, bottom strain of the M-10 wall varied a total of 70 µε. When considering the strain response at the bottom of the wall relative to temperature, Figure 3.28 shows linear dependence but with a high level of variability due to some data points showing high compressive strain over a short period of a few days (as is evident in the time history plots of Figure 3.27).



Figure 3.27. Wall response time histories of the M-10 wall: bottom-level strain history, daily precipitation and wall temperature at the bottom-level



Figure 3.28. Scatter plots of M-10 wall panel response to temperature: top tilt, midheight tilt and bottom strain

## 4. DEVELOPING THE MICHIGAN EARTH RETAINING STRUCTURE ELEMENT INSPECTION MANUAL

#### 4.1. Introduction

Developing the Michigan Earth Retaining Structure Element Inspection Manual (MiERSEIM) required extensive research in the preliminary stage. Our team broke down this development into five objectives to manage the R&D progress. These steps are:

- Reviewing existing retaining wall management methods •
- Conducting field reviews of 10 MDOT retaining wall systems •
- Development of Earth Retaining Structure criteria
- Develop Structure and Appraisal (SI&A) Items for retaining walls •
- Develop inspection procedure for retaining walls (MiERSEIM)

#### 4.2. **Reviewing Existing Wall Management Methods**

Researching current organizational retaining wall inspection, monitoring, and reporting manuals was the first objective undertaken to minimize the "reinvent the wheel" aspect of this project. It was important to have an understanding of the process each government agency had when creating their policies and what they considered to be important aspects of the program. 13 agencies were researched based on manual availability and are listed below:

- National Parks The Citv of Kansas DOT • Cincinnati Service (NPS)
- Oregon DOT
- Colorado DOT ٠

•

- British Columbia
- Penn DOT
- NYC DOT
- NY State DOT •

- California DOT
- Minnesota DOT
- Missouri DOT
- Maryland DOT

The information from the agencies policies and manuals were studied, compared, and analyzed to determine which methodologies would be desired for development of the MiERSEIM. Due to resource constraints, a few agencies have Inventory and inspection (I&I) programs specifically for Mechanically Stabilized Earth (MSE) Walls only. For this project, inventory should include all walls regardless of ownership: any wall failure may impact a highway facility. From these 13 agencies, the manuals from the Colorado DOT and NPS were selected to have the ideals that align with this project.

## 4.3. Conducting Field Review of 10 MDOT Retaining Wall Systems

The information from the above manuals served as half of the process in developing the MiERSEIM. The second half involves field inspection to put the material to use. The Michigan Department of Transportation (MDOT) provided 74 different sites to investigate and from there, ten were chosen based on factors including type, size, condition, risk/consequence view, past monitoring information, and ease of access. The ten sites selected are:

- I-696 in Detroit
- US-10 in Midland
- I-75 at Brush Street in Detroit
- I-94 in Kalamazoo
- (222–MGrand St) in Allegan

- I-196 at Baldwin Connector
- I-75 Grand River Exit
- I-94 Water Street in Port Huron
- M-50 in Dundee
- US-31 along Bayfront Drive, near Petoskey

Once the sample data sets were determined, a team of personnel from the MDOT, University of Michigan, and the Mannik & Smith Group (MSG) visited each site to determine inspection criteria and methods through hands-on inspection. Every structural and non-structural component was examined to determine the feasibility of including the element in the inspection criteria. All defects were examined for their importance to failure rate, safety, and aesthetics. This data was then organized and placed into a spreadsheet to assist in creating an ERS SI&A Database. The spreadsheet grouped the wall according to the following data:

- Location
- Type
- Function
- Geometrics

- Condition of structure and components
- Nature of roadway traffic levels and surrounding development

- Risks associated with the structure's failure
- Restricted zone for permits
- Maintenance/Repair/Replacement work orders
- GPS coordinates
- Wall attachments, adjacent features, external stability conditions

- Material, physical component types (foundation/wall/backfill/post/mount)
- Detailed dimensional data (begin/end stationing, offset, distance to road,
- slope, clearance)
- Historic eligibility, architectural forms
- Detour length, traffic class
- Photographs, letters, plans

• Date built/reconstructed

## 4.4. Development of Earth Retaining Structure Criteria

Utilizing the data recorded from the inspection and the information provided by the above mentioned selected government agency manuals, the criteria to select ERS's in the inventory process was undertaken. The criteria developed for selecting ERS's are:

- Minimum height of wall (visible or total)
- Minimum length of wall
- Minimum height of retained fill
- Minimum face slope greater than 45 degrees
- Wall batter relative to roadway
- Walls within 30 to 100 feet of abutments

Minimum surface area and legal definition were not used as criterion.

# 4.5. Develop Structure and Appraisal (SI&A) Items for Retaining Walls

In order to create the MiERSEIM, a matrix was created utilizing the Colorado DOT and NPS Manuals related to Material and ERS Wall Types. These items were then combined to develop the starting point for the manual. These were expanded to include the following wall elements/characteristics assessments:

- Wall type
- Foundation
- Wall alignment
- Facing structure/treatments
- Surface coatings Attachments
- Guardrails/parapets
- Backfill material
- Backfill slope
- Drainage
- Erosion
- Vegetation
- Roadway
- Curb/Berm/SW/shoulder
- Adjacent features

Risks associates with ERS failures were also tabulated based on the failure potential, extent of the failure, threats to life/safety, link criticality/redundancy, average daily traffic impacts, and budget impacts.

These items were separated into primary and secondary element groupings and given an MDOT element number for reference. New elements were given numbers in the 900 range.

## 4.6. Develop Inspection Procedure for Retaining Walls (MiERSEIM)

After all the investigation and data layout was finalized, the MiERSEIM could be developed. The format for the manual mirrors the Michigan Bridge Element Inspection Manual (MiBEIM) for its familiarity. The manual begins with a Preface explaining the definition of the manual which is followed by an Introduction which explains the manuals purpose. A Detailed Element Description is included to define the information herein.

There are five main chapters to the Manual:

- Primary Elements which describe the structural components of the ERS
- Secondary Elements which describe the non-structural components of the ERS
- Scour Protection which describes the scour protection devices for hydraulic ERS
- Appurtenances which describe the attachments to the ERS
- Condition State Tables which describe the deficiencies for each element type.

The finished manual will serve as a tool for ERS inspectors for the recommended 5-year interval inspections recommended by NCHRP. More frequent inspections may be triggered by: Walls performing poorly, environmental settings, age, consequence of failure, natural events, and condition.

# 5. RISK ASSESSMENT OF THE TALL I-696 RETAINING WALL SYSTEM

## 5.1 Introduction

The construction of highways in dense urban areas and in challenging terrain has increased the need for retaining walls. In the United States alone, more than 160 million square feet of new wall area is constructed every year within the national highway and road network (FHWA, 2008). This results in massive inventories of retaining wall structures requiring asset management including management of their risk of failure. The transportation asset management program is integral to the Moving Ahead for Progress in the 21st Century Act (MAP-21) which requires transportation agencies to adopt risk management strategies for all highway structures inclusive of retaining walls (FHWA, 2014). While risk management methods have been extensively studied for bridge structures, comparatively less research has focused on risk management methods for retaining walls. There has been recent interest in risk assessment of geotechnical assets in recent years. For example, a recently completed report by the National Cooperative Highway Research Program (NCHRP) describes the need for geotechnical asset management (GAM) (Vessely et al., 2019). The NCHRP report lays out frameworks for highway officials to begin planning for an implementation of GAM strategies inclusive of when to adopt monitoring; however, the report's aim was to offer a qualitative framework for GAM planning and not to provide details on how to execute specific GAM plans.

Visual inspection is the first step toward developing risk-based asset management methods for retaining wall systems. Visual inspection can provide a basis for assessing the physical condition of a retaining wall system. It can also provide insight to the movement and deformation of both the retaining wall structure and the geotechnical system it supports. While these qualitative observations can lead to a deeper understanding of how the system is behaving, a quantitative risk assessment is difficult to perform using visual observations. Hence, a fundamental question is how to go from visual inspection information (VII) to an assessment of the risk of failure of a retaining wall system (Figure 5.1a). Structural monitoring can serve as a powerful augmentation

to visual inspections offering *quantitative* data on the performance to the retaining wall system (Figure



Figure 5.1. Illustrations of how to go from visual inspection to risk-based asset management: (a) current state of practice; (b) proposed approach of using monitoring data to inform quantitative reliability and risk assessments (Source: O'Conner, 2015).

5.1b). Specifically, structural monitoring systems can be installed to assess retaining wall response to loads including lateral earth pressures from permanent, temporary and cyclic loads and temperature loads. Sensors can also be installed in a retaining wall backfill to measure pressures, moisture and other factors critical to the load imposed on a retaining wall system. Monitoring data combined with visual inspection information allow load demands on the wall to be quantitatively assessed. Evidence of the performance of in-service retaining walls from monitoring data can also help identify changes in design assumptions and boundary conditions that affect the structural capacity of the system (Admassu et al., 2019). Estimates of the load demand and structural capacity derived from monitoring data and visual inspection information can then be used to calculate the reliability of the retaining wall (*i.e.*, the probability of the retaining wall system exceeding a defined performance limit state). Finally, risk is simply the product of the probability and consequences of exceeding the defined limit state.
In this chapter, a data-driven risk assessment methodology based on long-term monitoring data and visual inspection information is described. The risk assessment method developed complements the GAM risk planning framework proposed in Vessely et al. (2019) but is significantly more quantitative due to its reliance on structural monitoring data. The risk assessment method relies on measured wall responses to estimate the loads imposed (e.g., backfill earth pressures, thermal). The loads estimated from the instrumented wall response data are then applied to a structural model to assess the load effect on the wall behavior. Visual inspection information is also used to inform assumptions necessary to assess the structural capacity of the instrumented wall. Assuming normally distributed load and capacity parameters, the first order reliability method (FORM) is adopted to assess the probability of the wall response exceeding a defined limit state in the form of a reliability index,  $\beta$ . Combined with a qualitative (e.g., high, medium, low) or quantitative (e.g., cost of damage and repair) definition of the consequences of exceeding the limit state (e.g., wall failure), the risk of a wall system can be ascertained. In this chapter of the project report, the data-driven risk assessment method is developed specifically for the tall I-696 wall system using the long-term monitoring data previously presented. While the method is presented specific to one wall system, it can be generalized and easily applied to almost any other GAM application. The chapter concludes with a summary of key study findings and a description of the future work needed to advance the risk assessment framework for retaining wall systems.

## 5.2 Risk Assessment Framework

The risk assessment framework proposed herein is summarized in Figure 5.2. It relies on three primary sources of information: structural drawings/design documents, visual inspection information, and measurement data. Structural drawings are essential for detailing the structural design of the retaining wall system, construction sequencing, and backfill soil properties. This information is essential for building a mechanics-based model (*e.g.*, finite element method model) of a retaining wall and identifying the appropriate limit states of the structural materials (*e.g.*, yield strengths). Visual inspections carried out by inspectors guided by the Michigan Earth

Retaining Structure Element Inspection Manual offer detailed information on structural conditions including the physical condition of primary and secondary structural elements. Visual observations and inspector assigned condition states (or condition ratings) offer insight to the health of the structure and inform an understanding of the capacity of the wall. For example, visually identified section loss, structural corrosion, and the weakening of boundary conditions may imply a reduced system



Figure 5.2. Proposed risk assessment framework for retaining wall structures.

capacity. Visual inspection can also uncover load demands not accounted for in design (*e.g.*, excessive tilts leading to second order P- $\Delta$  effects). Different from visual inspection, monitoring systems offer data associated with wall responses to loads. This data can be used to estimate the various loads imposed on the retaining wall and be vital to modeling changes in system capacity.

The risk-based method proposed in this section is based on quantitative calculation of the reliability of the retaining wall structure. The limit state established in the reliability analysis is the design limit state such as the yield stress of reinforcement bars or crack widths in the concrete wall. Reliability is a measurement of the probability of exceeding the defined limit state function,  $G(\mathbf{X})$ , where  $\mathbf{X}$  is a vector of random variables that are the inputs to limit state. Given

the load demand (D) and structural capacity (C) of the wall system, the limit state function is the difference between capacity and demand:

$$G(\mathbf{X}) = C - D \tag{5.1}$$

Failure is equivalent to  $G(\mathbf{X}) < 0$  and can be defined more precisely as



Figure 5.3. Illustration of demand and capacity distributions in relationship to the reliability index, β.

$$P_f = \int_{G(\mathbf{X})<0} f(\mathbf{X}) d\mathbf{X}$$
(5.2)

where  $f(\mathbf{X})$  is the joint probability density function of **X**. The reliability index,  $\beta$ , is a scalar value defined as the input to the cumulative density function,  $\Phi$ , to equate it to the probability of failure:

$$P_f = \Phi(-\beta) \tag{5.3}$$

The reliability index is a widely used parameter that defines the margins of safety in design codes. For example, load resistance factor design (LRFD) codes such as those adopted by AASHTO are designed to attain a reliability index of 3 or greater in most structural components.

When the random variables defining the load effect and capacity are assumed to be normally distributed, the first order reliability method (FORM) can be conveniently adopted to calculate the reliability index via a closed form solution,

$$\beta = \frac{\mu_C - \mu_D}{\sqrt{\sigma_c^2 + \sigma_D^2}} \tag{5.4}$$

where  $\mu_c$  is the mean capacity,  $\mu_D$  is the mean demand (load effect), and  $\sigma_c$  and  $\sigma_D$  are the standard deviations on the capacity and demand distributions. Figure 5.3 provides a graphical interpretation of the reliability index where  $\beta$  is a measure of how far apart the demand and capacity means are normalized by the square root sum of their variances.

Often, the system capacity and demand are stated as stress values in critical components of the structure with capacity described by the yield limit state of the materials used. As a result, the demand side of the reliability analysis is based on estimation of the stress response (or load effect) of the critical component. In design, the load effect on the structure is based on assumed statistical models of the worst case loading that the structure may experience over its complete life cycle. In risk assessments of the structure, the load effect is what is actually measured over the life cycle of the structure. In many instances, the load effect cannot be directly measured. In these cases, analytical models describing the structure are used to convert measured responses to a distribution of the stresses pertinent to the reliability analysis. Long-term monitoring data collected from retaining walls can be used to estimate the loads imposed on the structure. Load estimates can then be used to derive a statistical model of the load effect in a defined critical structural element or structural detail.

After the reliability of the structure is established, a risk assessment can be performed. The reliability analysis offers the probability of exceeding a defined limit state such as the point where materials begin to yield. This probability of failure,  $P_f$ , can be combined with the consequence of failure,  $C_{s}$ , to estimate the risk, R:

$$R = P_f * C_{\$} \tag{5.5}$$

The consequences can be described in the form of monetary costs such as the cost of system repair or the cost of damage to other physical assets. Also included in the cost can be opportunity cost (such as the cost of road closures) and that of human life (based on an equivalent cost per each life lost). The risk assessment can consider a total number of events,  $n_E$ , of exceeding a limit state,  $P_{f,E}$ :

$$R = \sum_{i=1}^{n_E} R_{E_i} = \sum_{i=1}^{n_E} P_{f,E_i} C_{\$,i}$$
(5.6)

# 5.3 Reliability of the I-696 Tall Wall

To highlight the general process of the risk management approach proposed, the tall I-696 retaining wall system was considered in this project. The tall I-696 wall was designed as a cantilever wall. The wall was inspected by the project team in 2017 to assess its general structural condition. Based on that field investigation, the wall was considered to be in very good structural condition with some surface cracking evident on the front side of the wall. Specifically, vertical cracking was observed at the horizontal midpoint of the wall panel, especially at the wall base. Some distress was observed at the panel joints with relative displacement evident between the tall I-696 wall panel and adjacent panels. Especially noticeable was water drainage from the bottom of the wall with water coming from the joints between adjacent panels as well as from holes in the wall associated with the form work used during construction. Water drainage was evident year round, even during relatively dry summer periods. In addition to the drainage at the base of the wall, the backside of the wall also illustrated distress with significant distortion of the sidewalk at the top of the wall backside. Specifically, the team observed sidewalk panels sinking down 3" to 6" near the tall wall panel. Due to the presence of a manhole cover associated with a buried water pipeline system behind the wall, it was hypothesized that the pipeline system might be failed allowing backfill to enter the system resulting in the loss of backfill volume over time.

Continuous drainage at the wall base suggested the wall backfill is saturated with water. With the cantilever wall experiencing tension on the wall backside, flexural cracks on the wall backside may expose the steel reinforcement to water. The steel reinforcement on the wall backside is

standard steel (*i.e.*, not epoxy coated) and has a 3" cover (see Figure 3.2). The steel reinforcement was suspected of having some form of corrosion and possible section loss, thereby reducing the flexural capacity of the wall. In this study, the base of the cantilever wall at its connection to the wall system footing was considered as the critical wall section experiencing maximum flexural load (*i.e.*, maximum moment). The limit state was considered to be the yield strength of the vertical steel reinforcement on the wall backside.

As previously described, the wall was instrumented in August 2018 with tilt sensors installed at the wall mid-height and top. In addition, long-gage strain sensors were installed at the mid-height and bottom of the wall (along with temperature sensors at each strain sensor locations). The wall was monitored from the end of August 2018 to the end of October 2019 offering over a year's worth of response data for the data-driven risk assessment method. The stress in the steel reinforcement was not monitored but the tilt and strain measurements from the wall front side were used to estimate the stress in the tensile steel reinforcement under assumed states of reduced capacity due to corrosion of the steel reinforcement.

#### 5.3.1. Steel Reinforcement Capacity

The vertical steel reinforcement on the backside of the retaining wall was considered to be the critical structural element that would control the wall overall reliability. As a result, the steel reinforcement yield stress was defined as the limit state of primary interest; exceeding the yield stress would constitute "failure" of the wall system. The steel reinforcement yield strength was not specified in the structural drawings; however, structural design codes in use at the time of the wall design would prescribe Grade 60 structural steel for the buried reinforcement. While Grade 60 steel has a nominal yield strength of  $f_y = 60 \text{ ksi}$ , it must be defined probabilistically to account for variations in material properties. Bournonville et al. (2004) has probabilistically modeled the properties of Grade 60 structural reinforcement steel based on extensive experimental testing. While ASTM A615 Grade 60 reinforcement steel best follows a beta distribution, a normal distribution is deemed to be a fair representation. Grade 60 steel reinforcement is manufactured to have a minimum yield stress of 60 ksi but in reality it will have a higher mean yield stress. For example, #6 and #9 reinforcement steel bars sizes have mean yield

strengths of approximately 69 ksi (Bournonville et al. 2004). The standard deviation was estimated to be approximately between 4.3 and 5.0 ksi. Figure 5.4 presents the histograms and probability density functions (beta and normal) for ASTM A615 Grade 60 reinforcement for bar sizes #6 and #9. In this study, the yield strength of the tensile steel reinforcement was assumed to be a normal distribution with a mean at 69 ksi and standard deviation of 5 ksi.



Figure 5.4. PDF of Yield Strength of ASTM A615 Grade 60 Steel Reinforcement Bars: (a) Size #6; (b) Size #9 (Bournonville et al. 2004).

#### 5.3.2. Effect of Corrosion

Failure of retaining walls due to the corrosion of steel components has been observed in the past. For example, panels of the M-10 retaining wall system in Detroit failed due to corrosion of tieback rods linking a section of the wall system to their caisson elements (MDOT 2013). Inspection of the failed wall panels found moisture ingression through the horizontal cracks that developed at the interface between the bottom of the wall and the footing. The moisture originated from a leaky water pipeline in the backfill. The water also led to the tieback cross section being so corroded that it failed. Evidence of continuous drainage at the base of the tall I-696 wall suggested a potentially corrosive environment that has the potential to reduce the size of the steel reinforcement. It was hypothesized that the loss of steel cross section reduces the capacity (as the subsequent study revealed).



Figure 5.5. Percent section/diameter reduction for different corrosion activity ratings (Source: Andrade and Alonso, 1996)

The challenge with retaining walls is that is near impossible to assess if corrosion is occurring in buried steel on the backside of the wall without excavating the backfill. Should the backfill be excavated, a number of approaches are available for measuring the degree of corrosion in buried reinforcement including half-cell potential measurements (Elsener et al. 2003). These invasive methods (due to the need to connect to a reinforcement bar to serve as a working electrode) provide a measure of the potential of the concrete reported in voltage per copper sulphate electrode (CSE). Measured potentials can provide a guide for probability of corrosion activity in buried reinforcement. More recently, electroimpedance spectroscopy techniques have also been developed to assess the polarization resistance,  $R_p$ , of buried reinforcement based on four probes

at the surface of the reinforced concrete structural element. The polarization resistance of the reinforcement bar,  $R_p$ , can be related to the corrosion rate,  $I_{cor}$ , of the steel reinforcement through Stern's law ( $I_{cor} = B/R_p$ ) where *B* is Stern's constant (which is experimentally derived). Corrosion rate is most closely tied to corrosion activity and corresponding weight loss due to corrosion (with higher corrosion rates over time leading to greater weight loss of steel). Andrade and Alonso (1996) report on the loss of steel reinforcement bar section due to corrosion rate,  $I_{cor}$ , and time. Figure 5.5 presents a figure from Andrade and Alonso (1996) that graphically tabulates the percent section loss of buried steel reinforcement as a function of time since carbon dioxide and chloride ingress into the concrete and corrosion rate.



Figure 5.6. Overview of the reliability and risk assessment framework for the I-696 retaining wall system using tilt and strain measurements from wall front side.

Based on Figure 5.5, the age (33 years) of the I-696 wall panels suggested very little section loss under the assumption of low to moderate corrosion rates in the wall. If it is conservatively assumed that carbonation and chloride ingress occurred on the first day of construction and a moderate corrosion rate (*e.g.*, 0.5  $\mu$ A/cm<sup>2</sup>) exists, a section loss of 3% in the buried steel reinforcement was estimated using Figure 5.5. In this study, three states of buried vertical steel reinforcement were assumed given the uncertainty associated with the degree of corrosivity of the operational environment: 0, 10 and 20% section loss of the vertical steel reinforcement. The reliability of the tall I-696 wall panel was calculated for these three corrosion states.

#### 5.3.3. Strain Response Analysis

The previous chapter described the data collected from the I-696 tall wall panel and the development of a discrete-element model that can be used to estimate the lateral earth pressures on the wall and to calculate deflection curves of the wall system. To perform a reliability analysis, the model previously developed (inclusive of material properties) was adopted for the tall I-696 wall panel. Material properties including the bulk and submerged weight of soil, friction angle, concrete effective elastic modulus, and steel elastic modulus are assumed to be deterministic in the



Figure 5.7. Load effect estimation from tilt responses

model. Future analyses could be made more sophisticated by treating all model parameters probabilistically. The reliability analysis framework adopted in this project for the tall I-696 wall system is summarized in Figure 5.6. More specifically, how the strain data will be processed from the front of the wall to estimate stress in the vertical steel reinforcement in the back of the wall is presented in Figure 5.7.

The wall system monitoring system provided tilt ( $\theta$ ), strain ( $\varepsilon$ ) and surface temperature (T) measurements for more than a year of monitoring. The daily mean measurement of these wall responses were used to assess the loads imposed on the retaining wall system using the discrete element model previously described. Two specific loads were considered in the analysis: lateral

earth pressures (resulting from a surface surcharge and saturation of the soil) and thermal loads associated with the temperature of the wall. Tilt measurements are not influenced by axial expansion of the wall due to thermal loads; this allows tilt measurements to be used to isolate the flexural behavior of the wall including the flexural strain in the steel reinforcement,  $\varepsilon_{steel,flex}$ , in the critical zone at the base of the wall.

The surcharge and hydrostatic pressures were defined based on the surcharge load, q, and the height of backfill soil saturation,  $h_{sat}$ . As previously described, the wall tilt can be used to estimate these load parameters by minimizing the error between the measured wall tilt and that estimated by the discrete element model:



Figure 5.8. Top and mid-height tilt for tall I-696 wall panel assuming no corrosion for variations in surcharge load, q, and water saturation level,  $h_{sat}$ .

$$\underset{q,h_{sat}}{\operatorname{argmin}} \|\theta_{meas} - \theta_{model}\|$$
(5.7)

Equation 5.7 was solved by use of a look-up table of pre-calculated top and mid-height tilts for a range of surcharge pressures ( $q \in \{0,1.74\} psi$ ) and water saturation levels ( $h_{sat} \in \{0,282\} in$ ).

Figure 5.8 shows how tilt ( $\theta_{top}$  and  $\theta_{mid}$ ) changes under different load parameters (q and  $h_{sat}$ ) for the case of no rebar corrosion. Hydrostatic pressure clearly dominates the tilt response of the wall as compared to top surface surcharge; moreover, tilt angles rise exponentially at higher levels of saturation in the backfill soil.

Provided the optimal surcharge load, q, and height of backfill saturation,  $h_{sat}$ , estimated for each daily set of tilt measurements, the flexural moment applied to the cantilever wall, M(x), was calculated using the methodology shown in Figure 5.7. The total bending moment acting on the cantilever wall at a height of element k, M(k), was the superposition of bending moments from hydrostatic,  $M_{HS}(k)$ , surcharge,  $M_q(k)$ , and backfill,  $M_b(k)$ , pressures:

$$M(k) = M_{HS}(k) + M_a(k) + M_b(k)$$
(5.8)



Figure 5.9. Daily mean response of tilt and estimation of surcharge load, q, and water saturation level,  $h_{sat}$ , based on tall wall tilt measurements over one year of monitoring. Three states of steel section loss of the tensile reinforcement considered: 0. 10 and 20%.

Using the discrete-element model, the flexural moment on the wall allowed the flexural strain to be estimated everywhere in the wall system. For example, it was used to estimate the flexural strain in the front face of the wall where strain is actually measured:  $\varepsilon_{flex}$ . It was also used to estimate strain in the vertical steel reinforcement on the backside of the wall at the controlling

section (namely, at the wall-footing interface zone):  $\varepsilon_{steel,flex}$ . This was the load effect on the steel reinforcement bars experiencing axial tension due to the backfill earth pressures. Using the steel elastic modulus ( $E_{st} = 29x10^6 psi$ ), stress in the steel reinforcement due to flexural response can be calculated:

$$\sigma_{steel,flex} = E_{st}\varepsilon_{steel,flex} \tag{5.9}$$

It should be noted that this approach to inverse modeling to estimate the lateral earth pressure loads (q and  $h_{sat}$ ) was reliant on the discrete element model and its material and geometric assumptions.

Figure 5.9 shows the estimated time series of surcharge and water saturation levels obtained using the tilt measurements at the top and mid-height of the tall wall system. The analysis was performed for the three assumed corrosion states: 0, 10 and 20% section loss in the



Figure 5.10. Daily maximum bending moment variation at the base of the wall for three states of assumed corrosion in the tensile steel reinforcement.



Figure 5.11. Vertical steel reinforcement strain at the base of the wall backside due to flexural moment.

steel reinforcement on the backside of the wall. The analysis was performed for each corrosion state because changes in the area of steel reinforcement in the wall cross section altered the location of the section neutral axis resulting in three different models for estimating the lateral earth pressures behind the wall. Figure 5.10 shows the calculated total bending moment at the wall base for the three corrosion states. Figure 5.11 shows the calculated flexural strain,  $\varepsilon_{steel,flex}$ , in the backside steel reinforcement due to the estimated moment for the three corrosion states.

The strain measurements collected from the front face of the tall wall were influenced by both flexural moment (*i.e.*, strain associated with flexural bending) and temperature (*i.e.*, axial strain). An illustration of this is shown in Figure 5.12. During the summer when the wall is warm, the

wall experiences expansion which is axial tension; during the winter it contracts creating axial compression. With estimated flexural strains estimated by the model, the strain response of the wall due to the thermal load environment can be obtained. Unlike tilt which is an absolute measurement, the strain measurements at the bottom of the wall represent change in strain since the start of measurement:  $\Delta \varepsilon_{bot}$ . Given the estimates of  $\varepsilon_{flex}$  at the point of the strain measurement, the change in strain due to flexure,  $\Delta \varepsilon_{flex}$ , since the start of the measurement of strain was calculated. Hence, the change in strain associated with temperature,  $\Delta \varepsilon_{temp}$ , can be found by subtracting the estimate of  $\Delta \varepsilon_{flex}$  at the bottom of the wall (at the point of strain measurement) from the measurement itself,  $\Delta \varepsilon_{bot}$ :

$$\Delta \varepsilon_{temp} = \Delta \varepsilon_{bot} - \Delta \varepsilon_{flex} \tag{5.10}$$

The change in strain due to temperature was plotted as a function of wall surface temperature shown in Figure 5.13. The plot is for changes between sequential daily measurements. The thermal expansion coefficient,  $\alpha$ , obtained was 1.89  $\mu\epsilon/^{\circ}F$  which was close coefficients documented in the literature for reinforced concrete (which are 3 to 6  $\mu\epsilon/^{\circ}F$ ) (Berwanger and Sarkar 1976).

Axial strain in the wall due to temperature was assumed uniform across the wall section. Hence, the  $\Delta \varepsilon_{temp}$  at the front of the wall was considered as the same as the thermal induced strain in the steel reinforcement in the wall backside. The strain in the steel reinforcement on the wall backside was equal to the residual strain at the start of measurement,  $\varepsilon_{steel,temp,o}$ , plus the change in strain due to temperature change,  $\Delta T$ , relative to the first day of measurement (when T =80°F). This was used to calculated thermal stress in the reinforcement:

$$\sigma_{steel,temp} = E_{st}\varepsilon_{steel,temp} = E_{st}(\varepsilon_{steel,temp,o} + \alpha\Delta T)$$
(5.11)

The residual strain in the reinforcement due to thermal behavior in the past (*i.e.*, prior to monitoring) is unknown. The temperature of the concrete at the time of casting was assumed to be approximately 50 °F; compared to the start of monitoring when the wall temperature was 80 °F, the thermal coefficient estimated for the tall I-696 wall system was used to calculate the thermal residual strain  $\varepsilon_{steel,temp,o}$  which was 56 µ $\varepsilon$  in tension.



Figure 5.12. Strain profiles on the I-696 tall and short walls at different seasons of the year: (a) During Spring and Summer seasons where the temperature gets warmer; (b) During Fall and Winter seasons where the temperature gets colder.



Figure 5.13. Scatter plot showing wall strain response to thermal load versus change in temperature is linear.

#### 5.3.4. Reliability Analysis

Using the methodology previously described, the daily mean wall response was used to estimate the flexural and thermal strain in the vertical steel reinforcement bars on the backside of the tall I-



Figure 5.14. Histograms of estimated vertical reinforcement steel tensile stress compared to probabilistic model of steel reinforcement yield stress.

696 wall. The total strain was then used to estimate the total tensile stress in the reinforcement. The analysis was performed for each day of measurement using the daily mean tilt and strain responses. The total stress time series for the three corrosion states were used to create histograms of the stress in the vertical steel reinforcement at the wall base on the backside of the wall. The histograms for the three corrosion states are plotted along with the probabilistic model of the steel reinforcement yield strength as shown in Figure 5.14.

The histograms of the total steel stress were far below that of the probabilistic model of the steel yield stress. The histograms of the estimated steel stress for the three corrosion states were treated as if they were normally distributed with their mean and standard deviations calculated. The closed form expression for the reliability index (Equation 5.4) was then used to estimate the reliability of the wall system. The mean and standard deviations of estimated stress in the steel reinforcement at the base of the wall along with the mean and standard deviation of the yield strength of the steel were used to estimate the reliability index,  $\beta$ . Assuming no loss of rebar

section due to corrosion, the reliability index was calculated to be 8.5; assuming 10% and 20% section loss, the reliability index reduced to 7.9 and 7.4, respectively. These reliability index values provide a robust margin of safety for the wall, suggesting the wall has a low probability of failure. It should also be noted the reliability indices obtained from monitoring data are far above those intended during the design process ( $\beta \approx 3$ ).

(1)	(2)	(3)	(4)	(5)	(6)
Code	Condition Description	% Section Loss	Condition	Qualitative Note	Cost
9	Excellent	0 %	$\beta \geq \beta_9$	No problems noted	C9
8	Very Good	+	$\beta_9 > \beta \ge \beta_8$	No problems noted	C <sub>8</sub>
7	Good	Superficial	$\beta_8 > \beta \ge \beta_7$	Some minor problems	C <sub>7</sub>
6	Satisfactory	Damage	$\beta_7 > \beta \ge \beta_6$	Some minor deteriorations	C <sub>6</sub>
5	Fair	(0%, 5%]	$\beta_6 > \beta \ge \beta_5$	Minor sec. loss, cracking, scour	C5
4	Poor	(5%, 10%]	$\beta_5 > \beta \ge \beta_4$	Advanced sec. loss, deterioration	C <sub>4</sub>
3	Serious	(10%, 20%]	$\beta_4 > \beta \ge \beta_3$	Loss of sec., seriously affected elements	C <sub>3</sub>
2	Critical	(20%, 50%]	$\beta_3 > \beta \ge \beta_2$	Adv. deterioration/s of elements	C <sub>2</sub>
1	"Imminent" Failure	(50%, 70%]	$\beta_2 > \beta \ge \beta_1$	Major deterioration/s that need closing of road	C <sub>1</sub>
0	Failed	(70%, 100%]	$\beta_1 \geq \beta$	Out of service	C <sub>0</sub>

Table 5.1. Updated FHWA condition ratings to include the reliability index (β<sub>i</sub>) thresholds that bound corresponding condition rating codes.

# 5.4 Risk Assessment

Risk assessment considers the reliability index (providing the probability of exceeding a defined limit state) and the consequences associated with exceeding the limit state with risk is simply the probability times the consequence of exceeding the limit state. Prior work by Flanigan et al. (2019) has established the concept of "lower" limit states that can be defined below the probability of failure and that correspond one-to-one to condition ratings. In other words, there is an equivalency between condition rating (given by an inspector) and the reliability index estimated from monitoring data. The work of Flanigan et al. (2019) has revealed the key for corrosion-based deterioration is attributing the percent section loss to reliability index thresholds

as shown in Table 5.1. Table 5.1 utilizes the 9-point condition rating scale with its associated condition description (column 2) and qualitative note (column 5) (FHWA, 1995). However, added is a qualification of the percent section loss (column 3) tolerable under each condition rating. Section loss of structural elements is already considered by visual inspectors when performing condition ratings for bridge elements. Table 5.1 provides a proposed set of percentage section loss of steel reinforcement for the tall I-696 wall that is consistent with existing qualitative description of section loss tolerated



Figure 5.15. Reliability index values for varying levels of section loss in the tall I-696 retaining wall panel backside vertical steel reinforcement.

in existing condition ratings. With some qualitative description of section loss, the reliability index thresholds can be considered (column 4). For example, if a condition rating if 6 corresponds to uncorroded steel reinforcement, under assumption of a fully saturated back fill with the AASHTO code specified surface surcharge (1.74 psi),  $\beta_6$  would then be 8.51 for the tall I-696 wall panel. For condition rating 3, assuming 20% section loss of the reinforcement, this would establish a reliability threshold of  $\beta_3$  of 7.2. Figure 5.15 provides the reduction in reliability index for each percent section loss of vertical steel reinforcement. This can be useful for rationally establishing the beta thresholds and the tolerable degree of section loss that can be considered. Once a condition rating is assigned, the retaining wall manager must make a decision on what to do; namely, to repair or not. There are costs to the actions taken triggered by the condition rating (or estimate reliability index,  $\beta$ ). These costs can be used along with the reliability indices (which now provide a probability of exceeding a condition rating reliability index threshold) to calculate risks associated with post-inspection decision making.

An alternative approach in the short term, is to simply bin the reliability index into three reliability categories: high, medium, low with low meaning low reliability or a higher probability of failure and high meaning high reliability or a low probability of failure. For example, two thresholds can be established by the asset manager for a given retaining wall type:  $\beta_{low}$  and  $\beta_{high}$ . For example,  $\beta_{low}$  could be selected to link to condition rating 1 and 0 while  $\beta_{high}$  could be selected to link to condition rating 6 and higher.

Table 5.2. Red-yellow-green (R-Y-G) risk categories mapping reliability index (β) values with consequence of failure event

<b>Reliability Index</b>	Consequences			
	Low	Medium	High	
$\beta > 3$				
$2 < \beta < 3$				
$\beta < 2$				

Table 5.3. Consequence Categories for Highway Retaining Wall Structures (Theme	of
Table Adopted from: (M. Imam et al, 2012))	

Consequence Categories	<b>Consequence Examples</b>			
	<ul> <li>Fatalities and injuries</li> </ul>			
Human	<ul> <li>Psychological damage</li> </ul>			
	<ul> <li>Inconvenience (affected utility lines, road closures, stranded state, etc.)</li> </ul>			
	<ul> <li>Replacement/repair costs and loss of functionality/downtime</li> </ul>			
	• Traffic delay, traffic re-routing and traffic management costs			
Economic	• Clean up costs (backfill soil falling on ramp or freeway lanes, etc.)			
	<ul> <li>Regional economic effects and loss of production/ business</li> </ul>			
	<ul> <li>Investigations/compensations and infrastructure interdependency costs</li> </ul>			
Environmental	$\circ$ CO <sub>2</sub> emissions and energy use			
Environmental	<ul> <li>Pollutant releases (from sewerage lines)</li> </ul>			
	• Loss of reputation (of the transportation agency in jurisdiction)			
Social	• Erosion of public confidence			
	• Undue changes in professional practice			

Similarly, consequences can be defined as high, medium and low allowing Table 5.2 to guide asset management decisions. In this approach, the reliability index is purely quantitative but the

consequences would be described more qualitatively. These consequences would be associated with the cost of failure of the retaining wall asset. High levels of judgement would need to be given to define specific costs, especially for some of the less tangible consequences such as social consequences (*e.g.*, inconvenience, loss of reputation, erosion of public confidence, psychological damage) and loss of human life which is invaluable. Table 5.3 provides an overview of the different consequence types one can consider. All of these consequences could be weighed when assigning for each asset being managed to a low, medium and high consequence category. In the case of the tall I-696 wall monitored in this study, its structural function of supporting a two-lane highway (Eleven Mile) above an eight lane freeway (I-696) in a high volume traffic region (i.e., metropolitan Detroit), this wall would mean high consequences if it failed including closure of Eleven Mile, closure of the I-696 ramp, and potential partial or full closure of eastbound I-696. Given the high reliability ( $\beta > 7$ ) but high consequences, Table 5.2 would classify this asset as "yellow" indicating more vigilant observation by visual inspection.

## 6. CONCLUSIONS – RECOMMENDATIONS

### 6.1 Conclusions

The implementation of a long-term, wireless and unattended monitoring solution for cantilever and caisson supported RC retaining walls (located in the Detroit metro region) was presented. The monitoring solution comprised of tilt-sensors, strain sensors and thermistors to measure wall panel tilt, compressive strains and wall temperatures, respectively. To detect very small responses in the walls, the sensors were selected to have extremely low noise floors offering high measurement resolutions (0.01° in tilt angle; 2µε in strain and 1 °F in temperature). The monitoring systems installed on the I-696 and M-10 panels performed well over a one-year monitoring period.

11 wireless sensor units were installed on the retaining wall systems along the I-696 (two wall panels) and M-10 (one wall panel) freeway corridors that collected a total of 16 measurements from the three wall panels. The tilt, strain and temperature of the wall panels were all collected to observe both the wall panel responses but also their thermal state. Data was collected for over one year allowing seasonal variations in the wall behavior to be observed. The long-term nature of the monitoring system deployments allowed the robustness of the sensors to be studied. The sensors installed on the tall I-696 and M-10 wall panels performed relatively well surviving the duration of the monitoring period; the long-term performance of the wireless sensors on the short I-696 wall panel were less robust. Issues observed during periods when the sensor nodes were serviced revealed issues associated with moisture penetration to the node enclosures and some power issues. None the less, the data collected from the tall I-696 panel and the M-10 panel were sufficient to perform a quantitative risk assessment of the walls.

The performance of the instrumented retaining wall systems was assessed using the data collected. Specifically, the wall panels exhibited strong dependence on environmental parameters, most notably temperature. In general, the cantilever wall system along I-696 exhibited higher drifts on its top sections as compared to the mid-height. The tall I-696 wall panel tilted as much as 3.5° while the mid-height maximum tilt was 1.45°. The top of the I-696 wall system was a parapet wall whose flexural rigidity was less than the lower wall portions and exhibited sensitivity to moisture

in the form of precipitation. The lower portion of the wall system was less variable day-to-day and had a behavior much better correlated to temperature, most especially mid-height tilt and bottom strain. The wall experiences maximum flexural response during the cold winter with maximum mid-height tilt (1.4°) and compressive strain. The differences in behavior at the top and the bottom portions of the cantilever wall are significant because they inform a monitoring strategy for future wall studies. Specifically, should tilt be measured on an irregular basis (for example, manually) or regularly over a short period (say a few weeks or months), the maximum wall response may not be observed. The short-term variations of the wall tilt can be significant (such as the top of the I-696 wall); a more accurate view of wall behavior require at least one year of monitoring to see the full range of seasonal variations.

The short I-696 wall panel behaved in a manner comparable to the tall I-696 wall panel but exhibited more variation in tilt at its mid-height. This also goes to show the variations wall panels can exhibit under nearly identical environmental and backfill soil conditions. Hence, behavior at one panel may not necessarily serve as a fully representative sample of the adjacent panels along a highway corridor. Instrumentation of retaining wall panels will always be sparse given current cost of purchasing and installing instrumentation.

The M-10 wall system has a history of failing tie-backs on the north side of Schaefer Highway. In this study, a representative wall panel on the south side of the Schaefer Highway was instrumented. The wall panel had mild variation in its tilt measurements with top tilt varying from 1.1 to 2.0° and mid-height tilt varying from 0.2 to 0.7°. The tilt was correlated to temperature with cold temperatures pulling the lower portion of the wall towards the backfill and the upper portion thrusted outward away from the backfill. It was hypothesize this was a result of contraction in the soil pulling the tie-back caisson back and the lower portion of the wall with it. The active earth pressures above the tie-back would then push the unrestrained upper portion outward. This suggests the tie-rod is engaged and working as expected. Similar to the cantilever wall panels of I-696, the maximum flexural demand on the lower portion of the wall and maximum tension in the tie-rod is during the cold winter.

After all the investigation and data layout was finalized, the MiERSEIM could be developed. The format for the manual mirrors the Michigan Bridge Element Inspection Manual (MiBEIM) for it

familiarity. The manual begins with a Preface explaining the definition of the manual which is followed by an Introduction which explains the manuals purpose. A Detailed Element Description is included to define the information herein.

There are five main chapters to the Manual:

- Primary Elements which describe the structural components of the ERS
- Secondary Elements which describe the non-structural components of the ERS
- Scour Protection which describes the scour protection devices for hydraulic ERS
- Appurtenances which describe the attachments to the ERS
- Condition State Tables which describe the deficiencies for each element type.

The finished manual will serve as a tool for ERS inspectors for the recommended 5-year interval inspections recommended by NCHRP. More frequent inspections may be triggered by: Walls performing poorly, environmental settings, age, consequence of failure, natural events, and condition.

Risk assessment methods have been widely studied for bridges, but comparatively less research has been focused on developing risk assessment tools for retaining walls. Most recently, some novel work has been completed in proposing risk assessment frameworks for geotechnical assets (*i.e.*, geotechnical asset management (GAM) planning). As Vessely et al. (2019) has concluded, GAM planning must include prescription of structural monitoring where appropriate. An aim of this project was the development of a risk assessment framework that utilized structural monitoring data to define the reliability of the retaining wall system. Combined with the consequence of exceed the limit state obtained from the reliability analysis, the risk of the asset can be determined. Quantifying risk in this way established for the first time, a clear pathway of using monitoring data within a broader GAM strategy. An especially valuable feature of the proposed risk assessment procedures was the explicit inclusion of visual inspection information to define structural conditions critical to the evaluation of structural capacity.

The risk assessment method was applied to the tall I-696 wall panel to illustrate its use. The longterm monitoring data from the wall panel, coupled with the discrete element model previously developed, were used to estimate wall loads. The thermal load was measured but the backfill earth pressure needed to be estimated by finding an optimal surcharge load, q, and level of soil saturation,  $h_{sat}$ , that predicted tilt measurements close to those measured. Provided the visually observed drainage from the lower portions of the wall, it was hypothesized that the uncoated vertical steel reinforcement bars may have experienced corrosion. Without a direct measure of corrosion state, three corrosion states were considered for the wall: 0%, 10%, and 20% section loss of the steel reinforcement. Tilt measurements provided a direct means of estimating the strain in the wall due to flexural bending. The strain measured on the front face of the wall was then used to isolate the thermal strain in the wall associated with the measured wall temperature. The thermal axial loading and flexural moment from the backfill were then used to estimate the load effect in the vertical steel reinforcement on the wall backside. Assuming normal distributions, FORM was used to estimate the probability of the wall exceeding the yield strength of the steel reinforcement under the different corrosion states (0%, 10% and 20% steel reinforcement section loss). Assuming no corrosion to the rebar, the reliability index was calculated as 8.5. Assuming 10% and 20% section loss, the reliability index reduced to 7.9 and 7.4, respectively. These reliability indices indicated an extremely low probability of exceeding the steel reinforcement yield strength. From a risk assessment perspective, the consequences of failure would be extremely high due to the fact that the I-696 wall system supports a very active two lane service road (Eleven Mile) at its top; in addition, failure of the wall would likely require closure of the eastbound I-696 Exit 11 (Evergreen Road). None the less, the low probability of failure implies the wall is a low risk asset.

### 6.2 Recommendations for Future Work

The research team collected images using a camera and then developed a 3D point cloud model through the Structure from Motion technique for one of the instrumented sections. This was outside the original scope of this project, however this example help demonstrates the possibilities for incorporating this type of data collection to the inspection procedure. Images can be collected using vehicle mounted cameras.

Another opportunity for future work, is to incorporate the data collected from monitored and/or instrumented retaining wall sections with other type of data, such as land use, local geology, lifelines, population density using a GIS based platform.

This study was intended to conceptually define a risk assessment framework for asset management of retaining walls using monitoring data. It could and should be further refined prior to adoption by transportation officials. Specifically, a more robust probabilistic assessment should be performed using the structural model including statistical modeling of all system properties (not just the yield strength of steel). For example, soil properties (e.g., dry and submerged weights, internal friction angles), concrete effective elastic modulus, and residual strain in the reinforcement require statistical models to define within the FORM analysis. Also, other critical limit states should be considered including crack width on the wall backside and overall deflections (to account for second order effects). Finally, future work should provide a more rational mapping between the condition rating and the loss of section that then would define the lower limit states ( $\beta_1$  through  $\beta_9$ ).

### 7. RECOMMENDED IMPLEMENTATION PLAN

## 7.1 List of products expected from research

The products of this research include a final report summarizing the research results from all the research tasks and presenting the development of a sensing strategy that can be used by structural inspectors to assess the coupled performance of the wall structure and the geotechnical system it supports, as well as a reliability framework using first-order reliability methods (FORM) to assess the reliability factor ( $\beta$ ) for wall components. A new inspection manual was developed to reflect the instrumentation strategies and risk analyses. Training materials will also be provided for the training session targeting MDOT engineers, but also consultants working with MDOT. The final inspection manual has been provided in Appendix C and described in Chapter 4.

#### 7.2 Audience for research results

The main audience for the research results includes MDOT's bridge designers, structural and geotechnical engineers and MDOT consultants. The extended audience can be other state DOTs and other government agencies involved in inspection and asset management for retaining walls. The deliverables from this project will significantly improve the ability to quantitatively, and within a risk-based framework, assess the performance and condition of retaining walls and reduce costs associated with the "worst-first" approach.

### 7.3 Activities for successful implementation

Successful implementation of the recommended procedures and new inspection manual was initiated through a training session by the research team for MDOT personnel who will be involved in relevant projects.

## 7.4 Criteria for judging the progress and consequences of implementation

The judging of progress was achieved by close supervision of the graduate student researcher, Mr. Kidus Admassu and post doctoral researcher involved in this research project, Dr. Athena Grizi, the field instruments installation, and the data analyses. Close collaboration with the subcontractor, Mannik & Smith allowed for consistent progress with the inspection manual. Regular meetings of the U-M based research team were made during the project to monitor progress and supervise the literature review and analyses. Additional meetings were also scheduled with Mannik & Smith senior engineers who worked on putting together the new inspection manual for retaining walls. An additional quality assurance and control of the research investigation will be implemented during submission of interim and final research publications in peer-reviewed scientific journals and conferences. This review process typically involves 2-3 independent reviews by researchers knowledgeable on the research topic.

# 7.5 Costs of implementation

The primary cost of implementation was the preparation of the final report and the preparation of the training session/s as needed for successful implementation of the final product of this research. This cost was included in the proposed budget.

# 8. REFERENCES

- AASHTO (2010a). "The Manual for Bridge Evaluation", First Edition with 2010 Interim Revisions, American Association of State Highway and Transportation Officials, Washington, DC.
- AASHTO (2010b). "LRFD Bridge Design Specifications", 5<sup>th</sup> Edition, American Association of State Highway and Transportation Officials, Washington D.C.
- AASHTO (2011). "Transportation Asset Management guide: a focus on implementation", American Association of State Highway and Transportation Officials, Washington, D.C.
- Admassu, K. A., Lynch, J. P., Athanasopoulos-Zekkos, A., and Zekkos, D., "Long-term Wireless Monitoring Solution for the Risk Management of Highway Retaining Walls," Proc. SPIE 10971, Nondestructive Characterization and Monitoring of Advanced Materials, Aerospace, Civil Infrastructure, and Transportation XIII, 1097103, Denver, CO, (2019).
- AECOM, "M-10 Feasibility Report: Retaining Wall Stabilization Alternatives", AECOM Corporation, Grand Rapids, MI, (2016).
- Alaska Department of Transportation and Public Facilities (2013). "Retaining Wall Inventory Procedures Manual", Draft.
- Alzamora, D. E. and Anderson, S. A. (2009). "Review of Mechanically Stabilized Earth Wall Performance Issues", In 88<sup>th</sup> Annual Meeting of the Transportation Research Board, CD-ROM, 16 pp.
- American Concrete Institute (ACI), Prediction of Creep, Shrinkage, and Temperature Effects in Concrete Structures, ACI Committee 209, Farmington Hills, MI (1997).
- AMSE (2006). "Reduced Zinc Loss Rate for Design of MSE Structures", Association form Metallically Stabilized Earth, White Paper, 61 pp.
- Analog Devices, "Using an Accelerometer for Inclination Sensing", Application Note, Norwood, MA, (2010).

- Andrade C. & C. Alonso, "Corrosion rate monitoring in the laboratory and on-site", Construction and Building Materials, Vol. 10, NO. 5, Elsevier Science Ltd, U.K., (1996).
- B. Elsener, C. Andrade, J. Gulikers, R. Polder, M. Raupach, "Half-cell Potential Measurements-Potential Mapping on Reinforced Concrete Structures." Materials and Structures, RILEM, 36(4):461-41 (2003)
- Bay, J. A., Anderson, L. R., Gerber, T. M. and Maw, R. B. (2010). "An Inspection, Assessment, and Database of UDOT MSE Walls", No. UT-09-21, Utah Department of Transportation.
- Berwanger C. and F. Sarkar, "Thermal Expansion of Concrete and Reinforced Concrete." ACI Journal, American Concrete Institute, 73(11): 618-621 (1976).
- Blythe, D. (1999). "Advanced Graphics Programming Techniques Using OpenGL", Siggraph.
- Bosch Sensortec, "BNO055: Intelligent 9-Axis Absolute Orientation Sensor", Sensor Datasheet -BST-BNO055-DS000-14, Bosch Sensortec GmbH, Reutlingen, Germany, (2016).
- Bournonville, M., J. Dahnke, D. Darwin, Statistical Analysis of the Mechanical Properties and Weight of Reinforcing Bars, Structural Engineering and Materials Laboratory, Report 04-1, University of Kansas, Lawrence, KS (2004).
- Bridge Diagnostics Inc. (BDI), "5.2 BDI Strain Transducer ST350", STS-WiFi Operations Manual, Boulder, CO, (2006).
- British Columbia Ministry of Transportation and Infrastructure (2013). "Retaining Wall Inspection Field Book", Unpublished, 24 pp.
- Brutus, O. and Tauber, G. (2009). "Guide to Asset Management of Earth Retaining Structures", prepared as part of NCHRP Project 20-07, Task 259, Transportation Research Board of the National Academies, 120 pp.
- Campbell Scientific Inc., "Time-Domain Reflectometry System Nondestructive Measurements", TDR100 Reflectometer, SDM8X50 Coax Mulitplexer, CS600-Series Soil TDR Probes, Logan, UT, (2016).

- Cerminaro, D. J. (2014). "Implementation of photogrammetry to improve proactive assessment of retaining walls along transportation corridors", MSc Thesis, Michigan Technological University, 69 pp.
- City of Cincinnati (2015). "Retaining Wall Condition Report", Department of Transportation and Engineering, 45 pp.
- City of Seattle (2014). "Retaining Walls Condition Assessment", Unpublished, 4 pp.
- Coduto, C.P. (2001). "Foundation Design: principles and practices", 2<sup>nd</sup> edition, Pearson
- Corbett, D.A., H. H. Coyle, R. E. Bartoskewitz and L. J. Milberger, "Evaluation of Pressure Cells Used for Field Measurements of Lateral Earth Pressures on Retaining Walls", Research Report No. 169-1, Texas A & M University, College Station, TX, (1971).
- CTC and Associates (2013). "Asset Management for Retaining Walls", Minnesota Department of Transportation, Report #TRS1305, 42 pp.
- Das, B.M (2011). "Fundamentals of Geotechnical Engineering", Cengage Learning
- Delphia, J. G. (2012). "MSE Wall Case Studies", Power Point presentation, Texas Department of Transportation.
- DeMarco, M., Keough, D. and Lewis, S. (2010). "Retaining Wall Inventory and Assessment Program (WIP) National Park Service Procedures Manual", No. FHWA-CFL/TD-10-003, Federal Highway Administration, 188 pp.
- Elias, V., Christopher, B. R. and Berg, R. R. (2001). "Mechanically Stabilized Earth Walls and Reinforced Soil Slopes Design and Construction Guidelines", Report No. FHWA-NHI-00-043, Federal Highway Administration, Washington, DC, 420 pp.
- Elias, V., Fishman, K. L., Christopher, B. R. and Berg, R. R. (2009). "Corrosion/Degradation of Soil Reinforcements for Mechanically Stabilized Earth Walls and Reinforced Soil Slopes", Report FHWA-NHI-09-087, Federal Highway Administration, Washington D.C., 142 pp.
- Exosite Website, "We were doing this before "IoT" was even a thing", Minneapolis, MN, 29 September 2018, <<u>https://exosite.com/about/</u>>

- Federal Highway Administration (FHWA), "2014 Transportation Asset Management Peer Exchange – Preparing for MAP-21 Implementation", Publication No. FHWA–HIF–14–013, U.S. Department of Transportation, Washington, DC (2014).
- Federal Highway Administration (FHWA), "Design and Construction of Mechanically Stabilized Earth Walls and Reinforced Soil Slopes – Volume II", Publication No. FHWA-NHI-10-025, Washington, DC, (2009).
- Federal Highway Administration (FHWA), "Recording and Coding Guide for the Structure Inventory and Appraisal of the Nation's Bridges", Report No. FHWA-PD-96-001, Washington, DC (1995).
- Federal Highway Administration (FHWA), Design and Construction of Mechanically Stabilized Earth Walls and Reinforced Soil Slopes: Volume II. Publication No. FHWA NHI-10-025, Federal Highway Administration, Washington DC (2009).
- Fishman, K. L. and Withiam, J. L. (2011). "LRFD Metal Loss and Service-Life Strength Reduction Factors for Metal-Reinforced Systems", NCHRP 675, Transportation Research Board, 116 pp.
- Fishman, K. L., Gladstone, R. A. and Wheeler Jr, J. J. (2009). "Condition Assessment of Earth Reinforcements for Asset Management". In 60th Highway Geology Symposium, New York State Thruway Authority, New York State Department of Transportation, pp. 348-365.
- Flanigan, K., and J. P. Lynch, "Community Engagement Using Urban Sensing: Technology Development and Deployment Studies," 25<sup>th</sup> International Workshop on Intelligent Computing in Engineering, Lausanne, Switzerland, 2018.
- Flanigan, K., J. P. Lynch, and M. Ettouney, "Quantitatively Linking Long-Term Monitoring Data to Condition Ratings Through a Reliability-based Framework," Structural Safety, submitted and under review (2019).
- Fleckenstein, L. J., Rister, B. W. and Allen, D. L. (1998). "Evaluation of Wall Structures in Kentucky", No. KTC-97-10, Kentucky Transportation Cabinet, 207 pp.
- Gerber, T. M. (2012). "Assessing the Long-Term Performance of Mechanically Stabilized Earth Walls", NCHRP 437, Transportation Research Board, 211 pp.

- Gladstone, R., Anderson, P., Fishman, K. and Withiam, J. (2006). "Durability of galvanized soil reinforcement: More than 30 years of experience with mechanically stabilized earth", Transportation Research Record: Journal of the Transportation Research Board, (1975), pp.49-59.
- Gridpoint Solutions Ltd. (2005). "Wellington Cut Retaining Wall Monitoring".
- Hearn, G. (2003). "Feasibility of a Management System for Retaining Walls and Sound Barriers", No. CDOT-DTD-R-2003-8, Colorado Department of Transportation, Research Branch, 106 pp.
- Hugenschmidt, J. and Kalogeropoulos, A. (2009). "The inspection of retaining walls using GPR", Journal of Applied Geophysics, Vol. 67, No.4, pp. 335-344.
- Huston, D. R., Pelczarski, N. and Esser, B. (2001). "Inspection of bridge columns and retaining walls with electromagnetic waves", In Proc. SPIE, Smart Structures and Materials 2001: Smart Systems for Bridges, Structures, and Highways Vol. 4330.
- Imam, M., & Marios K. Chryssanthopoulos, "Causes and Consequences of Metallic Bridge Failures", Structural Engineering International, Taylor & Francis Group, London, UK (2012).
- Jensen, W. (2009). "Inspector's Manual for Mechanically Stabilized Earth Walls", No. SPR-P1(09) P320, Nebraska Department of Roads, 36 pp.
- Jon W. Koloski, Sigmund D. Schwarz, and Donald W. Tubbs, Geotechnical Properties of Geologic Materials, Engineering Geology in Washington - Volume I, Washington Division of Geology and Earth Resources Bulletin 78, Olympia, WA, (1989).
- Kemeny, J. and Turner, K. (2008). "Ground-based LiDAR: Rock Slope Mapping and Assessment", Federal Highway Administration Report, No. FHWA-CFL/TD-08-006, 114 pp.
- Kim, Y. R., Hummer, J. E., Gabr, M., Johnston, D., Underwood, B. S., Findley, D. J. and Cunningham, C. M. (2009). "Asset Management Inventory and Data Collection", No. FHWA/NC/2008-15, North Carolina Department of Transportation, 286 pp.
- Koerner R., and G. R. Koerner, "Recommended Layout of Instrumentation to Monitor Potential Movement of MSE Walls, Berms and Slopes", Geosynthetic Institute, Folsom, PA, (2011).

- L. W. Schulze, H. M. Coyle and R. E. Bartoskewitz, "Field Measurements of Earth Pressure on a Cantilever Retaining Wall", Publication No. FHWA/TX-81/7+236-1, Texas A & M University, College Station, TX, (1981).
- Lostumbo, J. M. and Artieres, O. (2011). "Geosynthetic Enabled with Fiber Optic Sensors for MSE Bridge Abutment Supporting Shallow Bridge Foundation", In Geo-Frontiers 2011: Advances in Geotechnical Engineering, ASCE, Dallas, TX, March 13-16, 2011, pp. 3497-3504.
   MAP-21 (2012). Moving Ahead for Progress in the 21<sup>st</sup> Century Act, Public Law 112-141.
- Michigan Department of Transportation (MDOT), "Preliminary Findings of Retaining Wall Investigations", Office Memorandum, Michigan Department of Transportation, Lansing, MI, (2013).
- Michigan Department of Transportation (MDOT), 63102 I-696 Retaining Wall Details (Plan and Profile), Working Drawings, Michigan Department of Transportation (MDOT), Lansing, MI (1986).
- Michigan Department of Transportation (MDOT), 82112 M-10 Special Wall Details (Plan and Profile), Working Drawings, Michigan Department of Transportation (MDOT), Lansing, MI (1960).
- Molla, A. (2014). "City of Seattle Retaining Walls Condition Assessment", Unpublished, 4 pp.
- Narsavage, P. (2006). "MSE Walls Problems and Solutions", In Ohio DOT Geotechnical Workshop.
- New York City Department of Buildings (2007). "Castle Village Retaining Wall Collapse", Board of Inquiry Report, 41 pp.
- O'Connor, S. M., "Wireless Monitoring Systems for Long-Term Reliability Assessment of Bridge Structures based on Compressed Sensing and Data-Driven Data Interrogation Methods", Ph.D. Dissertation, University of Michigan – Ann Arbor, Ann Arbor, MI (2015).
- ODOT Specifications (2007). Oregon Department of Transportation, Section 00596.11.
- Omega Engineering, "KFH Series Pre-Wired Strain Gages", Datasheet, Omega Engineering, Norwalk, CT, (2018).

- Oskouie, P., Becerik-Gerber, B. and Soibelman, L. (2016). "Automated measurement of highway retaining wall displacements using terrestrial laser scanners", Automation in Construction, Vol. 65, pp.86-101.
- Pennsylvania Department of Transportation (2010). "Bridge Safety Inspection Manual", 2<sup>nd</sup> Edition, 482 pp.
- Potential Movement of MSE Walls, Berms and Slopes", GRI White Paper #19, Geosynthetic Institute, Folsom, PA.
- Priznar, N., Coxon, V. and Kemeny, J. (2010). "Using LiDAR laser scanning for geotechnical characterization in rock", In 61<sup>st</sup> Highway Geology Symposium, Oklahoma City, Oklahoma August 23-26, 2010.
- Raeburn, C. L., Monkul, M. M. and Pyles, M. R. (2008). "Evaluation of Corrosion of Metallic Reinforcements and Connections in MSE Retaining Walls", No. FHWA-OR-RD-08-10, Oregon Department of Transportation and Federal Highway Administration, 41 pp.
- Rasdorf, W., Gabr, M. A., Butler, C. J., Findley, D. J. and Bert, S. A. (2015). "Retaining Wall Inventory and Assessment System", No. FHWA/NC/2014-10, North Carolina Department of Transportation, 273 pp.
- RS Instruments, "Linear Variable Differential Transformer Displacement Transducers (LVDTs) and Instrumentation", Corby, Northampton, U.K, (1997).
- Ryan, T. W., Hartle, R. A., Mann, J. E. and Danovich, L. J. (2012). "Bridge Inspector's Reference Manual", No. FHWA NHI 12-049, Federal Highway Administration, 1029 pp.
- Sabatini, P. J., Elias, V., Schmertmann, G. R. and Bonaparte, R. (1997). "Geotechnical Engineering Circular No. 2: Earth Retaining Systems", No. FHWA-SA-96-038.
- Slope Indicator Inc., "Pneumatic Total Pressure Cells", Product Data Sheet, Stone Mountain, GA, (2017).
- TDK Group, "NTC Thermistors for Temperature Measurement", TDK, Tokyo, Japan, (2018).
- Tennessee Department of Transportation (2003). "Retaining Wall Failure: Jefferson Street On-Ramp to I-40 West Davidson County".

- US DOT (1995). "Recording and coding guide for the structure inventory and appraisal of the nation's bridges", No. FHWA-PD-96-001, US Department of Transportation, Federal Highway Administration, 124 pp.
- Vessely, M. (2013). "Geotechnical Asset Management-Implementation Concepts and Strategies", No. FHWA-CFL/TD-13-003, Federal Highway Administration, 74 pp.
- Vessely, M., W. Robert, S. Richrath, V. R. Schaefer, O. Smadi, E. Loehr, A. Boeckmann, Geotechnical Asset Management for Transportation Agencies, Volume 2: Implementation Manual, NCHRP Research Report 903, National Academies, Washington DC (2019).
- VicRoads Technical Services (2014). "Road Structures Inspection Manual", State Government of Victoria, Victoria, Australia.
- Walters, B. X., Collins, M. P., Funk, N. E., Vessely, M. J., Widman, B. L., Koonce, J. W., Michael J. Garlich, M. J. and Paul D. Thompson, P. D. (2016). "Colorado Retaining and Noise Walls Inspection and Asset Management Manual", Colorado, Department of Transportation, 300 pp.
- Washington State Department of Transportation (WSDOT), "Geotechnical Design Manual", Publication No. M 46-03.04, Environmental and Engineering Programs, Olympia, WA, (2011).
- WisDOT (2017). "Structure Inspection Manual", Wisconsin Department of Transportation, Part4-Ancillary Structures, Chapter 4-Retaining Walls, 66 pp.
- Wolf, R. E., Bouali, E-H, Oommen, T., Dobson, R., Vitton, S., Brooks, C. and Lautala, P. (2016). "Sustainable Geotechnical Asset Management along the Transportation Infrastructure Environment Using Remote Sensing", Final Report to USDOT, No. RITARS-14-H-MTU, Michigan Technological University.
- Yu, C., C. Loureiro, J.-J. Cheng, L.G. Jones, Y.Y. Wang, Y.P. Chia, and E. Faillace, Data Collection Handbook to Support Modeling the Impacts of Radioactive Material in Soil, Environmental Assessment and Information Sciences Division, Argonne National Laboratory, Argonne, IL, (1993).
# APPENDICES

A - 1 Materials from FHWA-NPS WIP (2010)

	DO DET (DIDIO)	THE DUCTOR	DE DDOCD (M	UTD FIFTD FOD	14			
-N	S RETAINING	WALL INVENIO	KY PROGRAM (	WIP) FIELD FOR	M-			
NPS Park Name		Route/Parking No.		Wall Start Milepoint				
Inspected By		Route/Parking Name		Wall End Milepoint				
Inspection Date		Side of Centerline	(R.L.P##)	Visidata Event Milepoint				
	WAI	LL FUNCTION, DIMEN	SIONS, and DESCRI	PTION	-			
Wall Function		Primary Wall Type		Architectural Facings				
Approx. Year Built		Secondary Wall Types		Surface Treatments				
Wall General Description	Notes: (e.g., wall purpo	se, setting, construction, c	onsequence of failure, s	pecial design, etc.)				
Wall Length (ft)		Wall Face Area (ft <sup>3</sup> )		Wall Start Offset (ft)				
Max. Wall Height (ft)		Vertical Offset (+/- ft)		Wall End Offset (ft)				
Photo Description/No. (e.)	j., approach, elevation, wall	l top, alignment, face detail, d	leficiencies, etc.)	Face Angle (deg)				
Park Designated Wall ID								
	REPAIR /	REPLACE RECOMME	NDATIONS AND WO	RK ORDER				
Wall Condition Rating		Design Criteria		Failure Consequence				
Investigation Req'd?	(Y/M)	Cultural Concern?	(YA)	Action				
REPAIR /REPLACE RECOMMENDATIONS AND WORK ORDER           Wall Condition Rating         Design Criteria         Failure Consequence           Investigation Req'd?         (Y/N)         Cultural Concern?         (Y/N)         Action           Brief Work Order Description (5-10 word maximum, key work element;)          Action								
Rev. 07-10-2007				Repair/Replace COST:				

# Field Inspection Form (Front Page)

Element	Condition Narrative	Condition Rating	Weighting Factor 1	Condition Sears	Data Ratability
	Primary Wall Elements				
Piles and Shafts		1-10	8		1-3
Lagging		1-10	8		1-3
Anchor Heads		1.10	8		1.3
Wire/Geosynthetic					
Facing Elements		1.10	8		1.3
Bin or Crib		1-10	8		1-3
Concrete		1.10	8		1.3
Shotcrete		1-10	8		1.3
Mortar		1.10	8		1.3
Manufactured					
Block/Brick		1.10	8		1.3
Placed Stone		1.10	8		1.3
Stone Masonry		1-10	8		1-3
Wall Foundation			•		
Material		1.10	•		1.3
Other Primary			8		
Wall Element		1.10			1.3
	Secondary Wall Elements (WF=0.5 for CK=8-10/WF=	1.0 for C	K=4-7/W	F=5 for	(K=1-3)
Wall Drains		1-10	0.5-5		1-3
Architectural					
Facing		1.10	0.5.5		1.3
TISIDC Barrier Farry a		1.10			
Road/Sidewalk/		1-10	0.3-2		1-0
Shoulder		1-10	0.5-5		1-3
Upslope		1.10	0.5.5		1.3
Downslope					
		1-10	0.5-5		1-3
Lateral Slope		1.10	0.5-5		1.3
Vegetation		1.10	0.5.5		1.3
Culvert		1-10	0.5-5		1-3
Carb/Berm/Ditch		1-10	0.5-5		1-3
Other Secondary					
Wall Elements		1.10	0.5-5		1.3
	Wall Performance				
Performance			8		
	Waighting Factor (v10) and Condition Score Tetals	1-10			
WALL RATING	Wall Condition Rating (= [Condition Score Total/Weighting Factor Total (x10) 1 X 100)				
				-	

# Field Inspection Form (Back Page)

# WIP Field Guide (Page 1)

- NPS Retaining Wall Inventory Program Field Guide (WIFG)-								
	Retaining Wall Acceptance Criteria							
*All classes of *Walls must re *Maximum wa *Consider kno *Walls have an *Include all we	*All classes of paved roadways and parking areas included in the RIP Route Investigation Report and/or identified by Park staff. *Walls must reside within the constructed roadway/parking area prism. *Maximum wall height, including only that portion actively retaining soil and/or rock, must be $\geq 4$ ft ( $\geq 6$ ft for culvert headwalls). *Consider known/verifiable wall embedment in determining maximum retaining wall height. Include fully buried retaining structures. *Walls have an internal wall face angle $\geq 45^{\circ}$ ( $\geq 1$ H:1V face slope ratio).							
Include di wa	ins where the intent is to sup	Definitions	e would require replacement with a retaining wait.					
Design Criteria	Measure of how well current design criteria are satisfied: None - Does not meet any known standards. Non-AASHTO - Does not meet AASHTO, but is consistent with other structures of its type/period with good performance. AASHTO - Apparently meets current AASHTO Geometric, Design, Materials, and Construction Standards.							
Consequence of Failure	Low - No loss of roadway, Moderate-Hourly to short- High- Seasonal to long-term	no to low public risk, no impact to traffic du term closure of roadway, low-to-moderate p a loss of roadway, substantial loss-of-life ris	uring wall repair/replacement public risk, multiple alternate routes available sk, no alternate routes available					
Action	Select from: No Action, M	onitor, Maintenance, Repair Elements, R	eplace Elements, and Replace Wall					
Weighting Factor	Weighting Factor to be appl WF=0.5 for CR=8-10; WF=	ied to the Condition Rating (CR). When in -1.0 for CR=4-7; and WF=5 for CR=1-3.	dicated on the Condition Assessment Input Form:					
Data Reliability	Estimate of how well observed conditions represent wall performance, and if additional investigations may be warranted. 1-Poor Conditions cannot be sufficiently observed to rate element(s), warranting additional investigations to better define element performance and/or to deturnine the cause(s) or poor performance. ty 2-Good Observed conditions are sufficient to rate the conditions of wall element(s); however, additional investigations would be useful to better understand element performance. 3-Very Good Observed conditions clearly describe wall performance. Additional investigations are not needed.							
		Wall Function Codes						
[FW] Fill Wal [BW] Bridge V	l [CW] Cut Wall Wall [SW] Switehback Wa	ll [HW] Head Wall [SP] Slope Protection	on [FL] Flood Wall					
		Wall Type Codes						
[AH] Anchor,	Tieback H-Pile	[CC] Crib, Concrete	[MG] MSE, Geosynthetic Wrapped Face					
[AM] Anchor,	Micropile	[CM] Crib, Metal	[MP] MSE, Precast Panel					
[AS] Anchor,	Tieback Sheet Pile	[CT] Crib, Timber	[MS] MSE, Segmental Block					
[BC] Bin, Con	crete	[GB] Gravity, Concrete Block/Brick	[MW] MSE, Welded Wire Face					
[BM] Bin, Me	tal	[GC] Gravity, Mass Concrete	[SN] Soil Nail					
[CP] Cantileve	er, Concrete er, Soldier Dilo	[CC] Gravity, Dry Stone	[IP] Iangent/ Secant File					
[CS] Cantileve	n, Sheet Pile	[CM] Gravity, Mortaned Stone	[NO] None					
[co] canacto	.,	Architectural Facing Type	Codes					
[BV] Brick Ve	neer	[PF] Planted Face	[SS] Simulated Stone					
[CO] Cementi	tious Overlay	[SC] Sculpted Shotcrete	[SV] Stone Veneer					
[FF] Fractured Fin Concrete		[SH] Shotcrete (nozzle finish)	[TI] Timber					
[FL] Formline	d Concrete	[SM] Steel/Metal	[OT] Other, User Defined					
[PC] Plain Co	ncrete (float finish or light	[SO] Stone	[NO] None					
texture)		Surface Treatment Cod	les					
[BG] Bush Gu	n (tool-textured concrete)	[PS] Preservative	[WS] Weathering Steel					
[CA] Color Ad	lditive	[SE] Silane Sealer	[OT] Other, User Defined					
[CL] Galvaniz	red	[ST] Stain	[NO] None					
[PA] Painted		[TR] Tar Coated						

# WIP Field Guide (Page 2)

Condition Ratings						
Condition Ratings apply to all Primary and Secondary Wall Elements, and are intended to assist in consistently defining element severity, extent, and repair/replace urgency of wall element distresses.						
9-10 (Excellent)	-Any defects are minor and are within normal range for <i>newly constructed or fabricated</i> elements. -Defects may include those typically caused from fabrication or construction.					
7-8 (Cood)	-Low-to-moderate extent of low severity distress. -Distress present does not significantly compromise the element function, nor is there significantly severe distress to major structural components of an element					
5-6 (Fair)	<ul> <li>High extent of low severity distress and -Distress present does not compromise el element failure in the near term.</li> </ul>	(or low-to-medium extent of medium to high severity distress. lement function, but lack of treatment may lead to impaired function/elevated risk of				
3-4 (Poor)	-Medium-to-high extent of medium-to-high severity distress. -Distress present threatens element function, and strength is obviously compromised and/or structural analysis is warranted. -The element condition does not pose an immediate threat to wall stability and road closure is not necessary.					
1-2 (Critical)	-Medium-to-high extent of high severity distress. -Element is no longer serving intended function. Element performance threatening overall stability of the wall at the time of inspection.					
	Wal	ll Performance Condition Ratings				
Perform:	Evaluation of overall wall performance as indicated by observations not necessarily captured by observed distresses for specific elements, including global wall distresses (rotation, settlement, translation, displacement, etc.) and/or evidence of prior repairs that may further indicate component	Cood to Excellent - No observation of distresses not already captured by individual element condition assessment. No combination of element distresses indicating unseen problems or creating significant performance problems. No history of remediation or repair to wall or adjacent elements. (7-10) Fair - Some observed global distress is not associated with specific elements. Some observation of element distress combinations that indicate wall component problems. Minor work on primary elements or major work on secondary elements has occurred improving overall wall function. (5-6) Poor to Critical - Global wall rotation, settlement, and/or overtunning is readily apparent. Combined element distresses clearly indicate serious stability problems with components or slobal wall students have occurred to wall structural element.				



# WIP Field Guide (Page 3)

Element Element Definition		Element Condition Rating Guidance		
Prim	ary Element Condition Ratings	Good to Excellent Pating (mingr to no distract, minimal to no immact fam to no accurrence)		
Piles and Shafts	Soldier piles, sheet piles, micropiles or drilled shafts; supplemental structures such as walers, comprising all/part of the visible wall.	Good to Excellent Kating (minor to no distress, minimal to no impact, few to no occurrences) Corrosion/Weathering • No evidence of corrosion/staining, contamination or cracking/spalling due to weathering or chemical attack. • Compacted, placed or masoned rock, and associated chinking, is dense, angular, fresh, and without post-placement		
Lagging	Structural lagging between piles and walers.	fracturing or chemical degradation. • No significant weathering/weakening of bedrock, softening of soil, or saturated ground conditions evident.		
Anchor Heads	All visible parts of tieback anchor, including pad (observed without removing cap).	<ul> <li>No impacts from vegetation noted within the wall or within adjacent elements. Cracking Breaking</li> </ul>		
Wire/Geosyn. Facing Elements	Visible facing/basket wire, soil reinforcing elements, hardware cloth, geotextile/geogrids, and facing stone.	<ul> <li>No evidence of rememeric racking, oreaking, or construction post-construction damage, opening of discontinuities in prock, or cracks or gullies in soils.</li> <li>Concrete, shotrete, and mortar is sound, durable, and shows little or no signs of shrinkage cracking or spalling.</li> <li>Drains are clearly open (flowing), and in full working order.</li> </ul>		
Bin or Crib	Visible portion of cellular gravity wall.	<ul> <li>Wall elements are as constructed, and/or show no signs of significant settlement, bulging, bending,</li> </ul>		
Concrete	Visible precast or cast-in-place concrete wall and footing elements (does not include piles, lagging, crib blocks, manufactured block/brick, and architectural facing).	heaving, or distortion/deflection beyond normal prescribed post-construction limits. Lost Bearing/Missing Elements • No wall elements are missing, • Wall elements are fully bearing against retained soil/rock units. • Enumérican colle lock are more than a docume to support the wall, consistently done, drained and strang		
Shotcrete	Visible shotcrete (does not include piles, lagging, architectural facing or other specific elements).	No slope failures have occurred either removing or adding materials to the wall area.      Drie Rating (moderate distract, cimificant to cubstantial impact, multiple occurrences)		
Mortar	Visible mortar used between uncut or masoned rock, manufactured blocks or brick, or used for wall repairs.	An example (underster desures), segmentation to substantian impact, manipute occurrences)     Corrossion/Weathering     Moderate corrossion/staining, contamination or cracking/spalling due to weathering or chemical attack.     Compacted, placed or massoned rock is not fresh or angular, showing significant weathering,     most placement functions, chemical demonstration, which to based attack to a substantiant of the second seco		
Manufactured Block/Brick	Manufactured blocks and bricks, including CMU's segmental blocks, large gravity blocks, etc. (does not include concrete lagging or crib wall components).	<ul> <li>Significant weathering weakening of bedrock, softening of the soil, or saturated ground conditions evident.</li> <li>Moderate impacts from vegetation are evident within the wall or within adjacent elements.</li> <li>Cracking/Breaking</li> <li>Localized element cracking, breaking, abrasion and/or construction/post-construction damage,</li> </ul>		
Placed Stone	Dry-laid or mortar-set uncut rock.	opening or discontinuities in rock or cracks or gullies in soil. • Concrete, shotcrete, and mortar is occasionally soft or drummy, has lost durability, and shows occasional		
Stone Masonry	Dry-laid or mortar-set cut rock.	cracking and/or spalling sufficient to intercept reinforcement. • Drains cannot be clearly determined to be fully operational.		
Wall Foundation Material	Soil or rock immediately adjacent to and supporting the wall.	Distortion/Deflection * Wall elements show significant localized settlement, bulging, bending, heaving, misalignment, distortion, deflection, and/or dispacement beyond normal prescribed post-construction limits (e.g., wall face rotation,		
Other Primary Wall Element	Any primary wall element not listed (provide detailed narrative definition).	basket buldging, anchor head displacement, bin displacement). Loss Bearing/Missing Elements		
Secon	dary Element Condition Ratings	<ul> <li>Some wall elements are missing (e.g., chinking, lagging, brick-work) or non-functional.</li> <li>Wall elements are generally bearing against retained soil/rock units, but localized open voids may exist</li> </ul>		
Wall Drains	Function and capacity of visible drain holes, pipes, slot drains, etc., that provide wall subsurface drainage.	lalong the back and top of the wall. • Foundation soils rock are adequate to support the wall, but susceptible to shrink-swell, erosion, scour, or vegetation impacts. • Joolated Jone failures have occurred either removing or adding material from the wall area.		
Architectural Facing	Facing that is not relied on for structural capacity, including concrete, shotcrete, stone, timber, vegetation, etc.	Poor to Critical Rating (severe distress, failure is imminent, pervasive occurrences) Corrosion/Weathering		
Traffic Barrier/ Fence	Traffic barrier or fence above or below wall, and within the influence of the wall.	<ul> <li>Metallic wall elements are corroded and have lost significant section affecting strength.</li> <li>Concrete/shotcrete is extensively spalled, cracked, and/or weakened, and may show evidence of widespread</li> </ul>		
Road/ Sidewalk/ Shoulder	Road and/or sidewalk surface above or below a wall, and within the influence of the wall.	aggregate reaction. • Compacted, placed or masoned rock is highly weathered, showing extensive post-placement fracturing, chemical degradation, and/or loosening within the placed volume.		
Upslope	Groundslope area above a wall affecting wall condition and/or performance.	Extensive weathering weakening of bedrock, softening of soil, or saturated ground conditions evident. Severe impacts from vegetation are evident within the wall or within adjacent elements.		
Downslope	Groundslope area below the wall, distinct from the Wall Foundation Material element, affecting wall condition and/or performance.	Cracking/Breaking • Extensive severe element cracking, breaking, abrasion or construction/post-construction damage, opening of discontinuities in rock, or cracks or gullies in soils. • Concrete, shortere, and mortar is consistently soft, drummy, or missing, has lost durability and strength, and		
Lateral Slope	Groundslope laterally adjacent to a wall affecting wall condition and/or performance.	shows pervasive cracking and/or spalling intercepting corroding weathering reinforcement. • Drainage is missing, clearly damaged, and/or obviously clogged and non-functional.		
Vegetation	Vegetation near wall or on wall face affecting wall condition and/or performance.	Destortion/Deflection • Wall elements show extensive settlement, bulging, bending, distortion, misalignment, deflection, and/or dis- nulacement well beyond normal post-construction limits, including loss of ground reinforcement and retention		
Culvert	Culverts and inlets/outlets through, below, or adjacent to walls.	Lost Bearing/Missing Elements • Many or key wall elements are missing (e.g., placed wall stone, chinking, lagging) or non-functional.		
Curb/ Berm/ Ditch	Lined or unlined surface drainage feature above or below wall.	<ul> <li>Many or key wall elements are no longer bearing against retained soil/rock units, with visible open voids evident behind a large portion of the wall.</li> <li>Foundation colls finds there sizes of follows are aviant applications to come aviant application in the second se</li></ul>		
Other Secondary Wall Element	Any secondary wall element not listed (provide detailed narrative definition).	<ul> <li>rouncasion sous-rock show signs of fature, excessive settlement, scour, erosion, substantial voids, bench failure, slope oversteepening, and/or may be adversely impacted by vegetation.</li> <li>Substantial slope failures have occurred either removing or adding materials to the wall area.</li> </ul>		



# WIP Field Guide (Page 4)

A - 2 Materials from North Carolina DOT (2015)

Wall ID: Date: N	CDOT Reviewer(s):
Revision Date:	Picture(s):
LOCATION DATA	
County: Division:	Travel Direction:
Route Number: Route Name:	Latitude: Longitude:
Location Description:	
Bridge Association: Bridge Number	er: Culvert Association: Culvert Number:
Road System: Tier:	
DIMENSION DATA	
Embedment (ft): Max. Wall Heig	ght (ft): Total Length (ft):
Wall Batter: Back Slope: Fro	ont Slope: Berm Dimension: Distance to Stream (ft):
Roadside Features:	
Plan View:	Profile View:
WALL TYPE AND FUNCTION DATA	
Wall Type: Wall F	Facing Veneer:
Construction Type: Fu	Inction Type: Traffic Volume:

Wall Identification and Data Attributes Form Page 2
HISTORY AND OWNERSHIP
Current Owner: Year Built: Design Life: Engineer of Record:
T.I.P. Number: Design Type: Design Category: Inspection Frequency:
Special Access Needs: Work Zone Requirments:
STRUCTURAL DATA
Wall Support: Foundation Dimension: Fill Material:
Bridge Foundation Type: Soil Reinforcement Type: Surcharge:
Reinforcement off ROW: External Drainage: Internal Drainage:
Wall Top Feature: Scour Depth: Wall Obstruction(s):
As-Built Drawings: Design Calculations: Subsurface Plans:
Conflict(s):
Comments On Design:
Comments On Construction Details:
Maintenance/Repair Details:

Val	I ID:	Date:				NCDOT Inspec	tor(s):		
ivi	sion:	Count	y:			Route Numb	er:	Route Name:	
oca	ation Description	n:							
ati	tude:		Lon	gitude:			Wall Type:		
01	CATEGORY BSERVATIONS	PERC	CENT 1 2	BY RA'	TING 4	AVERAGE RATING		COMMENTS	
	Facial Deterioration								
	Staining								
50	Damage								
acing	Cracking								
	Joint Allignment								
	Joint Spacing								
	Material Loss								
	Deflection/ Rotation								
ment	Bulges/ Distortion								
Move	Settlement								
	Heaving								
e	Erosion								
rainag	Scour								
â	Internal/ External Drains								
-	Wall Top Attachment								
terior	Road/Sidewalk /Shoulder								

Note: If the average rating for any of the criteria listed above is  $\geq$  2.5, please include a corresponding picture and comment for each observation respectively.

A - 3 Materials from Colorado DOT (2016)



# **Condition Ratings**

<u>Code</u>	Description
N	NOT APPLICABLE
9	EXCELLENT CONDITION - like new condition
8	VERY GOOD CONDITION - Main structure has very minor (and isolated) deterioration.
	<ul> <li>Concrete: minor cracking, leaching, scaling, or wear (no delamination or spalling).</li> </ul>
	<ul> <li>Timber: minor weathering - isolated (minor) splitting.</li> </ul>
	<ul> <li>Steel: no corrosion (paint/protection system remains sound).</li> </ul>
	<ul> <li>Masonry: slight weathering or cracks (joints have no deterioration).</li> </ul>
7	GOOD CONDITION - Main structure has minor (or isolated) deterioration. Evidences of light
	leakage may be present.
	<ul> <li>Concrete: minor cracking, leaching, scaling, or wear (isolated delamination, spalling, or</li> </ul>
	temporary patches).
	<ul> <li>Timber: minor weathering or splitting (no decay or crushing) - all planks are secure.</li> </ul>
	<ul> <li>Steel: minor paint failure or corrosion (no section loss) - all connections are secure.</li> </ul>
	<ul> <li>Masonry: minor weathering or cracking (joints have little or no deterioration)</li> </ul>
6	SATISFACTORY CONDITION - Main structure has minor to moderate deterioration (no repairs are
	necessary). Areas of slight backfill loss. Areas of leakage are minor and isolated. Scour or erosion (if
	present) is minor and isolated.
	<ul> <li>Concrete: moderate cracking, leaching, scaling, or wear (minor delamination or spalling).</li> </ul>
	<ul> <li>Timber: moderate weathering or splitting (isolated decay or crushing) - some planks may be</li> </ul>
	slightly loose.
	<ul> <li>Steel: moderate paint failure and/or surface corrosion (minor section loss) – some connections</li> </ul>
	may have worked loose.
	<ul> <li>Masonry: moderate weathering or cracking (joints may have minor deterioration). Evidence of</li> </ul>
	slight freeze-thaw.
5	FAIR CONDITION - Main structure has moderate deterioration (repairs may be necessary). Areas of
	backfill loss are minor and isolated. Areas of leakage are minor. There may be moderate scour,
	erosion, or undermining. There may be slight settlement, movement, misalignment, or bulging.
	Change in vertical batter is <4% different from intended design.
	<ul> <li>Concrete: extensive cracking, leaching, scaling, or wear (moderate delamination or spalling).</li> </ul>
	<ul> <li>Timber: extensive weathering or splitting (moderate decay or crushing) - some planks may be</li> </ul>
	loose, broken, or require replacement.
	<ul> <li>Steel: extensive paint failure and/or surface corrosion (moderate section loss) – several</li> </ul>
	connections may be loose or missing, but deck components remain secure.
	<ul> <li>Masonry: extensive weathering or cracking (joints may have slight separation or offset).</li> </ul>
	Evidence of minor freeze-thaw.
4	POOR CONDITION - Main structure has advanced deterioration (replacement should be planned).
	inderate backfill loss and/or leakage may be present. There may be extensive scour, erosion, or
	undermining. Minor settlement, movement, misaignment, or buiging may be present. Changes in
	Vertical batter are within 4%-10% compared to intended design.
	<ul> <li>concrete: advanced cracking, leaching, scaling, or wear (extensive delamination or spalling) - isolated full donth failures may be imminent.</li> </ul>
	Timber: advanced weathering splitting or decay - numerous planks may be loose, broken, or
	require replacement
	<ul> <li>Steel: advanced corrosion (significant section loss) – main structure elements may be loose or</li> </ul>
	slightly out of alignment
	<ul> <li>Masonry: advanced weathering or cracking (joints may have separation or offset). Evidence of</li> </ul>
	moderate freeze-thaw.

3	SERIOUS CONDITION - Main structure has severe deterioration - immediate repairs may be
	necessary. Heavy and/or active backfill loss or leakage may be present. Scour, erosion, or
	undermining may have resulted in severe settlement, movement, or misalignment. Significant
	settlement, movement, misalignment, or bulging may be present. Changes in vertical batter may
	be >10% compared to intended design.
	Concrete: severe cracking, leaching, delamination, or spalling - full-depth failures may be
	present.
	<ul> <li>Timber: severe splitting, crushing or decay - majority of planks may need replacement.</li> </ul>
	<ul> <li>Steel: severe section loss – main structure elements may be severely out of alignment.</li> </ul>
	<ul> <li>Masonry: severe cracking, offset or misalignment. Evidence of severe freeze-thaw.</li> </ul>
2	CRITICAL CONDITION - advanced deterioration of primary structural elements or extreme backfill
	loss. Severe backfill loss may be affecting the structural integrity of the wall. Cracks in steel or shear
	cracks in concrete may be present or scour may have removed substructure support. Unless closely
	monitored it may be necessary to close the routes carried or in front of the wall until corrective
	action is taken.
1	"IMMINENT" FAILURE CONDITION - major deterioration or section loss present in critical structural
	components or obvious vertical or horizontal movement affecting structure stability. Route carried
	or in front of the wall is closed to traffic but corrective action may put back in light service.
0	FAILED CONDITION - out of service - beyond corrective action.

# Appraisal ratings

Description	<u>Code</u>
Not applicable	Ν
Superior to present desirable criteria	9
Equal to present desirable criteria	8
Better than present minimum criteria	7
Equal to present minimum criteria	6
Somewhat better than minimum adequacy to tolerate being left in place as is	5
Meets minimum tolerable limits to be left in place as is	4
Basically intolerable requiring high priority of corrective action	3
Basically intolerable requiring high priority of replacement	2
This value of rating code not used	1
Wall failure	0

APP	ENDIX 2.A: STRUCTURE INVENTORY AN	D APPRA	ISAL SHEET
1.	State Name:	50A.	Curb or Sidewalk, Carried (lt)
2E,M	Region/Maint. Section	50B.	Curb or Sidewalk, Carried (rt)
з.	County Code	50C.	Curb or Sidewalk, In Front (lt)
4,4A.	Place Code	50D.	Curb or Sidewalk, In Front (rt)
5A.	Inventory Rte (On/In Front)	51.	Width, Curb-to-Curb, Carried
5B.	Inventory Rte (Signing)	51A.	Width, Curb-to-Curb, In Front
5C.	Inventory Rte (L.O.S.)	52.	Average Wall Height
5DN.	Inventory Rte (Rte. Num)	53.	Maximum Wall Height
5E.	Inventory Rte (Dir. Suffix)	54.	Minimum Wall Height
6.	Features Carried	58.	Main Structure
7.	Features in Front	60.	Foundation
8.	Structure Number	61.	Channel & Channel Protection
8A.	Structure ID Number	71.	Waterway Adequacy
9.	Location	72.	Adjacent Roadway Alignment
11.	Reference Point	90.	Inspection Date
12.	Base Hwy. Network	91.	Frequency
13.	LRS Rte, Sub. Rte Number	92A.	UW Frequency
16.	Latitude - Start of Wall	92B.	Special Frequency
16A.	Latitude - End of Wall	93A.	UW Inspection Date
16B.	Elevation - Start of Wall	93B.	Special Inspection Date
17.	Longitude - Start of Wall	98A.	Border Wall State Code
17A.	Longitude - End of Wall	98B.	Percent Responsibility
17B.	Elevation - End of Wall	99.	Border Wall Structure No.
18A.	Range	100.	STRAHNET Highway Desc.
18B.	Township	102.	Direction of Traffic
18C.	Section	103.	Temporary Structure
19.	Bypass, Detour Length	104.	Highway System
20.	Toll	105.	Federal Lands Highways
21.	Maintenance Resp.	106.	Year Reconstructed
22.	Owner	107.	Vertical Support Struct. Type
23.	Orig. Project Number	109.	Truck ADT
23E.	Subaccount Number	110.	Designated National Network
23EE.	Project Indicator	113.	Scour Critical Walls
26.	Functional Class	114.	Future ADT
27.	Year Built	115.	Year of Future ADT
28.	Lanes On Structure	125A,B.	Type of Wall Rail on Top
	Lanes In Front	125C,D.	Type of Wall Rail In Front
29.	Average Daily Traffic	133.	Special Inspection Equipment
30.	Year of ADT	136.	Mileage Log Section Letter
33.	Median	143A.	Avg. Dist Fr. Route, Carried
36AD	Traffic Safety Feat. on Top		Max. Dist Fr. Route, Carried
36H.	Height of Rail on Top		Min. Dist. Fr. Route, Carried
36IL.	Traffic Safety Feat. in Front	143B.	Avg. Dist Fr. Route, In Front
36M.	Height of Rail In Front		Max. Dist Fr. Route, In Front
37.	Historical Significance		Min. Dist. Fr. Route, In Front
42.	Type of Service, Carried	144A.	Speed Limit, Route Carried
	In Front	144B.	Speed Limit, Route In Front
43.	Struct Type, Wall Face Mat.	145.	Wall System
	Type of design/const	146.	Associated Bridge
47.	Total Horz. Clearance, Carried	147.	Vertical Batter
47A.	Total Horz. Clearance, In Front	148.	Slope Angle Carried & In Front
49.	Structure Length	149.	Type of Protective Coating



#### Retaining and Noise Walls Inspection and Asset Management Program

			Material D	efects					
Steel         Poc (100)           Corrosion (1000)         Delam/Spair/ Patched Area (1940)         Separation (1940)           Corrosion (1000)         Patched Area (1940)         Corrotion (1940)         Corrotion (1940)           Corrosion (1000)         Exposed Prebressing         Corrotion (1950)         Corrotion (1950)           Distortion (1920)         Exposed Prebressing         Casking Growth (1960)         Corrotion (1950)           Distortion (1950)         Emposed Prebressing         Casking (2000)         Corrotion (1950)         Corrotion (1950)           Vertical (1950)         Filternerst (4000)         Corrotion (1950)         Sour (6030)         Corrotion (6030)           Separation (1950)         Filternerst (1950)         Corrotion (1950)         Corrotion (6030)         Corrotion (6030)           Corrotion (1950)         Filternerst (1950)         Corrotion (1950)         Corrotion (6030)         Corrotion (6030)           Corrotion (1950)         Filternerst (1950)         Filternerst (1950)         Corrotion (1950)         Corrotion (6030)           Corrotion (1950)         Filternerst (1950)         Corrotion (1950)         Corrotion (1950)         Corrotion (1950)           Corrotion (1950)         Filternerst (1950)         Corrotion (1950)         Corrotion (1950)         Corotion (1950)           Cor	RC (400)         Timber (500)           Delamispail Patched Area (1020)         Connection (1020)           Delamispail Patched Area (1020)         Delayispail Section Loss (1140)           Ebroadd Stands (1190)         Delayispail Section (1140)           Efforescence / Rust Staning (1110)         Crack (1160)           Crackar (1160)         Crack (1160)           Cracking (RC (1160)         Splitbelamis (1170)           Abrasion/ (1160)         Distortion (1180)           Builging (1910)         Distortion (1920)           Noration (1930)         Builging (1910)           Vertical Rotation (1930)         Vertical Rotation (1930)           Separation (1930)         Separation (1930)           Caramit (1950)         Cramit (1930)           Settement (4000)         Caramit (1930)           Settement (4000)         Settement (4000)           Settement (4000)         Settement (4000)	Other Waterlais (600)           Corrision (1000)         Leskage (2310)           Conscion (1000)         Seal Adhesion (2320)           Connection (1000)         Seal Adhesion (2330)           DetamSpail/ Pather Area (1580)         Seal Conscion (2340)           DetamSpail/ Pather Area (1580)         Seal Conscion (2340)           DetamSpail/ Pather Area (1500)         Debris Debris (2370)           DetamSpail/ (120)         Debris Impacton (2370)           Detargrafion (1900)         Metal Detargrafion or Damage (2370)           Distortion (1920)         Settlement (4000)           Hotzontal (1900)         Settlement (4000)           Detargrafion (1920)         Temporary Support (1920)           Hotzontal (1900)         AsR (8030)	Materia D (150) Detam:Spail/ Patoned Area (100) Exposed Rebur/WWP( 1000) Efforescence (1120) Efforescence (1120) Efforescence (1120) Efforescence (1120) Efforescence (1120) (1120) Mortar Breakdown (1950) (195	CRS protective (S20) Effectiveness Protective (S300) (J300) (J300) (J300) (J300) (J300) (G30	Steel Protective Castings (850) Graffiti (1950) Charling (Steel Protective Castings) (3410) Crashing (Steel Protective Castings) (3420) Crashing (Steel Protective Castings) (3420) Crashing (Steel Protective Castings) (3420) Castings) (3420) Castings) (3420) Castings) (3420) Castings) (3430) Castings) (3430) Castings) (3430) Castings) (3430) Castings) (3430) Castings) (3430) Castings) (3430) Castings) Castings) (3430) Castings) Castings) (3430) Castings	Concrete and Masonry Protective Coatings (870) (150) (150) Vegetation Growth (1980) Wear (concrete Protective Coatings) (3510) Effective Scoatings) (3500) Challing (Concrete and Masonry Protective Coatings) (3550) Protective Coatings) (3550) Protective Coatings) (3550) Peeling/ Bubbling Crasting (Concrete Protective Coatings) (3560)	Timber Protective Coatings (890) Graffiti (1950) Vegetation Growth (1960) Wear (Timber Protective Coatings) (3610) Effectiveness (Timber Protective Coatings) (3620) Damage (7000)	Other Protective (900)     Retained (910)       Craftiti (1950)     Vegetation (1960)       Vegetation (1960)     Backfill (960)       Wear (other Protective Coatings)     Backfill (900)       Effectiveness (000)     (000)       (000)     Estion (2000)       Damage (7000)     Damage	Diahage (320) Corrosion (1000) Corrosion (1000) Consection (1000) Delam(Spati/ Pathed Area (1080) Delam(Spati/ Pathed Area (1080) Consection (1120) Cracking (RC and Other) (1120) Delatorization (Other) (1200) Delatorization (1500) Delatorization (1500) Delatorization (1500) Used Consection (1500) Delatorization (1500) Used Consection (1500) Used Consection (1500) U
Temporary Support (8020)	Erosion (5030)         Sedimenti (4000)           Sobur (6000)         Erosion (5000)           Damage (7000)         Scour (6000)           Temporary (8020)         Uanage (7000)           Support (8020)         Temporary (8020)	ý				Abbreviations: PSC: Prestressed Delaru: Delaruit	Concrete; RC: Reinfo	rced Concrete;	Blockage (1970) Leakage (2310) Erosion (5030) Damage (7000)

Figure 3.B-1 Material Defects

Delam: Delamination; CRS: Concrete Reinforcing System; ASR: Alkali-Silica Reactivity

Scoring criteria for Annual Average Daily Traffic (AADT)

<u>Score</u>	Descriptor
Wall Condition	
1	New condition or no noticeable condition loss
2	Acceptable performance, prior maintenance/repair evident
3	Deterioration or condition loss occurring
4	Potentially unstable conditions
Consequence of Failure	
1	Negligible: No impacts to structures, roadway, or off ROW property
2	Minor: No to slight impact to traffic (temporary: less than 1 day)
3	Major: Impact beyond CDOT ROW and/or debris on roadway
4	Critical: Collapse of at least one travel lane (essential structure)
Annual Average Daily Traff	ic (AADT)
1	AADT below the 25 percent quartile (less than 13,000)
1.25	AADT between 25 and 50 percent quartile (13,000-31,000)
1.5	AADT between 50 and 75 percent quartile (31,000-107,000)
2	AADT greater than 75 percent quartile (greater than 107,000)

Maintenance and mobility risk parameters

Performance <u>Area</u>	Factor	<u>Parameter</u>
Maintenance		
	Consequence	Element Category (primary or secondary)
		Element Type
		Quantity of Elements
		~Unit Costs
	Likelihood	Condition State
Mobility		
	Consequence	Avg. Wall Height
		Avg. Distance from Road in Front
		Avg. Distance from Road Carried
		AADT
		^Delay Time, 2 hours
		*User Value, \$30.50
		*Occupancy Rate, 1.67
		*ADT Delay, 33% of Actual ADT
	Likelihood	Main Structure Condition
		Foundation Condition
		Scour Critical

"Data compiled based on inspector experience and with CDOT input.

^Assumed value based on likely time of delay from an urgent adverse event, similar to delay associated with over-height bridge strikes.

\*Per AASHTO 2010, see Footnote 1.

Likelihood estimates for element cate	egories and condition states
---------------------------------------	------------------------------

Likelihood	Likelihood of Incurring Maintenance Cost				
Condition	Primary	Secondary			
State	Elements	Elements			
CS1	0%	0%			
CS2	11%	7%			
CS3	59%	37%			
CS4	98%	66%			

Probability of event and condition score

<b>Condition</b>	<u>Likelihood</u>		
9			
8	2%		
7			
6	E0/		
5	570		
4	26%		
3	2070		
2			
1	78%		
0			

Grade categories for parameters

<u>Level of</u> <u>Risk</u> <u>Grade</u>	<u>Maintenance</u> <u>Risk Score</u>	Mobility Risk Score
Α	0 to 0.1	\$0 to \$500
В	0.1 to 0.3	\$500 to \$5,000
С	0.3 to 0.5	\$5,000 to \$40,000
D	0.5 to 0.7	\$40,000 to \$95,000
F	>0.7	\$95,000

A - 4 Materials from Wisconsin DOT (2016)

## TO: STRUCTURES DEVELOPMENT SECTION

(January, 2016)

FROM:

## SUBJECT: Noise or Retaining Wall Inventory Data (Complete all data fields applicable)

1. Structure #	2. Region	3. County	4.1	Municipality	5. Ot	wner	6. Maintainer	10-State 30-County	40-Town 41-City	42-Vi1
7. Start. Latitud	e 8. Star	t. Longitude		9. End Latitu	de	10. End Lo	ongitude	11. Туре (	Retaining or	Noise)

## WALL TYPE AND GEOMETRICS

12.	Type CIP Piles CIP-Spread Block-Gravity MSE-Block MSE-Panel Post and Panel SHT Pile-Cant. Other
13.	Is Wall Anchored (Y/N)
14.	Installation Type(s) (check all that apply):  Ground Mounted  Structure Mounted (Bridge ID=)
15.	Foundation Type (if Applicable):  Piles Spread Footing Other:
16.	Pile/Post Material & Type (If Applicable):         Untreated Timber         Concrete (CIP)         Concrete (Precast)         Steel H-Pile         Other:
17.	Bile/Post Size (If Applicable):           8"         10" or 10 3/4"           12"         14"           Other
GEOM	IETRIC DATA
18.	Structure Length (ft.)
19.	Maximum Wall Height (ft.)
20.	Maximum Exposed Wall Height (ft.)
21a. 21b.	Average Wall Thickness (in.) Wall Area (ft <sup>2</sup> )
ROUT	E INFORMATION
22.	Enter name of closest primary route to Wall (Record in HSIS as Under Route):
23.	Direction: North East South West
24.	Designation: 🗌 Mainline 🗌 Other
25.	Inventory Route: On NHS Not on NHS

## MISCELLANEOUS INFORMATION

26.	Wall Functions:  Retain Cut  Dockwall  Multiple	Retain Fill Noise Wall
27.	Noise Abatement Met	hod (If Applicable):

Noise Abatement Method (If Applicable):
 Single Side Absorptive
 Double side Absorptive
 Reflective
 Other\_\_\_\_\_\_

## CONSTRUCTION DATA

28.	Plans Completed	YR	MO	DAY
29.	Letting Date	YR	MO	DAY

20. Year Built

31	WORK Performed	
	New Structure	Other

- 32. Designer:
- 33. Fabricator:
- 34. General Contractor:
- 35. Project ID: \_\_\_\_
- 36. Cost: \_\_\_\_\_

# FOR INTERNAL USE ONLY

- 37 Type Service On: <u>Retaining/Noise Wall</u>
- 38 Type Service Under: <u>Highway</u>
- 39 Primary Route On: <u>Retaining/Noise Wall</u>
- 40 Route on Designation: Water/Land/Other

225

#### General Instructions and Help

- Box 1. Enter the assigned ID for the structure. Structure ID's can be obtained by contacting the Regional Inspection Program Manager.
- Box 2. Enter the region where the structure is located. If located on a boundary between two Regions, enter the Region associated with the structure ID.
- Box 3. Enter the county where the structure is located. Note that the county must match the structure ID value.
- Box 4. Enter the municipality where the structure is located.
- Box 5/6. Enter the maintainer/owner of the structure. In some instances these codes will be different, but this is rare
- Box 7/8. Enter the Start Latitude and Longitude of the structure using the following format: DDMMSS.S where DD is degrees, MM is minutes, and SS.S is seconds. This shall be taken at the beginning of the wall, at the lower station value, as shown on the plans.
- Box 9/10. Enter the End Latitude and Longitude of the structure using the following format: DDMMSS.S where DD is degrees, MM is minutes, and SS.S is seconds. This shall be taken at the end of the wall, at the higher station value, as shown on the plans.
- Box 11. Indicate the type of wall (Retaining or Noise).
- Box 12. Indicate the type of wall system that was designed.
- Box 13. If the wall uses anchorages to tie back into the soil mass, indicate that here.
- Box 14. Walls may either be ground mounted, structure mounted, or a combination. If the wall is attached to another structure, indicate that structure number as well.
- Box 15. If applicable, enter the type of foundation used for the wall system
- Box 16. If applicable, enter the type and material of the posts/piles used in the wall system.
- Box 17. If posts/piles are used, please indicate the size.
- Box 18. Enter the overall length of the wall structure, in feet.
- Box 19. Enter the maximum overall height of the wall, in feet.
- Box 20. Enter the maximum exposed wall height, in feet. In general, this is the height of fill the wall is holding back.
- Box 21. Enter the average wall thickness, in inches for 21a. Enter the cross-sectional area of the wall, in square feet for 21b.
- Box 22. Enter the closest route for public highway traffic to the wall.
- Box 23 Enter the direction of travel for the highway listed in box 22.
- Box 24. Enter the function the primary roadway under performs (i.e. Mainline, Ramp, Frontage Road, etc.)
- Box 25. Indicate if the primary roadway under carries the National Highway System. All IH and USH carry the system, but so do some other roads. More information on this can be found on this website: <a href="http://www.fhwa.dot.gov/planning/national\_highway\_system/nhs\_maps/wisconsin/index.cfm">http://www.fhwa.dot.gov/planning/national\_highway\_system/nhs\_maps/wisconsin/index.cfm</a>
- Box 26. Enter the primary functionality of the wall. If multiple, check all that apply and describe in the multiple space provided.
- Box 27. Enter the noise abatement method as described on the plans.
- Box 28. Indicate the date the plans were completed, if known.
- Box 29. Enter the letting date, if known.
- Box 30. Enter the year built, or the estimated year the work will be performed.
- Box 31. Enter the type of work that was done on the structure.
- Box 32. Enter the designer of record for the structure.
- Box 33. Enter the primary fabricator of the structure.
- Box 34. Indicate the contractor who installed or will install the structure in the field.
- Box 35. Enter the construction project ID, if known.
- Box 36. If project has been completed, enter the total cost of the structure. Otherwise leave blank.

Box 37-40 Used by Central Office staff in entering data into HSIS. No further information is required.

# A - 5 Materials from British Columbia Ministry of Transportation (2013)

Ministry of Tansportation and Infrastructure	Retaining Wall Inventory	Legend Optional Mandatory
Structure R Structure Number	WALL Inventory Date (yyyy/mm/dd)	
Retaining Wall Details Total Length (m):	Slope of Wall Year Bui ice (degree): (yyy	(Not entered in system)
Average Wall Height (m):	leight (m): Slope Above V (degre Length Slope Above Wall (m)	Vall ses)
Retaining Wall Information Wall Type:	Reinforcing Material:	
	System Type	
	Approach Flares	
Wall Location Relative To Road	ffset Side Offset From Center Line Distance (m)	
Inspections Month:	Inspection Interval Type: Inspection Interval (months):	Ŀ
Sequence No.:	Inspection Interval Type: Inspection Interval (months):	Ŀ
Present Value (\$):	sent Value Date Construct (yyyy/mm/dd): Cost (	ion \$):

Page 1 of 2

## Retaining Wall Details LOV

Wall Ty	pe Description:				
Concre	te:	L02.	Landscape Precast Crib		
C01.	Concrete Cast in Place	L03.	Landscape Precast Concrete		
C02.	Precast Small Block - No Mortar	L04.	Landscape Geosynthetic,		
C03.	Precast Small Block - Partial Height	Cel	lular Type Wall	S05.	Steel Welded Wire/Wire Mesh
Mor	tar	L05.	Landscape Geosynthetic	S06.	Steel Sheet Pile
C04.	Precast Small Block - Full Height	Wra	apped Slope	S07.	Steel Soldier Pile
Mor	tar	L06.	Landscape - Other	S08.	Steel Piles Wood Lagging
C05.	Precast Lockblock - unreinforced	Rockfil	I Faced Wall:	S09.	Steel Piles Concrete Lagging
C06.	Precast Lockblock - reinforced	R01.	Rock Wall: Mortar	S10.	Steel Multiplate
C07.	Precast Panel "Hexagonal" - MSE	R02.	Rock Wall: No Mortar	S11.	Steel Other
C08.	Precast Panel "Plus" - MSE	R03.	Rock Gabion	Wood:	
C09.	Precast Panel "Diamond" - MSE	R04.	Rock - Other	W01.	Wood Log Crib - Treated
C10.	Precast Panel "Rectangular" - MSE	Sandba	ags:	W02.	Wood Log Crib - Untreated
C11.	Precast Crib	B01.	Sandbags	W03.	Wood, Sawn Crib - Untreated
C12.	Shotcrete	B02.	Concrete Filled Sandbags	W04.	Wood, Sawn Crib - Treated
C13.	Precast Stacked CRB	Steel F	ace Walls:	W05.	Wood Soldier Piles
C14.	Precast - Stesswall	S01.	Steel Binwall Concrete Paneled	W06.	Wood Piles Wood Lagging
C15.	Concrete - Other	S02.	Steel Binwall Wood Paneled	W07.	Wood - Other
Landsc	ape Wall:	S03.	Steel Binwall Steel Panel	X. Othe	<u> </u>
L01.	Landscape Precast RECO	S04.	Steel RECO MSE	Z. Unk	nown

# Reinforcing Material

- Code: Description:
  - A Concrete
  - В Metal
  - С Plastic
  - D Timber
  - Е Other
  - F Unknown
  - G None

# Drainage System

## Code: Description:

- A Drain pipes or weep holes visible on wall face
- B Base Draining Type Only
- Through Wall Facing С
- D Unknown
- E None

## Jurisdiction Agency ID

#### Number: Name:

- Ministry of Transportation 1
- 2 Other Provincial Ministry
- 3 Railroad
- Federal government Agency 4
- 5 Municipal Government
- 6 Company or Corporation
- 7 Private Citizen
- 8 Unknown (Historical)
- 9 Other

# Code: Description:

- L Left
- R Right
- Both Left and Right В

# Wall Location Relative to Road

- Code: Description: Α Above
  - В Below
- Offset Side

BRITSH COLUMBA Transportation and Infrastructure	Ministry of Transportation and Highways CON	RETAINING WALL	Inspection Type Routine
Structure Number	Structure Name		Inspection Date (yyyy/mm/dd)
COMPONENT	PERCENT CONDITION RATI Enter % in each condition. See BMIS User Manual 15.2.7	NG INSPECTIO All poor or very poor or documented by photos. L	ON NOTES BY COMPONENT onditions should be explained with notes and Label explanation(s) with component numbers.
HYDROTECHNICAL Channel Erosion Protection Substructure Scour FOUNDATION Wall Foundation	E G F P V X N		
STRUCTURE Movement of Wall Retaining Wall Embankment Embankment Tie back/Connectors Wall Drainage System Coating Railings Railings Roadway Flares		Partial Inspection Notes:	
		General Inspection Notes:	
		Utility Concern Notes:	
		Urgency Rating Notes:	
	Inspector follows	Condition Co         E       Excellent       V       Ve         G       Good       X       No         F       Fair       N       No         P       Poor       For Condition Guidelines see BMIS User Manual 15.2.2	des       Urgency Rating         ary Poor       ut Inspected         bt Applicable       For definition see BMIS         e       User Manual 15.2.9         "4" and "5" rating must         be explained.
H0382F.doc (Updated 2011-02	Inspector (please t)	pe or print)	Signature Side 1 of 2

Structure Number

## Instrumentation Notes

Drainage Area Description (water level fluctuation, logging debris, etc.)

## Scour Notes

\_\_\_\_

\_

### Rehab Work Notes

Maintenance Work Notes

H0382F.doc (Updated 2011-02-24)

Side 2 of 2

A - 6 Materials from Victoria Roads (2014)



Condition rati	onation rating sneet													
Structure ID No.: S	tructure ID No.: S Location (Km): (Fwd/Rev)													
Road name:	Noad name: Road number:													
Crossing/General L	Crossing/General Location:													
Region:	Region: Map reference: (Melways/VicRoads)													
Inspector: Date:														
Component Widening % of component in each condition														
No.	L/R	1/2		1			2			3			4	

Condition rating sheet

Notes

Shoot of	 		



# Structure defect sheet

Structure ID No.: S	Location (Km): (Fwd/Rev)				
Road name:		Road number:			
Crossing/General Location:					
Region:	Map reference	e: (Melways/VicRoads)			
Inspector:		Date:			

COMP. No.	COMP. Name	Location	Quantity	Photo Nos.	Defect description
Sheet of					

234



# Structure information sheet

Structure ID No.: S	Location (Km): (Fwd/Rev)				
Road name:	Road number:				
Crossing/General Location:					
Region:	e: (Melways/VicRoads)				
Inspector:	Date:				

COMP. No.	Information or comment (Including: Load, Height, speed limits; hydraulic performance or similar; and location of any material testing & sampling)

Sheet of



S	ketch sheet																																			
s	itructure ID No.: S														Location (Km): (Fwd/Rev)																					
R	Road name:														Ro	ad	nu	mb	er:																	
С	ros	sin	g/(	Gen	era	al Lo	oca	tio	n:																											
R	Region: Map reference: (Melways/VicRoads)											_																								
Ir	inspector:														Date:																					
	Γ		Γ	Τ	Γ	Γ		Γ														<u>+</u>														
	t	$\uparrow$	t	$\uparrow$	t	F	T	╞	$\square$	$\vdash$																										F
-	T		t	$\top$	F	F																														F
	T		T	T	F	F		F																												F
			Γ		Γ																															Γ
																																				L
																																				L
			L		L	L		L																												L
			L		L	L		L																												L
					L																															L
	$\vdash$			$\perp$	L																															L
			┡																																	Ļ
			┡	$\vdash$	┡			┡																												Ļ
	╞		╞	$\vdash$	┡	┡		┡																												┞
	$\vdash$		┡	+	┡	┝		┡																												┡
	$\vdash$	$\vdash$	┡	+	┡	┡		┡																												┡
	$\vdash$	$\vdash$	┞	+	┡	$\vdash$	-	$\vdash$																		_								$\neg$		┞
	$\vdash$	$\vdash$	┞	+	┞	┝		$\vdash$	-		$\vdash$								$\vdash$							_							$\vdash$	-		┞
	$\vdash$	$\vdash$	┞	+	╞	┝	-	$\vdash$	-	$\vdash$	$\vdash$								$\vdash$							_							$\left  - \right $	$\neg$		┞
	$\vdash$	$\vdash$	┞	+	┝	┝	-	$\vdash$	$\vdash$	$\vdash$	$\vdash$		-			-	-		$\vdash$							_							$\vdash$	-		┞
	$\vdash$	+	┞	-	╞	╞	$\vdash$	╞	$\vdash$	$\vdash$	$\vdash$		-	-		-			$\vdash$	$\vdash$	$\square$	$\square$	$\vdash$			_				$\square$			$\left  \right $	$\neg$		┞
	╞	+	┞	+	┢	┢	$\vdash$	╞	$\vdash$	$\vdash$			$\vdash$			$\vdash$	$\vdash$		$\vdash$	$\vdash$	$\vdash$	$\square$	$\vdash$			_			$\vdash$	$\square$				$\neg$		┞
	╞	+	┞	+	┢	┢	$\vdash$	╞	$\vdash$	$\vdash$	$\vdash$		$\vdash$			$\vdash$	$\vdash$		$\vdash$	$\vdash$	$\vdash$	$\vdash$	$\vdash$			_			$\vdash$	$\square$				$\neg$		┞
	$\vdash$	+	╞	+	┢	╞	$\vdash$	╞	$\vdash$	$\vdash$			$\vdash$	-	-	-			$\vdash$	$\vdash$	$\vdash$	$\vdash$	$\vdash$			_			$\vdash$	$\vdash$				$\neg$		┢
		<u> </u>	L	_	1	1	1	1	1																											L

| Road Structures Inspection Manual

50



Inventory and photographic record sheet

🛛 Visual Screen Walls (SV) 👘 🗆 Noise Wa	alls (SZ)	Retaining Walls (SR)
Structure ID No.:		Location (Km): (Fwd/Rev)
Road name:		Road number:
Crossing/General Location:		
Region:	Map reference	e: (Melways/VicRoads)
Inspector:		Date:
GPS start of wall		
Latitude South:	Longitude East:	
Location:		
GPS end of wall		
Latitude South:	Longitude East:	
Location:		

For Visual Screen or		Type of wall: (freestanding, on parapet or retaining wall, other)												
Noise	e Wall	Material: (steel, con	Material: (steel, concrete, timber, masonry, other)											
For R	etaining Wall	Materials: Facing, St	Supports											
Side	of road: (left	or right)		Clearance from carriag	eway: m									
Phot	Photos													
No.	Location		Description		Comment									
1.														
2.														
3.														
Gene	aral comments													

Sheet of






#### Component 85 P – Walls - precast concrete

Condition state 1



The elements are in good condition with no damage visible. There may be minor dampness or efflorescence powder visible in a few locations. No separation or relative movement between units is apparent

#### Component 85 C – Walls - cast-insitu concrete

Condition state 1



The elements are in good condition with no damage visible. There may be minor dampness or efflorescence powder visible in a few locations. Wall slopes are true to line with no separation of the cast sections.

Condition state 2 (no photo)



Shrinkage cracks obvious and early signs of corrosion of reinforcement.



#### Component 86 S - Columns - steel

Condition state 1



Steel supporting I beams in sound condition. No signs of corrosion or deflection

Condition state 2



Incomplete filling of concrete around supporting steel posts with potential for water penetration and corrosion

#### Component 86 C – Columns - cast-insitu concrete



# Condition state 1

Exposed bored pile retaining wall in sound condition with no evidence of significant cracking, movement, water penetration or corrosion. Condition state 2 (no photo)



#### Component 87 P – Foundations/Supports - pre-cast concrete

Condition state 1



The foundations or supporting walls are in sound condition. No visible evidence of damage from vehicle impact. No visible movement due to settlement or rotation of wall. Drainage system at base of wall is structurally intact and operating effectively.

Condition state 2 (no photo)

A - 7 Materials from the Kentucky Transportation Cabinet (1998)

0010						1	-	DISTR	E 8 8	1	2	3
ROUT	F			1. RIGID	1. END BENT		MAGE NO.	I. VERT, DIPP.	5. SPALLING	4	5	•
MILER	POST: _			A. ANCH	-	٦.	HOLL HO.	3, FORWARD DISP.	7. STAINING	7	8	9
DIREC	CTION:			*, PREPAR	·	Ę	PHOTO NO.	4. CRACKING	. OTHER	WA1	LL PROP	IL E
GPS:	_					ŀ	TAPE NO.	GENERAL NOTI	ES OR SKET	CH OF	WALL	
WALL	HEIGHT:					8	START					
WALL	LENGTH:_					۶ ا	END					
QUAD	DISTRESS	рното	THERMO		c	0	MENTS					
						-	5					· .
1												
2												
_												
З												
4												
•				ļ								
5												
6												
7												
8												



Figure 12: Distress Observed on Concrete Crib Walls

TABLE 1. Solution	oil Retent ATION	ion Cri	b Wall	(open and close	d face)	SECTION	SN					
					0	BSERVED D	ISTRESS					
COUNTY	RT	МР	DIR	1	2	3	4	5	6	7	8	9
Lee (open) fig. 6,10	US 52	n/a	n/a	7b	42	3b,7b	7b	7b	3b,7b	42	4b,7b	36,76
Powell (open) fig. n/a	KY 33	n/a	n/a	5a,4a,10b	10b	401	106	901	901	401	901	10b
Kenton (cld) fig. 4,5,8,11,15	175/71	187.7	z	4b,5b,6a	4b,6c,8b	4b,6c	4b	4b,8b	4b,6c	4b,7b	4b,7b	4b,7c
					Ι	DISTRESS OF	SERVED AV	.C. (%)				
				3a=33%	4b=33%	3b=33%	4b=33%	4b=33%	3b=33%	4b=33%	4b=66%	3b=33%
				4a=33%	6c=33%	4b=33%	7b=33%	5a=33%	4b=33%	7b=66%	7b=66%	4b=33%
				4b=33%	7b=33%	6c=33%	10b=33%	7b=33%	6c=33%	10b=33%	10b=33%	7b=33%
				5b=33%	8b=33%	7b=33%		8b=33%	7b=33%			7c=33%
				6a=33%	10b=33%	10b=33%		10b=33%	10b=33%			10b=33%
				7b=33%								
				10b=33%								
<b>FABLE 2.</b> Sc	oil Retenti	on Crib	/Wing	Wall (clos-	ed face)							
LOC	A'TION					SECTION	SA					
					0	BSERVED D	ISTRESS					
COUNTY	RT	МР	DIR	I	2	3	4	5	6	7	8	9
Kenton (cld) fig. 7,9,16	KY 8	7.57	w	4b,5b	106	60		901	5b,6c		9b,10b	Sc,6c,9b
					1	DISTRESS OB	SERVED AV	G. (%)				
				4b=100%	10b=100%	6c=100%		10b=100%	5b=100%		9b=100%	5b=100%
				5b=100%					6c=100%		10b=100%	6c=100%
												9b=100%
DISTRESS VERT, DISP. HORIZ, DISP.	<ol> <li>BACKWARL</li> <li>CRACKED C</li> <li>CRACKED C</li> </ol>	S DISP. RIB MEMBEB REMBERSIDE	ts TO EROSI	<ol> <li>WASH OU</li> <li>MIGRATION</li> <li>NI BULGING</li> </ol>	JT ÓF INTERIOR FIL N OF BACKFILL	L 9. SPALLING 10. VEGETATION		NSIGNI V = SIL	ECANCE OF DISTRI IGHT B = MODER/ C = SEVERF	SIL		

Ę 111. : ¢ £ ī ç Ľ

A - 8 Materials from the City of Seattle (2014)



3/27/2017 11:26

Street A	opurtenance Asset Type
	Asset ID RTW-018
Asset D	escription 3RD AVE W BETWEEN FLORENTIA AND DRAVUS
	Compkey 512661
Curr	ent Status INSVC
ocation	
Locatio	n Information
	City Sector W
IVe	District MGNL-QA
Maintena	nce District
Locatio	n Measurements
	On Street
Low	iross Street irb Space #
	Feet To 0.0
Meas	ured From? no
nigh (	Feet To 0.0
Dir	From Cross
Meas	Street ured From? no
Measure	ment Origin
3	ie of Street
Addres	s
Stree	Appurtenance Address
	Address
	Street #
	Street Name
	Suffix Back Dia
5	Post Dir Ibdesignation
	Cross Street
	Cross Street ity. State ZIP
ll ĭ	City Sector
	leighborhood
Structur	<u>al</u>
(Tab Not L	aded)
Associa	ed
(Tab Not Lo	aded)
Attachm	ents
(Tab Not Lo	aded)
Comme	nts
(Tab Not Lo	aded)

Associated Assets (Tab Not Loaded)
Perf Indicators (Tab Not Loaded)
Life History (Tab Not Loaded)
Retaining Wall Info
Retaining Wall Information
Built By         Built to City         Standard         Year Rehabilitated         O         Year Replaced         Structure Integrity         Status         Monitor         Monitor         Detour Length         0.00         Summit Activity ID         Summit SCAT         Miles         Installation and Maintenance Agreement         Field Book No.         LID         3626         Historic Register         SU Permit No.         Retaining Wall         Concrete Gravity         Retaining Wall         CRGRAV
Wall Length 275.00 Feet
Components Wall Min. Height 3.00 Wall Max. Height 12.00 Feet Post Material Type Rail Material Type Rail Length 0.00 Feet Feet Feet
Features / Functions Retaining Wall Function Affected Features Affected Private Property Affected Street Fall Protection Weep Hole Subsurface Drain Tilt Monitoring Reference
Record Modified Date Modified By Added By RS_INV_2013 AM Added Date 12/27/1990 00:00

Related Assets  ID 1  ID 2  ID 3  ID 4  ID 5  Status and Condition  Status and Condition  Current Status Date 12/21/2009  Asset Condition  Condition  Assest Condition  Condition  Assest Condit
Asset Disposition Asset Disposition Disposal Reason Disposal Date Disposal Cost 0.00 Salvage Date Salvage Value 0.00
Asset Ownership Ownership Date Ownership Assumed Maintained By Installation and Maintenance Responsibility/Agreement Maintenance Financial Responsibility
Council Districts Primary Council District Secondary Council District Override no Override Comments
Financial
Financial Information Replacement Value 0.00 Replacement Value Date Lifecycle Cost 0.00 Planned Maintenance Cost 0.00
Acquisition/Improvement Grid Acquisition/Improvement Comments 1 (No Data)

Work Order History

Work Order History Information

A - 9 Materials from the Ohio DOT (2007)

	MSE WALL INSPECTION CHECKLIST	
District	Date Inspected Name of Inspector	
Is MSE wall at a bridge? (Y/N)	County Boute Section I/B BA/EA End Sec	ons are d page.
Yes No N/A	Joints	Measure- ment
ΟΥΟΝΟΧ	<ol> <li>Is sand or gravel coming out of joints or are there piles of sand or gravel at the base of the wall? (Photos 2 &amp; 3)</li> </ol>	
ΟΥΟΝΟΧ	2. Is sand or gravel visible in the horizontal joints? (Photo 4)	
ΟΥΟΝΟΧ	3. Are the joints wide enough to see the sand, gravel or fabric behind the panels when looking perpendicular to the wall face using a flashlight? (Photo 5) If yes, record the approximate maximum joint width, in inches.	
ΟΥΟΝΟΧ	<ol> <li>If fabric is visible in the joints, is it torn?</li> <li>IMPORTANT - DO NOT POKE OR CUT THE FABRIC.</li> </ol>	
$\bigcirc$ Y $\bigcirc$ N $\bigcirc$ X	<ol><li>Do the joints have a nonuniform size, or are some joints noticeably wider than others? (Photo 6)</li></ol>	
ΟΥΟΝΟΧ	<ol><li>Are the panels offset at the joints either in or out of the wall? (Photo 7) If yes, record the approximate maximum offset.</li></ol>	
ΟΥΟΝΟΧ	7. Is there vegetation growing in the joints? (Photo 8)	
	Wall Facing	
O Y O N O X	If yes, record the approximate number of panels that are cracked.	
$\bigcirc$ Y $\bigcirc$ N $\bigcirc$ X	9. Is the face of the wall bowed or bulged? (Photo 11)	
	Drainage	
$\bigcirc$ Y $\bigcirc$ N $\bigcirc$ X	10. Are there any signs of water flow along the base of the wall?	
ΟΥΟΝΟΧ	11. Is there erosion of the embankment at the base of the wall? (Photo 12)	
$\bigcirc$ Y $\bigcirc$ N $\bigcirc$ X	<ol> <li>If there is erosion, is the leveling pad exposed at the base of the wall? (Photo 13)</li> </ol>	
0 Y 0 N 0 X	13. Are the catch basins or the catch basin outlets near the wall blocked? (Photo 14)	
O Y $O$ N $O$ X	14. Is the roadway drainage system above the wall malfunctioning?	
ΟΥΟΝΟΧ	15. Does water at the top of the wall collect behind the concrete coping?	
	Top of Wall	
ΟΥΟΝΟΧ	ro. Is there settlement at the top of the wall?	
$\bigcirc$ Y $\bigcirc$ N $\bigcirc$ X	<ol> <li>Are there any open cracks in the concrete coping (not hairline cracks)? If yes, record the approximate maximum crack width.</li> </ol>	
ΟΥΟΝΟΧ	<ol> <li>Have the construction joints in the concrete coping opened up? (Photo 6) If yes, record the approximate maximum joint width.</li> </ol>	
$\bigcirc$ Y $\bigcirc$ N $\bigcirc$ X	<ol> <li>Is there a gap larger than 1 inch between the approach slab and the approach pavement? (Photo 15) If yes, record the approx. max. gap size.</li> </ol>	
ΟΥΟΝΟΧ	<ol> <li>At abutments, has the joint between the wall coping and the abutment opened up more than two inches? (Photo 16)</li> </ol>	

	Date	Inspected		Name of I	nspecto	r			
s MSE wall at	7 [				] Г				Instructions are
	L	County	Route	Section	J L	L/R	RA/FA	End Sec.	on the 2nd page.
Instructions									
Instructions 1. Enter the Distric	t, date i	nspected, a	and the na	ame of the in	nspecto	r.			
Instructions 1. Enter the Distric 2. Identify the MSE For MSE walks	ot, date in E wall loc	nspected, a	and the na	ame of the in	nspector	r.	E For tu	un bridaes	with
Instructions 1. Enter the Distric 2. Identify the MSE For MSE walls a separate 1	et, date in E wall loc MSE wa MSE wa	nspected, a cation. is, use the s lls, identify	and the na section fo if the MSI	ame of the in r the bridge E wall is on	in the C	r. C/R/S or rig	S. For tw	vin bridges atify if the M	with ISE wall is
Instructions 1. Enter the Distric 2. Identify the MSE For MSE walls a separate M at the rear For MSE walls a Lentify if	et, date in E wall loc at bridge MSE wa r or forw away fro the wall	nspected, a cation. is, use the s lls, identify ard abutme m bridges,	and the na section fo if the MSI nt. use the s	ame of the in r the bridge E wall is on ection at the	in the C the left e beginn	r. C/R/S or rig	5. For tw ght. Iden	vin bridges tify if the M all in the C/	with ISE wall is R/S.
Instructions 1. Enter the Distric 2. Identify the MSE For MSE walls a separate M at the rear For MSE walls a Identify if t MSE wall.	et, date in E wall loc at bridge MSE wa r or forw away fro the wall	nspected, a is, use the s Ils, identify ard abutme m bridges, is on the let	ind the na section fo if the MSI int. use the s ft or right	ame of the in r the bridge E wall is on ection at the of the roads	in the C the left e beginr way. Fil	r. C/R/S or rig ning (	<ol> <li>For tw ght. Iden of the washe section</li> </ol>	vin bridges tify if the M all in the C/I n for the e	with ISE wall is R/S. nd of the
Instructions 1. Enter the Distric 2. Identify the MSE For MSE walls a separate M at the rear For MSE walls a Identify if t MSE wall. 3. Inspect the wall	et, date in E wall loc at bridge MSE wa r or forw away fro the wall and ans	nspected, a cation. es, use the s lls, identify ard abutme m bridges, is on the le	and the na section fo if the MSI ont. use the s ft or right question:	the bridge wall is on ection at the of the roads s. Refer to	in the C the left e beginn way. Fil the exa	r. C/R/S or rig ning ( Il in the mple	<ol> <li>For tw pht. Iden of the wa he section</li> </ol>	vin bridges tify if the M all in the C/ In for the e to identify t	with ISE wall is R/S. nd of the
Instructions 1. Enter the Distric 2. Identify the MSE For MSE walls a separate M at the rear For MSE walls a Identify if t MSE wall 3. Inspect the wall an MSE wall and indicate a poten	et, date in E wall loc at bridge MSE wa r or forw away fro the wall and ans d for gui tial prob	nspected, a cation. is, use the s ard abutme m bridges, is on the ler swer the 20 dance as to clem with th	and the national section for the MSI ont. use the s ft or right question: p what to b	ame of the in r the bridge E wall is on ection at the of the roads s. Refer to look for. A chen require	in the C the left e beginn way. Fil the exai "Yes" ar ed, take	r. C/R/S or rig ning o l in th mple an a	<ol> <li>For tw ght. Iden of the wa he section photos or to any pproxim.</li> </ol>	rin bridges atify if the M all in the C/ In for the e to identify t of the ques ate measu	with ISE wall is R/S. nd of the he parts of stions may rement and
Instructions 1. Enter the Distric 2. Identify the MSE For MSE walls a separate M at the rear For MSE walls a Identify if t MSE wall. 3. Inspect the wall an MSE wall and indicate a poten record it in the b	et, date in E wall loc at bridge MSE wa r or forw away fro the wall and ans d for gui tial prob box to th	nspected, a cation. is, use the s lls, identify ard abutme m bridges, is on the le swer the 20 dance as to dance as to lem with the	and the na section fo if the MSI ont. use the s ft or right question: o what to e wall. W e questio	ame of the in r the bridge E wall is on ection at the of the roady s. Refer to look for. A /hen require n.	in the C the left way. Fil the exat "Yes" ar ed, take	r. C/R/S or rig Il in th mple an a	5. For tw ht. Iden of the wa he section a photos or to any pproximation	vin bridges tify if the M all in the C/ on for the e to identify t of the ques ate measur	with ISE wall is R/S. nd of the the parts of stions may rement and
Instructions 1. Enter the Distric 2. Identify the MSE For MSE walls a separate M at the rear For MSE walls a Identify if t MSE wall. 3. Inspect the wall an MSE wall and indicate a poten record it in the b 4. Take photos sho the entire length	et, date in E wall loc MSE wa r or forw away fro the wall and ans d for gui tial prob box to the owing the o of the w	nspected, a cation. is, use the s sls, identify ard abutme m bridges, is on the le swer the 20 dance as to dance as to lem with th e right of th we entire hei wall. For wa	and the na section fo if the MSI out on the s ft or right o what to e wall. W e question ight of the ills at brid	ame of the in r the bridge E wall is on ection at the of the roads s. Refer to look for. A /hen require n. e wall. For l ges, take of	in the C the left e beginn way. Fil the exa "Yes" ar ed, take ong wall ne photo	r. C/R/S ning ( Il in the nswe an a ls, ta ls, ta	5. For tw ght. Iden of the wa he section photos proxim ke multij each sec	vin bridges ntify if the M all in the C/ on for the e to identify t of the ques ate measur ple photos tion of wall	with ISE wall is R/S. nd of the the parts of stions may rement and to show
Instructions 1. Enter the District 2. Identify the MSE For MSE walls a separate M at the rear For MSE walls a Identify if t MSE wall. 3. Inspect the wall an MSE wall and indicate a poten record it in the b 4. Take photos sho the entire length 5. For each "Yes" a understand the	et, date in E wall loc at bridge MSE wa r or forw away fro the wall and ans d for gui tial prob pox to the owing the o of the w answer i reason f	nspected, a cation. is, use the s lls, identify ard abutme m bridges, is on the le swer the 20 dance as to dance as to dem with the e right of th is e entire hei wall. For wa to a questic for the "Yes	and the national section for if the MSI ont. use the soft or right questions of the wall. We e question ight of the ills at brid on, take a " answer.	ame of the in r the bridge E wall is on ection at the of the roads s. Refer to look for. A then require n. e wall. For I ges, take of photo that s	in the C the left e beginn way. Fil the exat "Yes" ar ed, take ong wall ne photo shows th	r. C/R/S or rig Il in the nswe an a Is, ta Is, ta o of e he re	5. For tw pht. Iden of the wa he section photos r to any pproxim- ke multip each sec elevant ite	vin bridges ntify if the M all in the C/I n for the e to identify t of the ques ate measu ple photos tion of wall em in suffic	with ISE wall is R/S. nd of the the parts of stions may rement and to show cient detail to

A - 10 Materials from the Utah DOT (2009)

	STA	ге о	F UTA	H MSE V	VALL INSPECT	ION FORM	M	
			Compiled As P	art of Research By	The Utah Department of Transport	ation		
Instructions:					·····			
1-Fill out required sections for MSE Wa	all Inspector and Wall	Characteris	tics.					
2-Inspect the wall using the attached for or bridge number, nature of problem, da appropriate number. Note should be tak construction note should be taken in the	rm. Questions that re- ate, photo number for ten by the inspector the space provided for de	quire a 'Yes' wall, and a s at often ano rawings.	answer should be size reference, which malies are due to co	documented by noting t ch should be indicated i onstruction and should b	he extent of the problem in the right most n the photo (white board/paper). Photos t be distinguished from those that are a resu	column and photo documer taken should be placed on th alt of post-construction. If it	tation. Photo o e Top View lay is observable th	Jocumentation should consist of wall /out and indicated with the at they existed at the time of
3- Shoot digital photos of the entire wal	1. This may require th	ne use of a v	ariety of shots and	angles on each wall to c	over the wall in its entirety.	· · ·		
4- Indicate Layout of MSE Wall in resp Coordinates of Site of Interest in space	ect to major intersecti provided below	ons, roadwa	ys, potential hazard	ds, irrigation, vegetation	, locations of conditions for which 'Yes' v	was marked, etc. in space pro	ovided below. A	Also Indicate approximate GPS
T C D C			1	Inspector	· Information			
Inspection Date			-		Names Of Inspectors			
Region				Iden	tifying Road/Intersection			
			MSE	WALL CH	IARACTERISTICS			
MSE Wall at Bridge	Y	Ν	Bridge Number i	f applicable:		Wall Number	r	
Surrounding Structures					1	Maximum Height of Wall (ft)		
Distance to Each Structure					One Stage, Two Stage or	Block Wall		
State Route Number					Estimated M	ax Length of Wall Abutment:		
Approximate Mile Marker					Max Slo	pe of Ground in front of wall:		
GPS Datum	WGS/84,	NA	D/83, or	NAD/27	Max Height of wall burial line ab	ove surrounding level ground:		
MSE Wall GPS Coordinates (Location of Measurement shown on plan view)					Please draw rough lay	out of panel with approximate	dimensions in sp	ace provided below:
Measurement shown on plan viewy								
If known, Panel or System Manufacturer								
Are there coupons availabe for this wall? If so, how many?								
Summary of Key	v Observa	tions		and and the for the second second second				

**Plan/Drainage View:** 

1	
	1
••I	
<u>80</u>	N N N N N N N N N N N N N N N N N N N
1:51	
50	20
S 20	28
0000	008
Tross (	ross

				SPECIFIC WALL CI	HARACTERISTICS											
				DRAIN	NAGE											
	Required	Tools:		Nylon Mallet-Water Bottle-Camera-Tape Measure												
Yes	No	N/A	UKN	Drainage	Measurement/Explanation (Wayneed)			Perce	entage of v	vall affecto	ed/Extent	of Proble	m/Photo N	umber		
*	z	N/A	UKN	there an active water source near the toe of the wall (is the wall near a body of water with scour potential) (Percentage of wall affected?)		0//0	1%	5%	10%	25%	50%	75%	%06	95%	100%	
Y	N	N/A	UKN	applicable, are the catch basins at the base of the wall blocked?/Percentage of wall affectod?)		o/No	1%	5%	10%	25%	50%	75%	%06	95%	100%	
7	z	N/A	UKN	te these exiverts protruding through the wall? (Percentage of wall affected?)		0V/0	1%	5%	10%	25%	50%	75%	%06	95%	100%	
*	z	N/A	UKN	te there vertical drains that travel through the backfill? (Percentage of wall affected?)		o/No	1%	5%	10%	25%	50%	75%	%06	95%	100%	
×	z	N/A	UKN	there erosion at the base of the wall or leveling pat? (Percentage of wall affected?)		o/No	1%	5%	10%	25%	50%	75%	%06	95%	100%	
-	z	N/A	UKN	there erosion along the wing walk? (Percentage of wall affected?)		0//0	1%	5%	10%	25%	50%	75%	%06	95%	100%	
~	z	N/A	UKN	te there any signs of water flow slong the base of the wall? (Percentage of wall affected?)		o/No	1%	5%	10%	25%	50%	75%	%06	%56	100%	
7	z	NA	UKN	there less than 14 feet between irrigation sprinklers and wall? (Percentage of wall affected?)		0/No	1%	5%	10%	25%	50%	75%	%06	95%	100%	
*	z	N/A	UKN	oos the backfill or joint fabric appear to be saturated? (Percentage of wall affected?)		0/No	1%	5%	10%	25%	50%	75%	%06	95%	100%	
*	z	N/A	UKN	there vegetation growing in panel joints? (Percentage of wall affected?)		o/No	1%	5%	10%	25%	50%	75%	%06	95%	100%	
*	z	N/A	UKN	re the deck drains and outlets at the top of the wall blocked? (Percentage of wall affected?)		0/No	1%	5%	10%	25%	50%	75%	90%	95%	100%	
>	z	N/N	UKN	an water enter the wall between coping and slab (i.e. Drain appropriately)? (Percentage of wall affected?)		0/No	1%	5%	10%	25%	50%	75%	%06	95%	100%	
>	z	NA	UKN	there evidence at discharge point of fill washing through drain pipes? (Percentage of wall affected?)		0///0	1%	5%	10%	25%	50%	75%	90%	95%	100%	
S. S. S. S.																
				MSEWa	all Joints											
	Required	d Tools:		ong Level-String-Camora-Tape Measure												
Yes	ů	NIA	UKN	Johts	Measurement/Explanation of applicated			Perce	entage of v	wall affect	ed/Extent	of Proble	m/Photo N	umber		
*	z	N/A	UKN	backfill coming out of joints or are there piles of backfill at the base of the wall? (Percentage of wall affected?)		0//0	1%	5%	10%	25%	50%	75%	%06	95%	100%	
>	z	N/A	UKN	<ul> <li>in the second sec</li></ul>		0N/0	1%	5%	10%	25%	50%	75%	%06	95%	100%	
*	z	N/A	UKN	actuatory point where are revenue. exposed backfill visible in the horizontal joints? (Percentage of wall affected?)		0/No	1%	5%	10%	25%	50%	75%	%06	95%	100%	
>	z	N/A	UKN	re there visible tears in the fabric? (Percentage of wall affocted?)		0///0	1%	5%	10%	25%	50%	75%	%06	95%	100%	
~	x	NA	UKN	there evidence of backfill or water leaking through tears in fabric behind panels? (Do not induce additional damage to fabric) (Percentage of wall		0N/0	1%	5%	10%	25%	50%	75%	%06	95%	100%	
*	z	N/A	UKN	usereury) to the joints have a non-uniform horizontal specing/size? (i.e., Are some horizontal joints larger/smaller than others?) (Percentage of wall affected?)		0///0	1%	5%	10%	25%	50%	75%	%06	95%	100%	
7	z	N/A	V UKN	to the joints have a non-uniform vertical spacing/size? (i.e. Are some horizontal joints larger/smaller than others?) (Percentage of wall affected?)		0///0	1%	5%	10%	25%	50%	75%	%06	95%	100%	
Y	z	NIA	UKN	ce the panels offset at the joints either in or out of the wall? (Percentage of wall affected?) if yes, record the approximate maximum offset		o/No	1%	5%	10%	25%	50%	75%	%06	95%	100%	
>	z	NIA	V UKN	Nots the fabric appear twittle, or appear as if it has undergone excessive UV exposure? (Percentage of wall affected?)		0//No	1%	5%	10%	25%	50%	75%	%06	95%	100%	
		2														
				WALLI	FACING											
	Required	:d Tools:		.org Level-String-Camera-Crack Gauge-Tape Measure												
Yus	No	N/A	V UKN	Wall Facing	Mcasurement/Explanation (// applicate)			Pero	entage of	wall affect	ed/Extent	of Proble	em/Photo N	umber		
>	z	NVA	V UKN	Vere the panels built using "Tilt-Up" construction? (Percentage of wall affected?) is there excessive crecking in the panels?		0//0	1%	5%	10%	25%	50%	75%	%06	95%	100%	
>	z	NIA	V UKN	s there excessive stacking in panels? (Percentage of wall affected?)		0/No	1%	5%	10%	25%	50%	75%	%06	95%	100%	
×	z	N/A	V UKN	ve there cracks that continue vertically through adjacent panels? (Percentage of wall affected?)		0N/0	1%	5%	10%	25%	50%	75%	90%	95%	100%	
*	z	V/N	V UKN	te there crecks that continue horizontally through adjacent panalis? (Percentage of wall affected?)		0//0	1%	5%	10%	25%	50%	75%	%06	95%	100%	
*	z	NIA	A UKN	te the panel concess making contact with each other? (Percentage of wall affected?)		0/No	1%	5%	10%	25%	50%	75%	%06	95%	100%	
>	z	V/N	1 UKN	ter the panel corners "popped-off" or chipped from contact with adjacent panel(s)? (Percentage of wall affected?)		0//0	1%	5%	10%	25%a	50%	75%	%06	95%	100%	
*	z	N/A	A UKN	boss erzek spacing suggest differential settlement? (Percentage of wall affected?)		o/No	1%	5%	10%	25%	50%	75%	%06	95%	100%	

																Γ
*	z	V/V	A UK	Does the overlying coping exhibit vertical offset? (Percentage of wall affected?)		0N0	1%	5%	10%	25%	50%	75%	%06	95%	100%	
λ	N	N/A	A UK	te the coping and parapets loose or deteching? (Percentage of wall affectod?)		0/No	1%	5%	10%	25%	50%	75%	%06	95%	100%	Т
¥	x	N/A	A UK	dre the panels in danger of falling off? (Percentage of wall affected?)		0/No	1%	5%	10%	25%	50%	75%	%06	95%	100%	
Y	z	NIA	A UK	the panels exhibit builging? (Percentage of wall affected?) If so, record maximum deformation from accessible coping to leveling pad.		0/No	1%	5%	10%	25%	50%	75%	%06	95%	100%	
۲	z	NIA	A UK	s there tupping at the top or bottom of the wall? (Percentage of wall affocted?) (Record maximum degree of tipping from azimuth using vertical level and fifteded area).		o/No	1%	5%	10%	25%	50%	75%	%06	95%	100%	1
Y	z	NIA	A UK	s there excessive degradation of wall panels? (Percentage of wall affected?)		0///0	1%	5%	10%	25%	50%	75%	%06	95%	100%	
		1														
		100 M		MSE TOP (	OF WALL											
	Requires	cd Tools:		Long Level-Crack Gauge-Camera-Tape Measure												
Yes	No	NIA	A UK	Top Of Wall	Measurement/Explanation (1/ 400 <sup>400</sup>			Percent	lage of wa	ll affected	Extent of	Problem/F	hoto Num	ber		
X	z	N/A	A UK	s there evidence of settlement at the top of the wall (pavement cracking, etc)? (Percentage of wall affected?)		0/No	1%	5%	10%	25%	50%	75%	%06	95%	100%	
×	z	N/A	A UK	the three arry non-haid line cracks in the concrete coping? (Percentage of wall affocted?) If yes, record the approximate maximum crack width.		0/No	1%	5%	10%	25%	50%	75%	%06	95%	100%	
Y	z	NA	A UK	fare the construction joints in the constrcting coping opened up? (Percentage of well affected?) If yes, record the maximum joint width.		0/N/0	1%	5%	10%	25%	\$0%	75%	%06	95%	100%	
Y	z	N/A	A UK	s there a large gap between the approach slab and parvement? (Percentage of well affector?) Often this produces a bumping sensation as the overpass is researd. Record: the annewintate maximum and size.		0/No	1%	5%	10%	25%	\$0%	75%	%06	95%	100%	
×	z		M	account record or the point between the wall coping and abuttnent opened up significantly/ (Percentage of wall affector?) If so record maximum the behavior of the abuttnents has the joint between the wall coping and abuttnent opened up significantly/ (Percentage of wall affector?) If so record maximum the abuttnent opened up significantly/ (Percentage of wall affector?) If so record maximum the abuttnent opened up significantly/ (Percentage of wall affector?) If so record maximum the abuttnent opened up significantly/ (Percentage of wall affector?) If so record maximum the abuttnent opened up significantly/ (Percentage of wall affector?) If so record maximum the abuttnent opened up significantly/ (Percentage of wall affector?) If so record maximum the abuttnent opened up significantly/ (Percentage of wall affector?) If so record maximum the abuttnent opened up significantly/ (Percentage of wall affector?) If so record maximum the abuttnent opened up significantly/ (Percentage of wall affector?) If so record maximum the abuttnent opened up significantly/ (Percentage of wall affector?) If so record maximum the abuttnent opened up significantly/ (Percentage of wall affector?) If so record maximum the abuttnent opened up significantly/ (Percentage of wall affector?) If so record maximum the abuttnent opened up significantly/ (Percentage of wall affector?) If so record maximum the abuttnent opened up significantly/ (Percentage of wall affector?) If so record maximum the abuttnent opened up significantly/ (Percentage of wall affector?) If so record maximum the abuttnent opened up significantly/ (Percentage of wall affector?) If so record maximum the abuttnent opened up significantly/ (Percentage of wall affector?) If so record maximum the abuttnent opened up significantly/ (Percentage of wall affector?) If so record maximum the abuttnent opened up significantly/ (Percentage of wall affector?) If so record maximum the abuttnent opened up significantly/ (Percentage of wall affector?) If so record maximum the abuttnent		0/No	1%	5%	10%	25%	\$0%	75%	%06	95%	100%	
Y	z		A A	s the coping/wall pulling away from pavement and roadway section? (Percertage of wall affected?) Please record maximum displacement for wall.		0///0	1%	5%	10%	25%	50%	75%a	%06	95%	100%	
																- 110 Å
				FOUNDATION CONDITIONS.	AND EXTERNAL STABILITY											
	Require	ed Tools:		Shovel, GEO-Probe-Tape Measure												
Yes	No.	NA	A UK	Foundation Conditions & External Stability	Measurement/Explanation of opplicated			Percent	tage of wa	II affected	/Extent of	Problem/	hoto Num	ber		
¥	z	NA	A UK	What is the location depth of Loveling part (Percentage of wall affected?) Pound Geo-Probe into soil located 2 inches from wall to a maximum depth of 24 achies 724 inches is the minimum death for MSE Wall).		0///0	1%	5%	10%	25%	50%	75%	%06	95%	100%	
×	z	NA	A UK	to the leveling and exposed?		0/No	1%	5%	10%	25%	50%	75%	%06	95%	100%	
*	z	N/A	A UK	s there cracking in the leveling part? (Percentage of wall affected?) If so, record maximum crack size with page.		0/No	1%	5%	10%	25%	50%	75%	%06	95%	100%	
7	z	NA	A UK	is there a four foot bench (level slope) along the base of the wall before the slope changer? (Percentage of wall affected?) If not record width of bench?		0/No	1%	5%	10%	25%	50%	75%	%06	95%	100%	
×	z	N/N	A UK	is there a slope storper than H: 1.5 to V:1 behind the wall? (Percentage of wall affected?) Please record slope.		0/No	1%	5%	10%	25%	50%	75%	%06	95%	100%	
×	z	N/A	A UK	is there a slope greater than Y: 1.5 to H:1 belowin front of the wall? (Percentage of wall affected?) Please record slope.		0/No	1%	5%	10%	25%	50%	75%	%06	95%	100%	
3																
		1		CORR	NOISO											
	Require	ed Tools:		Nylon Mallet-Camera-Zip Lock Bag-Trowel-Tape Measure												Τ
Yes	No	N/N	A UK	Corresion	Measurement/Explanation of opplicated			Percent	tage of wa.	II affected	/Extent of	Problem/	hoto Num	ber		
Y	z	N/A	A UK	is there excessive corrosion on guardraits or other exposed metal that might indicate corrosive conditions? (Percentage of wall affected?)		0//0	1%	5%	10%	25%	50%	75%	%06	95%	100%	
*	z	V/N	A UK	Are there significant rust stains on face of the wall? Along joints? (Percentage of wall affected?)		0/No	1%	5%	10%	25%	50%	75%	%06	95%	100%	
Y	Z	V/N	A UK	Are any internal straps exposed? (Percentage of wall affected?)		0/No	1%	5%	10%	25%	50%	75%	%06	95%	100%	
Y	z	N/A	A UK	f exposed, does there appear to be corrosition on straps? (Percentage of wall affected?)		0/No	1%	5%	10%	25%	50%	75%	%06	95%	100%	
*	z	N/A	A UK	Was a resistivity sample taken of soil? If so, please indicate depth taken in inches.		o/No	1%	5%	10%	25%	50%	75%	%06	95%	100%	Т
*	z	N/A	A UK	is these indication of rebor corrosion (i.e., swelling bars, rust, exposed metal inside epoxy coaling.)? (Percentage of wall affected?)		0///0	1%	5%	10%	25%	50%	75%	%06	95%	100%	-1
*	z	N/N	A UK	is there evidence of Test Wircs or Coupons included in the wall? If so, please indicate the possible number of Test wires available?		0///0	1%	5%	10%	25%	50%	75%	%06	95%	100%	
		and the second s	and the					2								
				IMPACT AND	COLLISION											
	Require	ed Tools:		Camera												

Yes	No	N/A	Impact and Collision	Measurement/Explanation (Nappleship)			Ferce	ntage of w	vall affect	cd/Extent	of Proble	m/Photo N	umber		
×	z	N/A 1	Are guardraits/will protections in place at the base of the wall (to protect it from potential traffic hazards)? (Percentage of wall affected?)		0//0	1%	5%	10%	25%	50%	75%	%06	95%	100%	
¥	z	N/A 1	Does it uppear that the wall has been involved in an accident (replaced panel, recent dings in the wall)? (Percentage of wall affected?)		0///0	1%	5%	10%	25%	50%	75%	%06	95%	100%	
*	z	N/A 1	Does it appear the walls functionality and integrity has been compromised by a collision or accident? (Percentage of well affected?)		0N/0	1%	5%	10%	25%	50%	75%	%06	95%	100%	
			MISCELI	CLANEOUS											
	Required 1	Tools:	Available drawings-Camera-Tape Measure												
Yes	No	N/A	Miscellaneous	Measurement/Explanation @ray@conted			Perce	ntage of w	vall affect	ed/Extent	of Proble	m/Photo N	umber		
×	z	N/A	Are there acute well angles (<50)? (Percentage of wall affected?)		0///0	1%	5%	10%	25%	50%	75%	%06	%56	100%	
Y	z	N/A	1. Are there available drawings for the wall? Please indicate type (Situation and Layout, Design, As Built, etc.) (Percentage of wall affected?)		0N/0	1%	5%	10%	25%	50%	75%	%06	95%	100%	
×	z	N/A	1 is the layout in general accordance with drawings? (Percentage of wall affected?)		0///0	1%	5%	10%	25%	50%	75%	%06	95%	100%	
x	z	N/A	Are the panels CIP (Cast in Place)? (Percentage of wall affected?)		0N/0	1%	5%	10%	25%	50%	75%	%06	95%	100%	
×	z	N/A	Was Geofoam used in the construction of the wall? (Percentage of wall affected?)		0N/0	1%	5%	10%	25%	50%	75%	%06	95%	%001	
×	z	N/A	Ace there any structures on or near wall that were not included in initial drawings? (Percentage of wall afflected?)		oN/0	1%	5%	10%	25%	50%	75%	%06	95%	100%	
x	z	N/A	f. Are there any irrigation, utilities, or intrusions that are not part of the initial drawings? (Percentage of wall affected?)		0//0	1%	5%	10%	25%	50%	75%	%06	95%	100%	
•	z	N/A	Have there been any excavations or evidence of excavations near the wall? (Fercentage of wall affected?)		o/No	1%	5%	10%	25%	50%	75%	%06	95%	100%	
*	z	N/A	i Have local property owners changed the dynamics of the wall (additional structures, irrigation, vegetation, etc.)? (Percentage of wall affected?)		0N/0	l%⊓	5%	10%	25%	50%	75%	%06	95%	100%	
x	z	N/A	Are there piles or other tridge support systems located in the wall (bridge abutanent)? (Percentage of wall affected?)		o/No	1%	5%	10%	25%	50%	75%	%06	95%	100%	

# B. MOVEMENT OF THE M-10 WALL PANEL DURING JANUARY 2020

# **B.1. Introduction**

In early January, during a period of high precipitation and cold temperatures, the M-10 wall panel instrumented in this study was reported to have failed. On January 14, 2020, the Structure Management Section (SMS) of Michigan Department of Transportation (MDOT) issued an emergency contract to closely inspect the M-10 retaining wall panel located at Thatcher Avenue (on the South Bound (SB)) along M-10 corridor in Detroit, MI. The wall section that was suspected of failing was identical to where the long-term wireless monitoring system was deployed in November 2018 offering the unprecedented opportunity to compare the wall response measured with the field observations. It should also be noted that the panel suspected of failing in January 2020 is within 1000 feet of the wall panel that failed in 2013 due to corrosion of the panel tieback rod (MDOT, 2013). The visual inspection of the suspected wall panel in January 2020 revealed significant displacement at the top of the wall (Figure B.1). The suspicion of the wall failing was further confirmed from a detailed analysis of the wall response data collected. This Appendix will highlight the data collected to assess the wall performance; the quantitative data is correlated to the visual observations made during inspection in January 2020.

## **B.2.** Visual Inspection of Wall

Upon visual inspection of the wall, two signs of potential wall panel failure were evident in January 2020. First, the concrete parapet at the top of the M-10 wall panel was evidently to have displace. Inspectors measured the movement of the parapet on the suspect failed wall panel was compared to the parapets of the adjacent wall sections. A displacement of 1 inch was measured by inspectors at both ends of the wall panel parapet. Figure B.1 highlights the measured displacement by inspectors using a ruler. These displacements were lower than those observed in 2013 during the failure of another panel along the M-10 freeway (that displacement was in the 3 to 4 inch range). Additional evidence of wall movement was a noticeable crack in the road asphalt surface running parallel to the wall. This crack was suspected to have developed when the wall panel thrusted

forward; the thrusting of the wall forward would place the asphalt in tension resulting in the parallel crack observed (Figure B.2).



(a)

(b)

Figure B.1. Manually measured movement of the M-10 wall panel at the top wall parapet in January 2020.



Figure B.2. Top road surface at the top of the suspected failed M-10 wall panel (January 2020); major tensile crack in asphalt surface evident parallel to wall face.



Figure B.3. Measurement of the M-10 wall panel including early January 2020 with a high precipitation event on January 12, 2020 evident.

## **B.3. Measured Wall Movement**

The top and mid-height tilt of the M-10 panel was continuously measured up to the point when the monitoring system was removed in February 2020. The monitoring system was operational during the time period of possible wall failure with the wall tilt data presented in Figure B.3. There was one major rain event on January 12, 2020 where the city of Detroit had a daily sum precipitation of 2.1 inches. This January 12, 2020 rainstorm was the largest rainfall event observed in the metro region of Michigan during the monitoring study (Weather Underground, 2020). A day after the rainstorm, the movement of the M-10 retaining wall panel caught the attention of MDOT managers who elected to pursue visual inspection and investigation. Prior to the rain event, the instrumented M-10 wall behaved normally as had been observed over the prior year with the top and mid-height tilts at approximately 1.6 degrees and 0.5 degrees, respectively (with positive tilt towards the freeway). However, immediately following the heavy rain, the wall mid-height tilt remained unaltered but the top wall tilt significantly changed to 0.34 degrees. A few days after the rain event, the wall appeared to have returned at the top portion to a tilt of 1.6 degrees. The dramatic change in rotation in the wall during and after the rain event resulted in a permanent



Figure B.4. Top tilt of instrumented M-10 retaining wall panel histogram

deformation of the wall top as observed by inspectors. The dramatic movement of the wall over the course of a few days was likely the culprit of the parallel crack observed in the asphalt road surface.

To provide a sense of the dramatic wall response, a histogram of the top tile of the wall is plotted in Figure B.4. The tilt of the top portion of the M-10 wall panel in January 13, 2020 is shown relative to the variation of wall tilt during the project monitoring period. The normal tilt changes at the top of the wall panel appear to be relatively Gaussian with a mean of approximately 1.57 degree and standard deviation of 0.14 degrees; the tilt observed on January 13, 2020 was approximately 7 deviations away from the distribution mean.

On January 22, 2020, an autonomous UAV flight was executed with MDOT personnel to collect additional aerial imagery of the retaining wall. This flight was conducted using a DJI Phantom 4 Professional UAV equipped with a high-resolution RGB camera. Evenly spaced, overlapping, geotagged imagery was collected of the site using flight capture software. Approximately 2.5 acres were surveyed in a total of 355 images. Using the collected imagery, a 3D model was generated using the Structure-from-Motion (SfM) photogrammetry technique which uses overlapping imagery to compute 3D data using the software package Pix4D Capture. The model achieved a ground sampling distance (spatial resolution) of 0.29 inches per pixel. An snapshot taken from the 3D model mesh can be seen in Figure B.5. Cross sections taken along the retaining wall point cloud model were extracted and exported into AutoCAD to compute the angle at which the wall is leaning at different locations. The yellow lines in Figure B.6 identify the locations of





Figure B.5. (a) 3D point cloud generated by Pix4D from aerial UAV imagery collected on January 22, 2020; (b) cross section of Panel 4.

the cross sections that were analyzed. Panels 1 through 9 span the entire area that was surveyed with the UAV. It should be noted that "Panel 4" corresponds to the instrumented panel. As shown, the panels are indeed tilted with tilts ranging from 1 degree (Panel 9) to 2.5 degrees (Panels 2 and 4). These tilts are larger than that measured by the long-term monitoring system (1.6 degrees as measured at Panel 4 by the monitoring system). These differences are likely due to calibration errors in the UAV photogrammetry method; specifically, there is an absence of a true sense of the horizontal plane associated with the cross-sectional analysis shown in Figure B.5. None the less, the relative tilt measurements shows more distress at Panels 1 through 5 with less at Panel 6 through 9. The measurements also confirm that the measurement of the wall leaning toward the highway is valid.



Figure B.6. (Left) plan view of 3D point cloud model with nine panels of the M-10 retaining wall system identified; (right) measured angle of tilt of the wall at each panel.

# **B.4.** Conclusions

The instrument M-10 wall panel showed significant movement during the large rain event of January 12, 2020. The wall movement is hypothesized to be a direct response to the hydrostatic pressures of the undrained backfill during and immediately following the large rain event. Figure B.7 provides a description of the wall deflection before and during the rain event. Prior to rain, the wall deflected shape with the fully engaged caisson is shown as deflected shape (a-b-c) in Figure B.7. The tilt at the top of the wall can be attributed to the earth pressures on the back of the wall with the caisson stabilizing the wall with minimal tilt below its support point. The mid-height tilt is smaller and nearly stationary under environmental variation indicating the caisson is engaged and restraining the bottom portions of the wall from moving. During the rain event, the build-up of hydrostatic pressures near the top of the wall resulted in the wall moving towards the highway but with the top tilt moving to a more upright position as shown by deflected shape (a-b-d). A few

days after the hydrostatic pressure subsided, the wall returned to a more normal tilt pattern. These observations suggest the tieback is still engaged but is perhaps offering less horizontal support to the wall indicating a tieback at the early stages of breakdown.



Figure B.7. Deformation curves of the drifted M-10 retaining wall system with the tieback caissons shown: (a) hypothesize deflection curve of the wall panel before the major rain event (a-b-c) and during high hydrostatic pressures immediately following the rain (a-b-d); (b) toplevel rotation of retaining wall panel before movement showing a large tilt toward the freeway; (c) top-level rotation of retaining wall panel after the buildup of hydrostatic pressures resulting in displacement and reduced top tilt of the panel. The wall returns to deflected shape (a-b-c) after hydrostatic pressure is relieved.

# MICHIGAN EARTH RETAINING STRUCTURE ELEMENT INSPECTION MANUAL (MIERSEIM)









#### PREFACE

The Michigan Earth Retaining Structure Element Inspection Manual (MiERSEIM) provides condition state information for earth retaining structures that are maintained within the state of Michigan. Earth retaining structures, for the purposes of this manual, are defined as any structure that retains and stabilizes an unstable soil mass by means of lateral support or reinforcement, has an exposed height of 4 ft. or greater, and a vertical, or near vertical face with an angle of inclination greater than 45 degrees from horizontal. Retaining walls are probably the most familiar type of earth retaining structures, but structures such as mechanically stabilized earth, crib walls, rock buttresses, gabion walls, rockeries, etc., that meet the previously stated height and face-angle requirements, are also considered important earth retaining structures. This manual is for use by earth retaining structure owners and inspectors when collecting element-level data for the assessment of the condition of earth retaining structures.

The consistent condition assessment of earth retaining structure elements is the most effective tool for the management of earth retaining structures. The element-level inspection method assesses the entire structure by breaking it down into several elements. Each element of the earth retaining structure is inspected closely and thoroughly. Each element is assigned a condition state based upon its observed and recorded amount of deterioration. Element-level inspection is a quantity based inspection method, and each quantity is described with a condition state to reflect the differing categories of deterioration that may exist on an earth retaining structure element.

The generation of a database for an earth retaining structure management system is a benefit of performing element-level inspections. By developing a database, earth retaining structure deterioration rates can be estimated, based upon the earth retaining structure's material, geographic location, age, usage, type, prior rehabilitation or preventative actions, etc. Software models that utilize the database will allow for comparisons between the effectiveness of preventative and corrective actions, predictions of deterioration, and life cycle cost analysis. Owners of earth retaining structures can then more easily make decisions regarding the prioritization of funds, when (or when not) to take action, and what type of action to take, so as to get the most benefit from the capital that is spent on their earth retaining structures.

#### INTRODUCTION

The purpose of the Michigan Earth Retaining Structure Element Inspection Manual (MiERSEIM) is to provide condition state information for structures inspected within the state of Michigan. These structures typically include highway earth retaining structures with height 4' or greater and angle of face inclination greater than 45 degrees from horizontal. This manual is to be used by earth retaining structure owners and inspectors when collecting element level data. This manual supplements the AASHTO Manual for Bridge Element Inspection, Michigan Bridge Element Inspection Manual and provides further classification of the AASHTO elements and descriptions for Agency Developed Elements.

The element level inspection method breaks the Earth Retaining Structure down into several elements. The element level inspection is a quantity based inspection and each quantity is assigned a Condition State to reflect the differing categories of deterioration that often exist on any Earth Retaining Structure element.

One of the results of performing element level inspections is the generation of a database for an Earth Retaining Structure management system. By developing a database over time, Earth Retaining Structure deterioration rates based upon material, geographic location, age, usage, type of crossing, prior rehabilitation or preventive actions, etc. can be estimated. The software modeling capabilities allow comparisons between the effectiveness of preventive and corrective actions, predictions of estimated future deterioration, and life cycle costs. Decisions can be made regarding prioritizing funds, when (or when not) to take action, and what type of action to take for the maximum benefit of capital spent.

#### **DETAILED ELEMENT DISCRIPTIONS**

This manual describes the individual wall elements evaluated in earth retaining system inspection and management processes.

The first section of the manual contains a detailed description for each element and is broken down into the following subsections:

- Element Number and Name
- Condition State Table to Reference
- Description Detailed identification and classification of the element.
- Quantity Calculation General guidelines on how to collect the quantity of the element and units.
- Element Commentary Additional considerations the inspector is to be aware of during data collection, as appropriate.

The condition state tables are in the second section of the manual. They contain the following information:

- Condition State Definitions Defect descriptions and severity with guidelines for the inspector on defect severity categorization.
- Pictures Example cases of condition states.

Structural elements described are included in the standard set of National Bridge Elements (NBE), Bridge Management Elements (BME), or MDOT Agency Developed Elements (ADE). The elements are organized by element type; Primary Structural Elements and Secondary Non-Structural Elements.

# INDEX

PRIMARY ELEMENTS	
WALL FACING (sg. ft.)	
FOUNDATION (ft.)	
ANCHORS (ea.)	
VERTICAL SUPPORTS (ea.)	5
SECONDARY ELEMENTS	6
VERTICAL COPING / PILASTERS (ft.)	6
HORIZONTAL COPING (ft.)	7
RETAINED MATERIAL (t.).	
JOINTS (ft.)	9
DRAINAGE ELEMENTS (ft.)	
RAILINGS/BARRIERS (ft.)	
ARCHITECTURAL FACING (ea.)	
PROTECTIVE COATING AND SYSTEMS (sq. ft.)	
OVERLAND CONDITIONS (ea.)	
SCOUR PROTECTION	
APPURTENANCES (ea.)	

# INDEX (cont.)

CONDITION STATE TABLES	
CS TABLE 1 – REINFORCED CONCRETE	
CS TABLE 2 – PRESTRESSED CONCRETE	
CS TABLE 3 – STEEL	
CS TABLE 4 – TIMBER	
CS TABLE 5 – MASONRY	
CS TABLE 6 – OTHER MATERIALS	
CS TABLE 7 – ANCHORS	
CS TABLE 8 – JOINTS	
CS TABLE 9 – RETAINED MATERIALS	
CS TABLE 10 – SCOUR PROTECTION	
CS TABLE 11 – OVERLAND CONDITIONS	
CS TABLE 12 – PROTECTIVE SYSTEMS	
CS TABLE 13 – MSE WALLS	
CS TABLE 14 – APPURTENANCES	
CS TABLE 15 – ASPHALT	
#### **PRIMARY ELEMENTS**

The Primary Elements of an Earth Retaining Structure (ERS) describe the structural components that are responsible for retaining earth for transportation purposes.

Included in the Primary Element category are Wall Facings, Foundations, Anchors, and Vertical Supports.

All primary elements can be supplemented with one or more associated protection systems.

# WALL FACING (sq. ft.)

Descr	<b>Description:</b> The plane of the front surface of the exposed portion of the ERS.			
No.	Name	CS Table	Description	
860	MSE Walls	13	Mechanically Stabilized Earth (MSE) walls including the reinforced concrete stub, panels, coping, and drainage system. When piles are exposed use the appropriate material specific element.	
900	Stone Masonry	5	Randomly laid natural stone, or cut stone laid in courses, with or without mortared joints.	
901	Steel	3	Coated or uncoated flat steel plate, cold-formed steel panel sections, or corrugated steel sheets.	
902	Prestressed Concrete	2	Shop-cast Portland cement concrete containing prestressed steel tendons in addition to conventional reinforcing steel.	
903	Reinforced Concrete	1	Site-cast or shop-cast Portland cement concrete containing conventional reinforcing steel.	
904	Timber	4	Pressure treated natural wood, such as sawn lumber or timber, or engineered wood, such as glulam timbers, engineered wood or plywood panels.	
905	Placed Stone	6	Natural or cut stone, laid without mortar (dry-laid). Rockery.	
906	Grouted Stone	6	Natural stone with cement-grouted interstices.	
907	Masonry	5	Concrete masonry units (CMU's) or fired clay brick, laid in courses with mortared joints.	
908	Grouted Block/Brick	6	Rectangular block of concrete or clay laid in courses together jointed with grout.	
909	Shotcrete	6	Pneumatically placed fine-aggregate concrete.	
910	Other	6	Other material earth retaining systems that cannot be classified by any other defined element.	
Quan	tity Calculation: Include the	area of the v	vall from ground elevation to the top of the wall bottom.	
Eleme	lement Commentary:			

# FOUNDATION (ft.)

<b>Descr</b> on th	<b>iption:</b> Those elements that the soil or rock), or indirectly (su	ransfer the ich as throu	loads acting on the ERS to the underlying soil or rock either directly (such as by a spread footing bearing igh piles/caissons, or anchors).
No.	Name	CS Table	Description
220	Reinforced Concrete, Spread Footing	1	Site-cast or shop-cast Portland cement concrete containing conventional reinforcing steel.
225	Pile/ Caisson, Steel	3	Steel H-pile, concrete incased steel H-pile, or concrete filled steel pipe pile.
226	Pile/Caisson, Prestressed Concrete	2	Shop-cast portland cement concrete pile containing prestressed steel tendons.
227	Pile/ Caisson, Reinforced Concrete	1	Site-cast caissons or shop-cast piles consisting of Portland cement concrete with conventional reinforcing steel.
228	Pile/ Caisson, Timber	4	Pressure treated natural wood pile or engineered wood pile.
229	Pile/ Caisson, Other	6	Pile/caissons consisting of a composite of materials, or pile/caissons that cannot be defined as any other pile/caisson element.
231	Pile/ Caisson Cap, Steel	3	Coated or uncoated flat steel plate, cold-formed steel shape sections, or corrugated steel sheets.
233	Pile/ Caisson Cap, Prestressed Concrete	2	Shop-cast Portland cement concrete containing prestressed steel tendons in addition to conventional reinforcing steel.
234	Pile/ Caisson Cap, Reinforced Concrete	1	Site-cast or shop-cast Portland cement concrete containing conventional reinforcing steel.
235	Pile/ Caisson Cap, Timber	4	Pressure treated natural wood, such as sawn lumber or timber, or engineered wood, such as glulam timbers, engineered wood or plywood panels.
236	Pile/ Caisson Cap, Other	6	Pile/caisson cap consisting of a composite of materials, or a pile/caisson cap that cannot be defined as any other pile/caisson cap element.
834	Gabion/ Bin Wall	10	Basket or compartmented rectangular containers made of wire mesh filled with cobbles or other rock.
Quan	tity Calculation: Include the a	rea of the f	ace of the foundation from beginning to end and reference line to reference line.

# ANCHORS (ea.)

**Description:** Those elements that transfer the loads acting on the ERS to the underlying stable soil or rock mass through tension (such as soil or rock anchors, micropiles, or ties attached to deadmen).

No.	Name	CS Table	Description	
911	Wall Anchor, Ground Anchors	7	Also known as earth anchors, or mechanical anchors, either driven by impact into the ground, or run in spirally.	
912	Wall Anchor, Soil Nails	7	Slender reinforcing elements, often either conventional reinforcing bars or proprietary solid or hollow- system bars, drilled or driven into a slope, and then pressure grouted tightly into place.	
913	Wall Anchor, Micropiles	7	Also known as mini-piles, small-diameter (5-12 in.) cast-in-place, reinforced piles that are post-tensioned.	
914	Wall Anchor, Heads	7	The part of the anchor that distributes the anchor loads into that element of the ERS to which it is attached. Often the only visible part of an anchor.	
915	Wall Anchor, Others	7	An anchor that cannot be defined as any other anchor element.	
<b>Quan</b> recor	Quantity Calculation: The quantity for these elements is each. The condition of one element may affect another causing multiple elements to be recorded.			
Eleme	Element Commentary:			

# **VERTICAL SUPPORTS (ea.)**

Description: Those vertical elements that transfer the loads acting on the ERS Wall Facing elements to the ERS Foundation elements, such as soldier pile, counterforts, and buttresses.

No.	Name	CS Table	Description
916	Vertical Support/Column, Steel	3	Hot-rolled steel shapes, including steel H-pile, and/or plates and/or cold-formed steel shapes.
917	Vertical Support/Column, Prestressed Concrete	2	Shop-cast Portland cement concrete containing prestressed steel tendons in addition to conventional reinforcing steel, including precast concrete soldier pile and/or counterforts.
918	Vertical Support/Column, Reinforced Concrete	1	Site-cast or shop-cast Portland cement concrete counterforts and/or buttresses containing conventional reinforcing steel.
919	Vertical Support/Column, Timber	4	Pressure treated natural wood soldier pile or engineered wood soldier pile.
920	Vertical Support/Column, Masonry	5	Buttresses constructed of concrete block, fired clay brick, and/or cut stone, laid in courses with mortared joints.
921	Vertical Support/Column, Other	6	A vertical support element that cannot be defined as any other vertical support element.
<b>Quan</b> record	<b>tity Calculation:</b> The quantity f ded.	or these e	lements is each. The condition of one element may affect another causing multiple elements to be
Eleme	ent Commentary:		

#### SECONDARY ELEMENTS

The Secondary Elements of an Earth Retaining Structure (ERS) describe the non-structural components which condition could affect the Primary Elements long term.

Included in the Secondary Element category are Vertical Coping/Pilasters, Horizontal Coping, Retained Material, Joints, Drainage Elements, Sidewalks, Railings/Barriers, Architectural Facings, Protective Coatings and Systems, and Overland Condition.

### VERTICAL COPING / PILASTERS (ft.)

Description: Relief on the front face of the ERS consisting of a shape of any cross section, that projects out from the plane of the wall facing, and extends vertically upward from the base of the ERS.

No.	Name	CS Table	Description	
922	Vertical Coping/Pilaster, Steel	3	Hot-rolled steel shapes, including steel H-pile, and/or plates and/or cold-formed steel shapes.	
923	Vertical Coping/Pilaster, Reinforced Concrete	1	Site-cast or shop-cast Portland cement concrete containing conventional reinforcing steel.	
924	Vertical Coping/Pilaster, Timber	4	Pressure treated natural wood, such as sawn lumber or timber, and/or engineered wood, such as glulam timbers, engineered wood or plywood panels.	
925	Vertical Coping/Pilaster, Masonry	5	Cut stone, concrete block, or fired clay brick, laid in courses with mortared joints.	
926	Vertical Coping/Pilaster, Other	6	A vertical coping/pilaster element that cannot be defined as any other vertical coping/pilaster element.	
Quantity Calculation: The quantity for these elements is the sum of the vertical heights of the coping.				
Eleme	Element Commentary:			

# HORIZONTAL COPING (ft.)

**Description:** Relief on the front face of the ERS consisting of a shape of any cross section, that projects out from the plane of the wall facing, and extends horizontally along the wall at a constant elevation.

No.	Name	CS Table	Description		
927	Horizontal Coping, Steel	3	Hot-rolled steel shapes and/or plates, and/or cold-formed steel shapes.		
928	Horizontal Coping, Reinforced Concrete	1	Site-cast or shop-cast Portland cement concrete containing conventional reinforcing steel.		
929	Horizontal Coping, Timber	4	Pressure treated natural wood, such as sawn lumber or timber, and/or engineered wood, such as glulam timbers, engineered wood or plywood panels.		
930	Horizontal Coping, Masonry	5	Cut stone, concrete block, or fired clay brick, laid in courses with mortared joints.		
931	Horizontal Coping, Other	6	A horizontal coping element that cannot be defined as any other horizontal coping element.		
231	Sheet Pile Cap, Steel	3	Hot-rolled steel shapes and/or plates, and/or cold-formed steel shapes.		
234	Sheet Pile Cap, Concrete	1	Site-cast or shop-cast Portland cement concrete containing conventional reinforcing steel.		
Quan	tity Calculation: The quantity f	or this ele	ment is the sum of the length of the coping.		
Eleme	lement Commentary:				

# **RETAINED MATERIAL (ft.)**

Descrip	<b>Description:</b> The difference in the elevation of where the extended plane of the surface of the retained material behind the wall intersects with the plane of the surface of the material is front of the wall's intersection with the plane of the wall facing			
No. Name CS Table Description				
932	Retained Material	9	The difference in the elevation of where the extended plane of the surface of the retained material behind the wall intersects with the plane of the wall facing, and the elevation of the plane of the surface of the material in front of the wall's intersection with the plane of the wall facing.	
Quantity Calculation: The quantity for these elements is the sum of the length of the retained material from beginning to end.				
Elemen	t Commentary:			

# JOINTS (ft.)

**Description:** Vertical or horizontal discontinuities in the wall facing, created intentionally to relieve differential movement (such as expansion and contraction joints), as a result of construction procedures (such as construction joints, cold joints, or bolted connections), or as a characteristic of the wall facing material (such as the edges of timbers or precast panels).

No.	Name	CS Table	Description
959	Expansion Joint	8	Joints that are open and not sealed.
306	Joints, Other	8	Joints that cannot be classified using any other defined joint element.
Quantity Calculation: The quantity for this element is the sum of the lengths of all joints.			
Element Commentary:			

# DRAINAGE ELEMENTS (ft.)

**Description**: Elements that collect and convey water around and/or through the ERS (such as porous backfill, weepholes, underdrains and collector pipes, lined drainage swales, etc.).

No.	Name	CS Table	Description	
933	Weep Holes	6	Small diameter conduit cast in the stem of the ERS that dissipates static water pressure behind the wall (backfill side) by allowing water to flow through to the front side of the wall.	
934	Area Drainage	6	Underdrains and collector pipes that convey water from the ERS backfill to an outfall.	
935	Drainage Swale	11	A shallow ditch either immediately in front of the ERS or behind the ERS (backfill side) for the purpose of conveying water away from the ERS.	
936	Drainage Element, Other	6	Drainage elements that cannot be classified using any other defined drainage element.	
Quantity Calculation: The quantity for this element is the sum of the lengths of the element from beginning to end.				
Eleme	Element Commentary:			

# RAILINGS/BARRIERS (ft.)

Descr	<b>Description:</b> Either elements that protect the ERS from vehicular impact, (such as barrier along the bottom of the ERS) or elements that protect			
pedes	trian and vehicular traffic from	the drop-	off created by the ERS (such as barriers, parapets, or railing along the top of the ERS).	
No.	Name	CS Table	Description	
937	Wall Railing/Barrier, Steel	3*	All types and shapes of metal railing/barrier. Steel, aluminum, metal beam, rolled shapes, etc. will all be considered part of this element. Included in this element are the posts of metal, timber, concrete, masonry, blocking and curb. This includes thrie-beam retrofit.	
938	Wall Railing/Barrier, Reinforced Concrete	1	All types and shapes of reinforced concrete railing/barrier. All elements of the railing (not including incidentals such as handrails or pedestrian fencing) must be concrete.	
939	Wall Railing/Barrier, Timber	4*	All types and shapes of timber railing/barrier. Included in this element are the posts of metal, timber, concrete, masonry, blocking and curb.	
940	Wall Railing/Barrier, Masonry	5	All types and shapes of masonry, stone railing/barrier. All elements of the railing must be masonry, stone.	
941	Wall Railing/Barrier, Other	6*	Any type of railing/barrier that cannot be classified using any other defined railing element.	
Quantity Calculation: The quantity for this element is the number of railings/barriers times the length of element.				
Eleme	ent Commentary: * Mixed mat	erials on ra	ailings may require referring to multiple CS-Tables	

# ARCHITECTURAL FACING (ea.)

Descri	Description: Any of several possible aesthetic treatments that may be done to the wall facing of the ERS that do not affect the structural integrity of the			
ERS (s	uch as form-lined or precast re	lief, sculpt	ted surfaces, embedments, thru-color, etc.).	
No.	Name	CS Table	Description	
942	Architectural Facing, Steel	3	Any type of architectural facing done to a wall with steel facing elements.	
943	Architectural Facing, Concrete	1	Any type of architectural facing done to a wall with concrete facing elements.	
944	Architectural Facing, Timber	4	Any type of architectural facing done to a wall with timber facing elements.	
945	Architectural Facing, Masonry	5	Any type of architectural facing done to a wall with masonry facing elements.	
946	Architectural Facing, Other	6	Any type of architectural facing that cannot be classified using any other defined architectural facing element.	
Quantity Calculation: The quantity for these elements is each. The condition of one element may affect another causing multiple elements to be				
recorded.				
Eleme	Element Commentary:			

# PROTECTIVE COATING AND SYSTEMS (sq. ft.)

Descrip	Description: Any of several possible surface treatments that may be done to the exposed surfaces of the wall facing or other elements of the ERS to			
protect	the surface from weathering	, chemical	attack, vandalism, etc. (such as paint, stain, galvanizing, waterprooting, etc.	
No.	Name	CS Table	Description	
515	Steel Protective Coating	12	This element is for steel elements that have a corrosion inhibiting protective coating.	
521	Concrete Protective Coating	12	This element is for concrete elements that have a protective coating. These coatings include silane/siloxane water proofers, crack sealers such as High Molecular Weight Methacrylate (HMWM), or any topcoat barrier that protects concrete from deterioration and reinforcing steel from corrosion.	
849	A588 Steel Patina	12	This ADE should be used instead of element 515 and is only for the quantity of A588 steel patina that is exposed directly to the elements and not protected with any other system.	
850	Healer Sealer	12	This element is for penetrating sealer (healer sealer) that has been applied as a flood coat to the horizontal surface in order to inhibit moisture and chloride intrusion. The material is designed to wear from the exposed surface over time, and maintain an impermeable seal in cracks that were present prior to application. For the evaluation of healer sealers use CS Table 12 defect Effectiveness – Concrete Protective Coatings.	
899	Fiber Reinforced Polymer	12	This element is for FRP sheet and adhesive composite systems that have been applied to columns, beam ends, or other elements.	
947	Timber Protective System	12	This element is for timber elements that have a field-applied preservative or fire resistant coating applied to them.	
948	Other Protective System	12	Any type of protective coating system that cannot be classified using any other defined protective coating system element.	
Quanti	ty Calculation: The quantity f	or this eler	nent should include the entire area of protected surface for the element. The steel protective coating for	
superst	superstructure elements for superstructure elements will be calculated by first determining the visible surface area of the primary structural elements			
(i.e. the	(i.e. the top face of top flange is excluded) then adding 10% to account for secondary members.			
Elemer	nt Commentary:			

# **OVERLAND CONDITIONS (ea.)**

Description: Any additional elements of land or water features adjacent but a part of the ERS.					
No.	Name	CS Table	Description		
949	Channel Condition	11	This element describes any waterway feature along the front face of the ERS.		
950	Channel Protection Material and Condition	11	This element describes the bank condition of a waterway feature along the face of the ERS. The protection material includes soil, fill, stone, rock, or concrete.		
951	Upslope Material and Condition	11	This element describes the condition of the slope upward from the back/top of the ERS. The protection material includes soil, fill, stone, rock, or concrete.		
952	Downslope Material and Condition	11	This element describes the condition of the slope downward from the face/bottom of the ERS. The protection material includes soil, fill, stone, rock, or concrete.		
953	Lateral Material and Condition	11	This element describes the lateral condition of the ERS. The protection material includes soil, fill, stone, rock, or concrete.		
954	Leveling Pad/Toe Protection	11	This element describes the condition material that the footing/foundation is set on, and the protection of the foundation of the ERS. The material includes soil, fill, stone, rock, or concrete.		
955	Adjacent Slope	11	This element describes any adjacent slopes that may affect the ERS. The protection material includes soil, fill, stone, rock, or concrete.		
956	Berm	11	This element describes a flat strip of land, raised bank, or terrace bordering a waterway feature.		
<b>Quan</b> record	Quantity Calculation: The quantity for these elements is each. The condition of one element may affect another causing multiple elements to be recorded.				
Eleme	ent Commentary:				

#### SCOUR PROTECTION

Desc	Description: These elements define scour protection devices used to armor piers and abutments.			
No.	Name	CS Table	Description	
830	Plain Rip Rap	10	Angular interlocking stone with a median diameter of 8" The quantity for this element is measured in square feet.	
831	Heavy Rip Rap	10	Angular interlocking stone interlocking with a median diameter of 16". The quantity for this element is measured in square feet.	
829	Field Stone	10	Natural rounded stone with diameters varying from 8"-24". The quantity for this element is measured in square feet.	
832	Channel Armoring	10	Channel bed, banks or embankment slopes surfaced with cast-in-place concrete to resist erosion and scour. The quantity for this element is measured in square feet.	
833	Articulating Concrete Block	10	Preformed units that either interlock, are held together by cables, or both to form a continuous blanket or block matrix. The quantity for this element is measured in square feet.	
834	Gabion	10	Basket or compartmented rectangular containers made of wire mesh filled with cobbles or other rock. The quantity for this element is measured in feet along the length of the protected structure.	
835	Grout Filled Bags	10	Fabric bags filled with grout used for scour protection. The quantity for this element is measured in feet.	
836	Sheet Piling	10	A continuous line of driven steel sheeting used for scour protection. The quantity for this element is measured in feet along the length of the protected structure.	
837	Other Scour Countermeasures	10	Countermeasures that cannot be classified by any other defined scour countermeasure. The quantity for this element is measured in feet along the length of the protected structure.	
<b>Qua</b> from	ntity Calculation: The quantition the structure. See description	es are mean for units	asured along the substructure or culvert element protected and the extensions upstream and downstream of measure.	

#### Element Commentary:

# **APPURTENANCES (ea.)**

Descri	Description: These elements define the components included in installing an overhead sign or utility conduits mounted to an ERS.				
No.	Name	CS Table	Description		
960	Wall Sign, Cantilever Mounted	14	This element describes any signage attached to the ERS structure with a cantilever mounted or bracketed system.		
961	Wall Sign, Mounted	14	Description: This element describes any signage attached to the ERS structure flush mounted with bolted, anchored, or other attachment hardware.		
957	Access Panels	15	This element describes any access panels for utilities or structural access within the ERS structure which failure would have a negative effect on the integrity of the ERS		
958	Utilities	14	This element describes any conduits, pipes, or appurtenances that are attached to the ERS that carry a utility or utilities.		
<b>Quant</b> record	Quantity Calculation: The quantity for these elements is each. The condition of one element may affect another causing multiple elements to be recorded.				
Eleme	nt Commentary:				

#### **CONDITION STATE TABLES**

The condition state descriptions herein are adopted from the National Bridge Elements and Bridge Management Elements and follow guidance provided by the AASHTO Bridge Element Manual and the FHWA. The condition state descriptions for Agency Defined Elements (ADEs) are defined by MDOT. This manual attempts to cover the majority of all conditions observed in the field, but during the course of an inspection, the inspector may find conditions that are not described. In these cases, the inspector should use the general description of the condition states to determine the appropriate condition. Overarching descriptors for the four condition states are as follows:

**Condition State 1 (Good)** – that portion of the element that has either no deterioration or the deterioration is insignificant to the management of the element, meaning that portion of the element has no condition based preventive maintenance needs or repairs. Areas of an element that have received long lasting structural repairs that restore the full capacity of the element with an expected life expectancy equal to the original element can be coded as good condition.

**Condition State 2 (Fair)** – that portion of the element that has minor deficiencies that signifies a progression of the deterioration process. This portion of the element may need condition based preventive maintenance. Areas of the element that have received structural repairs that improve the element, but the repair is not considered equal to the original member can be coded as fair.

**Condition State 3 (Poor)** – that portion of the element that has advanced deterioration requiring repair. The summation of the quantity of the element in poor or worse condition determines the need for repairs, rehabilitation, or replacement activities.

**Condition State 4 (Severe)** – that portion of the element that warrants a review to determine the effect on strength or serviceability of the element or bridge; OR a structural review has been completed and the defects impact strength or serviceability of the element or bridge. Elements with a portion or all of the quantity in state 4 may often have load capacity implications warranting a structural review. Within this manual, the term structural review is defined as a review by a person qualified to evaluate the field observed conditions and make a determination of the impacts of the conditions on the performance of the element. Structural reviews may include a review of the field inspection notes and photographs, review of as-built plans or analysis as deemed appropriate to evaluate the performance of the element.

# CS TABLE 1 – REINFORCED CONCRETE

Defect	Condition State 1	Condition State 2	Condition State 3	Condition State 4
	GOOD	FAIR	POOR	SEVERE
Spalls/ Delaminations/ Patch Areas (1080)	None.	Delaminated. Spall 1 in. or less deep or less than 6 in. diameter. Patched area is sound.	Spall greater than 1 in. deep or greater than 6 in. diameter. Patched area is unsound or showing distress. Does not warrant structural review.	The condition
Exposed Rebar (1090)	None.	Present without section loss.	Present with section loss that does not warrant structural review.	warrants a structural review to
Efflorescence / Rust Staining (1120)	None.	Surface white without build-up or leaching without rust staining.	Heavy build-up with rust staining.	determine the effect on strength or serviceability of
Cracking (1) Reinforced Concrete and Other (1130)	Insignificant cracks or moderate-width cracks that have been sealed.	Unsealed moderate-width cracks or unsealed moderate pattern (map) cracking.	Wide cracks or heavy pattern (map) cracking.	the element or bridge; OR a structural review has been
Abrasion/Wear (1190)	No Abrasion of wearing	Abrasion or wearing has exposed coarse aggregate	Coarse aggregate is loose or has popped out of the concrete matrix due to abrasion or wear.	completed and the defects impact
Settlement – Substructure Elements (4000)	None.	Exists within tolerable limits or arrested with effective actions taken to mitigate.	Exceeds tolerable limits but does not warrant structural review.	strength or serviceability of the element or bridge.
Scour - Substructure / Culvert Elements (6000)	None.	Exists within tolerable limits or arrested with effective countermeasures.	Exceeds tolerable limits but is less than the limits determined by scour evaluation, and does not warrant structural review.	
Damage (7000)	Not applicable.	The element has minor damage caused by vehicular or vessel impact.	The element has moderate damage caused by vehicular or vessel impact.	The element has severe damage caused by vehicular or vessel impact.

#### CS TABLE 1 – REINFORCED CONCRETE (Continued)

(1) The inspector should use judgment when utilizing the condition state defect conditions, especially for concrete cracking. The crack defect description definitions describe generalized distress, but the inspector should consider width, spacing, location, orientation, and structure or nonstructural nature of the cracking. The inspector should consider exposure and environment when evaluating crack width. In general, reinforced concrete cracks less than 0.012 inches can be considered insignificant and a defect is not warranted. Cracks ranging from 0.012 to 0.05 inches can be considered wide.



#### Michigan Earth Retaining System Element Inspection Manual

Condition State 2	Condition State 3	Condition State 4
FAIR	POOR	SEVERE
27/24/2242		

# CS TABLE 2 – PRESTRESSED CONCRETE

Defect	Condition State 1	Condition State 2	Condition State 3	Condition State 4
	GOOD	FAIR	POOR	SEVERE
Spalls/ Delaminations/ Patch Areas (1080)	None.	Delaminated. Spall 1 in. or less deep or less than 6 in. diameter. Patched area is sound.	Spall greater than 1 in. deep or greater than 6 in. diameter. Patched area is unsound or showing distress. Does not warrant structural review.	
Exposed Rebar (1090)	None.	Present without section loss.	Present with section loss that does not warrant structural review.	The condition
Exposed Prestressing (1100)	None.	Present without section loss.	Present with section loss that does not warrant structural review.	varrants a structural review to determine the effect on strength
Cracking <sup>(1)</sup> - PSC (1110)	Insignificant cracks or moderate-width cracks that have been sealed.	Unsealed moderate-width cracks or unsealed moderate pattern (map) cracking.	Wide cracks or heavy pattern (map) cracking.	element or bridge; OR a structural review
Efflorescence / Rust Staining (1120)	None.	Surface white without build-up or leaching without rust staining.	Heavy build-up with rust staining.	and the defects impact strength or
Settlement – Substructure (4000)	None.	Exists within tolerable limits or arrested with effective actions taken to mitigate.	Exceeds tolerable limits but does not warrant structural review.	element or bridge.
Scour - Substructure (6000)	None.	Exists within tolerable limits or arrested with effective countermeasures.	Exceeds tolerable limits but is less than the limits determined by scour evaluation, and does not warrant structural review.	
Damage (7000)	Not applicable.	The element has minor damage caused by vehicular or vessel impact.	The element has moderate damage caused by vehicular or vessel impact.	The element has severe damage caused by vehicular or vessel impact.

#### CS TABLE 2 – PRESTRESSED CONCRETE (Continued)

(1) The inspector should use judgment when utilizing the condition state defect conditions, especially for concrete cracking. The crack defect description definitions describe generalized distress, but the inspector should consider width, spacing, location, orientation, and structure or nonstructural nature of the cracking. The inspector should consider exposure and environment when evaluating crack width. In general, reinforced concrete cracks less than 0.004 inches can be considered insignificant and a defect is not warranted. Cracks ranging from 0.004 to 0.009 inches can be considered moderate, and cracks greater than 0.009 inches can be considered wide.



# CS TABLE 3 – STEEL

Defect	Condition State 1	Condition State 2	Condition State 3	Condition State 4
	GOOD	FAIR	POOR	SEVERE
Corrosion (1000)	None.	Freckled Rust. Corrosion of the steel has initiated.	Section loss is evident or pack rust is present but does not warrant structural review.	
Cracking/Fatigue (1010)	None.	Cracks that have self-arrested or have been arrested with effective arrest holes, doubling plates or similar.	Identified cracks exist that are not arrested and do not require structural review.	The condition
Connections (1020)	Connection is in place and functioning as intended.	Pack rust without distortion is present but the connection is in place and functioning as intended. Loose or missing fasteners accumulating in less than 10% of total fasteners.	Missing bolts, rivets, broken welds, fasteners or pack rust with distortion but do not warrant a structural review.	warrants a structural review to determine the effect on strength or serviceability of the element or bridge; OR a structural review
Distortion (1900)	None.	Distortion not requiring mitigation or mitigated distortion.	Distortion that requires mitigation but does not require structural review.	has been completed and the defects
Settlement – Substructure Elements (4000)	None.	Exists within tolerable limits or arrested with effective actions taken to mitigate.	Exceeds tolerable limits but does not warrant structural review.	impact strength or serviceability of the element or bridge.
Scour – Substructure Elements (6000)	None.	Exists within tolerable limits or arrested with effective countermeasures.	Exceeds tolerable limits but is less than the limits determined by scour evaluation, and does not warrant structural review.	
Damage (7000)	Not applicable.	The element has minor damage caused by vehicular or vessel impact.	The element has moderate damage caused by vehicular or vessel impact.	The element has severe damage caused by vehicular or vessel impact.

#### CS TABLE 3 – STEEL (Continued)



#### CS TABLE 3 – STEEL (Continued)



#### CS TABLE 4 – TIMBER

Defect	Condition State 1	Condition State 2	Condition State 3	Condition State 4
	GOOD	FAIR	POOR	SEVERE
Decay/ Section Loss (1140)	None.	Affects less than 10% of the member section	Affects 10% or more of the member but does not warrant structural review.	
Checks/Shakes (1150)	Surface penetration less than 5% of the member thickness regardless of location.	Penetrates 5% - 50% of the thickness of the member and not in a tension zone.	Penetrates more than 50% of the thickness of the member or more than 5% of the member thickness in a tension zone. Does not warrant structural analysis.	The condition warrants a structural review to determine the effect on strength or serviceability of the
Cracks - Timber (1160)	None.	Cracks that have been arrested through effective measures.	Identified cracks exist that are not arrested and do not require structural review.	element or bridge; OR a structural review has been completed
Splits/ Delamination - Timber (1170)	None.	Length less than the member depth or arrested with effective actions taken to mitigate.	Length greater than the member depth and does not require structural review.	and the defects impact strength or serviceability of the element or bridge.
Abrasion (1180)	None or no measurable section loss.	Section loss less than 10% of the member thickness.	Section loss 10% or more of the member thickness but does not warrant structural review.	
Damage (7000)	Not applicable.	The element has minor damage caused by vehicular or vessel impact.	The element has moderate damage caused by vehicular or vessel impact.	The element has severe damage caused by vehicular or vessel impact.

#### CS TABLE 5 – MASONRY

Defect	Condition State 1	Condition State 2	Condition State 3	Condition State 4
	GOOD	FAIR	POOR	SEVERE
Spalls/ Delaminations/ Patch Areas (1080)	None.	Delaminated. Spall 1 in. or less deep or less than 6 in. diameter. Patched area is sound.	Spall greater than 1 in. deep or greater than 6 in. diameter. Patched area is unsound or showing distress. Does not warrant structural review.	
Efflorescence (1120)	None.	Surface white without build-up or leaching without rust staining.	Heavy build-up with rust staining.	warrants a structural
Mortar Breakdown - Masonry (1610)	None.	Cracking or voids in less than 10% of joints.	Cracking or voids in 10% or more of the joints.	the effect on strength or serviceability of the element or bridge; OR
Splits or Spalls - Masonry (1620)	None.	Block or stone has split or spalled with no shifting.	Block or stone has split or spalled with shifting but does not warrant a structural review.	has been completed and the defects
Patched Areas - Masonry (1630)	None.	Sound patches.	Unsound patches.	serviceability of the element or bridge.
Masonry Displacement (1640)	None.	Block or stone has shifted slightly out of alignment.	Block or stone has shifted significantly out of alignment or is missing but does not warrant structural review.	
Damage (7000)	Not applicable.	The element has minor damage caused by vehicular or vessel impact.	The element has moderate damage caused by vehicular or vessel impact.	The element has severe damage caused by vehicular or vessel impact.

#### CS TABLE 5 – MASONRY (Continued)



# CS TABLE 6 – OTHER MATERIALS

Defect	Condition State 1	Condition State 2	Condition State 3	Condition State 4
	GOOD	FAIR	POOR	SEVERE
Corrosion (1000)	None.	Freckled rust. Corrosion of the steel has initiated.	Section loss is evident or pack rust is present but does not warrant structural review.	
Cracking/Fatigue (1010)	None.	Cracks that have self-arrested or have been arrested with effective arrest holes, doubling plates, or similar.	Identified cracks exist that are not arrested and do not require structural review.	The condition
Connections (1020)	Connection is in place and functioning as intended.	Loose fasteners or pack rust without distortion is present but the connection is in place and functioning as intended.	Missing bolts, rivets, broken welds, fasteners or pack rust with distortion but do not warrant a structural review.	review to determine the effect on strength or
Spalls/ Delaminations/ Patch Areas (1080)	None.	Delaminated. Spall 1 in. or less deep or less than 6 in. diameter. Patched area is sound.	Spall greater than 1 in. deep or greater than 6 in. diameter. Patched area is unsound or showing distress. Does not warrant structural review.	element or bridge; OR a structural review has been
Efflorescence / Rust Staining (1120)	None.	Surface white without build-up or leaching without rust staining.	Heavy build-up with rust staining.	defects impact strength or
Cracking <sup>(1)</sup> Reinforced Concrete and Other (1130)	Insignificant cracks or moderate-width cracks that have been sealed	Unsealed moderate-width cracks or unsealed moderate pattern (map) cracking.	Wide cracks or heavy pattern (map) cracking.	element or bridge.
Deterioration (Other) (1220)	None.	Initiated breakdown or deterioration.	Significant deterioration or breakdown that does not warrant structural review.	

Michigan Earth Retaining System Element Inspection Manual

Defect	Condition State 1	Condition State 2	Condition State 3	Condition State 4
	GOOD	FAIR	POOR	SEVERE
Damage (7000)	Not applicable.	The element has minor damage caused by vehicular or vessel impact.	The element has moderate damage caused by vehicular or vessel impact.	The element has severe damage caused by vehicular or vessel impact.

(1) The inspector should use judgment when utilizing the condition state defect conditions, especially for concrete cracking. The crack defect description definitions describe generalized distress, but the inspector should consider width, spacing, location, orientation, and structure or nonstructural nature of the cracking. The inspector should consider exposure and environment when evaluating crack width. In general, reinforced concrete cracks less than 0.012 inches can be considered insignificant and a defect is not warranted. Cracks ranging from 0.012 to 0.05 inches can be considered wide.

## CS TABLE 7 – ANCHORS

Defect	Condition State 1	Condition State 2	Condition State 3	Condition State 4
	GOOD	FAIR	POOR	SEVERE
Corrosion (1000)	None.	Freckled Rust. Corrosion of the steel has initiated.	Section loss is evident or pack rust is present but does not warrant structural review.	The condition warrants a structural review to determine the effect on
Connections (1020)	Connection is in place and functioning as intended.	Loose fasteners or pack rust without distortion is present but the connection is in place and functioning as intended.	Missing bolts, rivets, broken welds, fasteners or pack rust with distortion but do not warrant a structural review.	strength or serviceability of the element or bridge; OR a structural review has been completed and
Distortion (1900)	None.	Distortion not requiring mitigation or mitigated distortion.	Distortion that requires mitigation but does not require structural review.	the defects impact strength or serviceability of the element or bridge.
Seal Adhesion – Anchors with Seals Only (2320)	Fully adhered.	Adhered for more than 50% of the joint height.	Adhered 50% or less of joint height but still some adhesion.	Complete loss of adhesion.
Seal Damage – Anchors with Seals Only (2330)	None.	Seal abrasion without punctures.	Punctured, ripped or partially pulled out.	Punctured completely through, pulled out, or missing.
Seal Cracking – Anchors with Seals Only (2340)	None.	Surface cracks.	Cracks that partially penetrate the seal.	Cracks that fully penetrate the seal.
Debris Impaction (2350)	None.	Partially filled, but still allowing free movement.	Completely filled and impacts joint movement.	Completely filled and prevents joint movement.

Michigan Earth Retaining System Element Inspection Manual

Defect	Condition State 1	Condition State 2	Condition State 3	Condition State 4
	GOOD	FAIR	POOR	SEVERE
Damage (7000)	Not applicable.	The element has minor damage caused by vehicular or vessel impact.	The element has moderate damage caused by vehicular or vessel impact.	The element has severe damage caused by vehicular or vessel impact.

# CS TABLE 8 – JOINTS

Defect	Condition State 1	Condition State 2	Condition State 3	Condition State 4
	GOOD	FAIR	POOR	SEVERE
Leakage (2310)	None.	Minimal. Minor dripping through the joint.	Moderate. More than a drip and less than free flow of water.	Free flow of water through the joint.
Seal Adhesion – Joints with Seals Only (2320)	Fully adhered.	Adhered for more than 50% of the joint height.	Adhered 50% or less of joint height but still some adhesion.	Complete loss of adhesion.
Seal Damage – Joints with Seals Only (2330)	None.	Seal abrasion without punctures.	Punctured, ripped or partially pulled out.	Punctured completely through, pulled out, or missing.
Seal Cracking – Joints with Seals Only (2340)	None.	Surface cracks.	Cracks that partially penetrate the seal.	Cracks that fully penetrate the seal.
Debris Impaction (2350)	None.	Partially filled, but still allowing free movement.	Completely filled and impacts joint movement.	Completely filled and prevents joint movement.
Adjacent Deck or Header (2360)	Sound. No spalls, delamination or unsound patches.	Edge delamination or spall less than 1 in. deep or less than 6 in. diameter. No exposed rebar. Patched area is sound.	Spall greater than 1 in. deep or greater than 6 in. diameter. Exposed rebar. Delamination or unsound patched area that makes the joint loose.	Spall, delamination, unsound patched area or loose joint anchor that impacts joint performance.
Damage (7000)	None.	The element has minor damage caused by vehicular or vessel impact.	The element has moderate damage caused by vehicular or vessel impact.	The element has severe damage caused by vehicular or vessel impact.
Pressure Relief (TBD)	Relief Joint is fully adhered and measures 4" wide	Joint measures 3" wide	Joint measures 2"wide	Joint material is missing, has lost adhesion or measures 1″ wide



PRJ

#### CS TABLE 9 – RETAINED MATERIALS

Defect	Condition State 1	Condition State 2	Condition State 3	Condition State 4
	GOOD	FAIR	POOR	SEVERE
Settlement	None.	within tolerable limits or arrested with no observed structural distress.	Exceeds tolerable limits but does not warrant a structural review.	Safety: Requires immediate action to ensure safety of public traffic.
Distortion	None.	Exists but does not require mitigation. Distortion that has been mitigated.	Distortion that requires mitigation that has not been addressed.	Serviceability: The condition is beyond the limits established in condition state three (3), warrants a
Erosion	None.	Erosion less than 2-ft wide or deep.	Exposed top corner of leveling pad that is on rock.	structural review to determine the strength or serviceability of the element or ERS, or both.

### CS TABLE 10 – SCOUR PROTECTION

Defect	Condition State 1	Condition State 2	Condition State 3	Condition State 4
	GOOD	FAIR	POOR	SEVERE
Scour or Erosion	None.	Countermeasure is substantially effective. Scour or Erosion exists without undermining.	Countermeasure device has limited effectiveness Erosion may be evident with undermining of countermeasure.	
Material Defect (scaling, abrasion, spalling, corrosion, cracking, splitting and decay)	Insignificant or minor defects.	Countermeasure device is substantially effective. Extensive minor to isolated advanced defects.	Scour countermeasures have limited effectiveness. Extensive advanced to major defects.	The channel protection device or scour countermeasure are unstable, missing or no longer effective.
Damage (unraveling, displacement, separation, and sagging)	Insignificant or minor damage.	Countermeasure device is substantially effective. Extensive minor to isolated advanced damage.	Scour countermeasures have limited effectiveness. Extensive advanced to major damage.	
# CS TABLE 11 – OVERLAND CONDITIONS

Defect	Condition State 1	Condition State 2	Condition State 3	Condition State 4
	GOOD	FAIR	POOR	SEVERE
Counter measures	Present	Minor Damage	Undermined, rip rap washed away, structure is still stable.	
Banks	Stable	Minor Slumping	Slumping	Serviceability or
Alignment	As Constructed	Minor Problems, Misalignment, angle has changed.	Misaligned, flow along foundation to expose footing or behind wall, structure is still stable.	Deficiency: The condition is beyond the limits
Embankments or slope protection	Moderate rutting from drainage. Minor bare soil exposed.	Minor Erosion caused by drainage or channel, Evidence of minor or stable foundation settlement, Erosion to embankment impacting guardrail performance or encroaching on shoulder.	Major erosion caused by drainage or channel; Erosion to embankment impacting guardrail (up to 6" of guardrail post exposed) performance or encroaching on shoulder. Evidence of foundation settlement.	state three (3), warrants a structural review to determine the strength or serviceability of the element or bridge, or both.

## CS TABLE 12 – PROTECTIVE SYSTEMS

Defect	Condition State 1	Condition State 2	Condition State 3	Condition State 4
	GOOD	FAIR	POOR	SEVERE
Chalking - Steel Protective Coatings (3410)	None.	Surface dulling.	Loss of pigment.	Not applicable.
Peeling/Bubbling/C racking - Steel Protective Coatings (3420)	None.	Finish coats only.	Finish and primer coats.	Exposure of bare metal.
Oxide Film Degradation Color/ Texture Adherence – Steel Protective Coatings (3430)	Yellow-orange or light brown for early development. Chocolate-brown to purple- brown for fully developed. Tightly adhered, capable of withstanding hammering or vigorous wire brushing.	Granular texture.	Small flakes, less than 1/2 in. diameter.	Dark black color. Large flakes, 1/2 in. diameter or greater or laminar sheets or nodules.
Effectiveness - Steel Protective Coatings (3440)	Fully effective.	Substantially effective.	Limited effectiveness.	Failed. No protection of the underlying metal.
Wear - Concrete Protective Coatings (3510)	None.	Underlying concrete not exposed. Coating showing wear from UV exposure. Friction course missing.	Underlying concrete is not exposed and thickness of the coating is reduced.	Underlying concrete exposed. Protective coating no longer effective.
Effectiveness - Concrete Protective Coatings (3540)	Good condition. Fully effective.	Substantially effective.	Limited effectiveness.	The protective system has failed or is no longer effective.

Michigan Earth Retaining System Element Inspection Manual

Defect	Condition State 1	Condition State 2	Condition State 3	Condition State 4
	GOOD	FAIR	POOR	SEVERE
Effectiveness - Protective System [e.g. cathodic, scour monitoring] (3600)	Fully effective.	Substantially effective.	Limited effectiveness.	The protective system has failed or is no longer effective.
Damage (7000)	Not applicable.	The element has minor damage caused by vehicular or vessel impact.	The element has moderate damage caused by vehicular or vessel impact.	The element has severe damage caused by vehicular or vessel impact.

Condition State 2	Condition State 3	Condition State 4
FAIR	POOR	SEVERE







Condition State 2	Condition State 3	Condition State 4
FAIR	POOR	SEVERE











### CS TABLE 13 – MSE WALLS

Defect	Condition State 1	Condition State 2	Condition State 3	Condition State 4
	GOOD	FAIR	POOR	SEVERE
Cracking – Reinforced Concrete Stub	Insignificant cracks or moderate-width cracks that have been sealed.	Unsealed moderate-width cracks or unsealed moderate pattern (map) cracking.	Wide cracks or heavy pattern (map) cracking.	
Cracking - Wall Panels/Coping	None.	Insignificant non-structural cracks without surface staining.	Structural cracks or cracking with surface staining.	
Joints	Wall panel joint spacing is substantially uniform.	Wall panel joint width exceeds as-built spacing without geotextile fabric exposure.	Wall panel joint width exceeds as- built spacing or is irregular with exposed geotextile fabric. Does not warrant structural review.	The condition warrants a structural review to
Wall Tilting & Misallignment	None.	Minor uniform tilting of wall section. Minor wall misalignment.	Moderate uniform tilting of wall section. Moderate wall misalignment.	strength and serviceability of the element or bridge;
Panel Bowing	None.	Panels have bowed without geotextile fabric exposure.	Panels have bowed with geotextile fabric or connections visible. Does not warrant structural review.	been completed and the defects impact strength or serviceability of the
Erosion	None.	Minor erosion visible without exposure of the leveling pad.	Erosion has exposed the leveling pad without undermining. No wall reinforcement exposed or loss of engineered fill. Does not warrant structural review.	element of bridge.
Damage	Not applicable.	The element has minor damage caused by vehicular impact.	The element has moderate damage caused by vehicular impact.	

Condition State 2	Condition State 3	Condition State 4
FAIR	POOR	SEVERE



Loss of Backfill



Joint Spacing



Panel Bowing



Minor Spall



Moderate Misalignment



**Deteriorated Panel** 

# CS TABLE 14 – APPURTENANCES

Defect	Condition State 1	Condition State 2	Condition State 3	Condition State 4
	GOOD	FAIR	POOR	SEVERE
Alignment	Steel cantilevers, aluminum columns, and elastomeric pads are properly aligned.	Minor misalignment of steel cantilevers, or aluminum columns do not cover the full length of the sign. Elastomeric pad exhibits 20% loss of contact.	Sagging or misalignment of steel cantilever or aluminum columns is evident. Elastomeric pad exhibits between 20% and 60% loss of contact.	Sagging or misalignment of steel cantilever or aluminum columns warrants replacement. Elastomeric pad exhibits greater than 60% loss of contact.
Steel Wall Connections	Bolts are tight, sound, and well engaged. All washers present. Nuts are located on the interior face of the fascia beam. No corrosion present on the bolt, nut, or washer.	There is evidence of misalignment but tight bolts, or bolts missing washers, or small washers in oversized holes.	Bolts are missing, or there is evidence of broken welds. Between 0-20% of bolts in the connection are loose. Between 0-10% of fasteners in the connection are cracked, broken, or missing. Misaligned but tight bolts, or bolts with missing washers.	Greater than 20% of the bolts in the connection are loose. Greater than 10% of the fasteners in the connection are cracked, broken, or missing. Fastener proximity to edge of member is less than 1.5 times the bolt diameter. Fastener is missing, corroded, or improperly aligned. Washers cupped or bolt hole visible.
Concrete Anchored Connections	Bolts are tight, sound, and well engaged. Steel member flush with concrete surface.	Less than 10% of bearing surface exhibits light scaling, honeycombing, or ASR. Insignificant concrete cracking present.	Between 10% to 40% of bearing surface exhibits scaling, honeycombing, or ASR. Moderate concrete cracking or map cracking present.	Greater than 40% of the bearing surface exhibits scaling, honeycombing, or ASR. Wide concrete cracks or heavy map cracking present. Spalling, delamination, or anchor failure present.

Defect	Condition State 1	Condition State 2	Condition State 3	Condition State 4
	GOOD	FAIR	POOR	SEVERE
Cracking/Fatigue	None.	There is evidence of superficial concrete cracking in the vicinity of the bridge connection.	Concrete cracking continues to develop or increase in length and width within the vicinity of the bridge connection.	Cracking within the vicinity of the bridge connection, or fatigue cracks in the steel or aluminum sign members have developed.
Vertical Adhesive Anchors	Present with no visible defects.	Adhesively anchored rod, bolt, or bar is exhibiting severe corrosion.	Misalignment or annular gap at one or more adhesively anchored rod/bolt/bar, but no evidence of pull out. Or, anchors into cracked, delaminated, or spalled concrete. Or, loose or missing hardware.	Any evidence of pull out of one or more adhesively anchored rod/bar/bolt. This warrants immediate attention as some adhesives are susceptible to creep resulting in sudden failure.
Damage/Deterio ration	Free of damage and debris.	There is minor damage to the sign or bridge connection caused by vehicular impact or environmental conditions.	There is moderate damage to the sign or bridge connection elements caused by vehicular impact, environmental conditions, or graffiti but the message is legible.	There is severe damage to the sign or bridge connection elements caused by vehicular impact, environmental conditions, or graffiti and the message is not legible.
Supports	Properly anchored and sound	Minor problem, active corrosion, loose joints but no exposed wires or leaks	Loose or missing support element but the utility is adequately supported, problems are not affecting ERS elements or public safety.	Broken or missing supports, affecting ERS elements or public safety.

Condition State 2	Condition State 3	Condition State 4
FAIR	POOR	SEVERE





Vertical Anchor



Vertical Anchor



Vertical Anchor

### CS TABLE 15 – ASPHALT

Defect	Condition State 1	Condition State 2	Condition State 3	Condition State 4
	GOOD	FAIR	POOR	SEVERE
Spalls/ Delaminations/ Patch Areas/ Potholes (3210)	None.	Delaminations. Spalls 1 in. or less deep or less than 6 in. in diameter. Patched areas are sound. Partial depth potholes.	Spalls greater than 1 in. deep or greater than 6 in. in diameter. Patched areas are unsound or showing distress. Full depth potholes.	
Cracks (3220)	Widths less than 0.012 in. or spacing greater than 3.0 ft.	Widths 0.012–0.05 in. or spacing of 1.0–3.0 ft.	Widths of more than 0.05 in. or spacing of less than 1.0 ft.	The asphalt is no longer
Effectiveness (3230)	Fully effective. No evidence of leakage or further deterioration of the protected element.	Substantially effective. Deterioration of the protected element has slowed.	Limited effectiveness. Deterioration of the protected element has progressed.	effective.
Damage (7000)	Not applicable	The element has minor damage caused by vehicular or vessel impact.	The element has moderate damage caused by vehicular or vessel impact.	